THESIS ON NATURAL AND EXACT SCIENCES

Inverse problems to determine non-homogeneous degenerate memory kernels in heat flow

ENNO PAIS

TALLINN UNIVERSITY OF TECHNOLOGY

Faculty of Science Department of Mathematics

Dissertation was accepted for the defence of the degree of Doctor of Philosophy in Natural Sciences on June 12, 2007

Supervisor: Prof. Jaan Janno, Department of Mathematics, Faculty of Sciens, Tallinn University of Technology

Opponents: Dr. rer. nat. habil., Prof. Lothar von Wolfersdorf, Freiberg University of Mining and Technology, Germany
Cand. Sck, Prof. Arvet Pedas, University of Tartu, Estonia

Defence of the thesis: October 2, 2007

Declaration:

Hereby I declare that this doctoral thesis, my original investigation and achievement, submitted for the doctoral degree at Tallinn University of Technology has not been submitted for any degree.

/Enno Pais/

Copyright: Enno Pais, 2007 ISSN 1406-4723

ISBN 99:/;;:7/7;/934/7

Contents

 Formulation of direct and inverse problems 1.1 Problems in time domain 1.2 Problems in Laplace domain 1.2.1 Laplace transform and its basic properties 1.2.2 Direct problem in the case of boundary conditions of first kind 	the the	66 68 88 10 12 14
 1.1 Problems in time domain	the the	8 8 10 12
 1.2 Problems in Laplace domain	the	10 12
1.2.1 Laplace transform and its basic properties1.2.2 Direct problem in the case of boundary conditions of	the the	10 12
	the	12
first kind	the	12
	· · ·	
1.2.3 Direct problem in the case of boundary conditions of	3.	
third kind		14
1.2.4 Summary. Generalized formulation of inverse problems		
2. Functional spaces and properties of Green function	• •	16
2.1 Functional spaces		16
2.2 Asymptotical representation of Green function		17
2.2.1 Asymptotical representation of a solution of Cauchy pro	blem	17
2.2.2 Basic properties and representation of Green function case of the first kind boundary conditions		23
2.2.3 Basic properties and representation of Green function		
case of the third kind boundary conditions		26
2.3 Auxiliary results		29
2.4 Asymptotical properties of Green function in case of the first k	ind	
boundary conditions		32
2.5 Asymptotical properties of Green function in case of the third k	ind	
boundary conditions		37
3. Inverse problem with temperature observations		42
3.1 Reduction of the inverse problem to a fixed-point form		42
3.2 Analysis of direct problem system		48
3.3 Existence and uniqueness for inverse problem		54
3.4 Interpretation of assumptions		61
3.4.1 Interpretation of regularity assumptions		61
3.4.2 Interpretation of non-vanishing condition		65

4.	Inve	rse problem with flux observations	67	
	4.1	Reduction of the inverse problem to a fixed-point form	67	
	4.2	Analysis of direct problem system	71	
	4.3	Existence and uniqueness for inverse problem	75	
	4.4	Interpretation of assumptions	80	
		4.4.1 Interpretation of regularity assumptions	80	
		4.4.2 Interpretation of non-vanishing condition	81	
	4.5	The case of singular matrix. Severe ill-posedness of homogeneous		
		problem	82	
Αł	ostract	:	87	
Pö	ördüle	esanded mittehomogeensete mälutuumade määramiseks soojuslevi prot	_	
	sessi	s: eestikeelne kokkuvõte	88	
B	ibliog.	raphy	90	
El	Elulookirjeldus			
Curriculum vitae				

Introduction

In the linear theory of heat conduction in materials with memory the constitutive relations between the heat flux and the gradient of temperature and between the internal energy and the temperature involve integral terms over the past history of the material containing time-dependent convolution kernels. These memory terms bring a certain inertia into the heat flow process and enable to model the propagation of heat by finite speed [3, 14, 26, 27, 33, 34]. Models with memory are used also in coupled processes, e.g. thermoelasticity [40] and phase transition [1, 4, 5].

In case the memory kernels and other physical data are known, we can solve the direct problem, i.e. the problem to determine the temperature function. But sometimes it is necessary to solve an inverse problem, i.e. a problem where the kernels are unknown. To recover the kernels in the inverse problems additional measurements of the temperature or heat flux (so-called observation data) are used. Inverse problems serve for two purposes:

- 1) testing the relevance of the model solving the inverse problem several times with different data, an approximate coincidence of the solutions proves the relevance of the model, whereas a big difference between the solutions shows the irrelevance;
- 2) practical determination of the memory kernels for particular materials.

Different inverse problems for memory kernels in heat flow in homogeneous case have been posed and studied in a number of papers [2, 10, 11, 12, 16, 18, 19, 20, 22, 25, 29, 30, 31, 41, 42, 43]. Most of these works deal with the case when the heat equation contains a single memory kernel of the heat flux. However, the papers [11, 20] treat the inverse problems for two kernels, i.e. for the kernels of heat flux m and the internal energy n. The character of the problem for two kernels very much depends on the type of observation conditions. In case temperature observations in two points are given ahead, one determines n with higher smoothness than m [20]. In case both temperature and flux observations in single points are provided, one gets n and m with the same level of smoothness [11, 20]. The third case when purely flux observations are given, was not covered by the papers [11, 20]. This case turns out to be more complicated.

When the material is non-homogeneous, then the memory kernels depend on the space variable(s), too. In this case the inverse problems require more observation data. The most general approach is to make use of a restricted Dirichletto-Neumann map to determine the kernel of heat flux [15]. This requires a lot of measurements from different experiments. The situation simplifies in case one possesses additional a priori information on the kernels. For instance, if the body under consideration is stratified or satisfies other symmetry properties, it is possible to recover the kernel of heat flux from measurements obtained from a single experiment [6, 7, 9, 17, 32].

In some contexts the non-homogeneous kernels can be degenerate, i.e. represented as finite sums of products of known space-dependent functions times unknown time-dependent coefficients. This is so when either the medium is piecewise homogeneous or a problem for a general kernel is replaced by a related problem for an approximated kernel. Then the unknown coefficients can be recovered by the measurement of temperature or heat flux in finite number of points over the time. The problem requires only a single physical experiment. In [21, 23, 24] inverse problems of such a type were studied. Again, these papers deal with the problems to recover the kernel of heat flux.

The main task of the present thesis is to work out certain inverse problems to determine both the kernel of heat flux m and the kernel of internal energy n in the non-homogeneous degenerate case. We will be limited to the one-dimensional problems. We pose and study two different problems. The first one is a problem with purely temperature observations. We prove the existence and uniqueness of the solution. In the solvability theorems the solution occurs with one step higher smoothness in n than in m. The results concerning the problem with temperature observations have been published in the cases of first and third kind boundary conditions in the papers of the author [37] and [38], respectively.

The second problem is the inverse problem with purely flux observations. Treatment of this problem is more complicated. First of all it is necessary to establish a certain second order asymptotical relation for the Green function of the corresponding elliptic problem in the Laplace domain. This requires a lot of technical work. However, having this asymptotical relation already, the proof of existence and uniqueness of the solution is somewhat easier than for the problem with temperature observations. The results in the case of flux observations have been published in the papers of the author [35, 36]. Unfortunately, these papers contain a mistake in the estimation of the Green function. Namely, the estimation of higher-order O-terms in proof of Lemma 3 in these papers is not right. This in turn leads to an error in the asymptotical estimate of an integral of G_x . More precisely, the estimates (2.4.7) and (2.5.7) of the thesis occur with the factor $\frac{V'(x)}{\beta(x)}$ instead of

 $\left(\frac{V(x)}{\beta(x)}\right)'$ in these papers. This means that they are valid only in the case of constant β . This error is improved in the thesis. Moreover, the thesis contains the treatment of the problem with flux observations in the case of boundary conditions of the third kind, not published by the author so far.

The character of the inverse problem with flux observations is worse than the character of the problem with temperature observations. In many combinations of the data the matrix of the problem is singular. Even more, the corresponding homogeneous problem is severely ill-posed.

Summing up, the *main novelties of the thesis* are as follows.

- 1. First time inverse problems for both the kernels of heat flux and internal energy in the non-homogeneous case are dealt with.
- 2. First time an inverse problem to determine the kernels of heat flux and internal energy using purely flux observations is studied.
- 3. First time severe ill-posedness of a memory-identification problem in continuous media is noticed.

The major results of the thesis are published in the papers

- 1. E. Pais, Identification of degenerate time- and space-dependent kernels in heat flow. In: *Proc. of 5-th International Conference on Inverse Problems in Engineering* (Cambridge, 11-15.07.2005), D. Lesnic ed., Leeds Univ. Press, Leeds, 2005, Vol. III, P01, pp. 1–10.
- 2. E. Pais and J. Janno, Identification of two degenerate time- and space-dependent kernels in a parabolic equation. *Electron. J. Diff. Eqns.* (http://www.emis.de/journals/EJDE) 2005, No. 108, 1–20.
- 3. E. Pais, Degenerate memory kernels identification problem with flux-type additional conditions. *J. Inv. Ill-Posed Problems* **14** (2006), 397 418.
- 4. E. Pais and J. Janno, Inverse problem to determine degenerate memory kernels in heat flux with third kind boundary conditions. *Math. Model. Anal.* 11 (2006), 427–450.

The results of the thesis have been presented in

- 1. 5-th International Conference on Inverse problems in Engineering, Cambridge, 11-15.07.2005 poster,
- 2. seminar "Applied Functional Analysis", Tallinn, 3.03.2006 lecture,
- 3. international conference "Simulation and Optimization in Business and Industry", Tallinn, 17-20.05.2006 contributed talk,
- International Congress of Mathematicians, Madrid, 22-30.08.2006 short communication.

Let us give an overview of the content of the thesis. Chapter 1 is devoted to the formulation of the inverse problems. In Section 1.1 we pose the direct problems in the cases of boundary conditions of the first and third kind and formulate inverse problems with temperature and flux observations. In the inverse problems we assume the kernels to be degenerate, i.e. representable in the forms (1.1.8), where n_k , $k=1,\ldots,K_1$, and $m_k, k=1,\ldots,K_2$, are K_1+K_2 unknown time-dependent coefficients. In Section 2 we apply the Laplace transform to these problems. This section begins with the Subsection 1.2.1 where certain basic properties of the Laplace transform are listed. Further, in Subsections 1.2.2 and 1.2.3 we rewrite the direct problems in the Laplace domain in the form of system (1.2.15), (1.2.18) containing the Green functions of the elliptic operator of the problem. This system is common both for the boundary conditions of the first and third kind. Finally, in Subsection 1.2.4 we formulate the inverse problems in the Laplace domain and define the generalized solutions of the inverse problems in time domain. The latter ones are simply the inverse Laplace transforms of the solutions of the inverse problems in the Laplace domain.

Chapter 2 plays a preparative role. It starts with Section 2.1 where we collect definitions of the functional spaces used in our analysis. Further, in Section 2.2 basic properties and asymptotical representations of the Green functions are derived. These representations contain solutions of the corresponding Cauchy problems for the elliptic operator. Therefore, a preliminary analysis of such Cauchy problems is provided at the beginning of this section. Section 2.3 includes auxiliary results and Sections 2.4, 2.5 contain further properties of the Green functions. More precisely, in Sections 2.4, 2.5 we give estimates of integrals of Green functions and establish asymptotical behavior of quantities of the type

$$Q = \int_0^1 \mathcal{G}(x, y, p) V(y) dy$$

in the process $\operatorname{Re} p \to +\infty$ where V is a given function and $\mathcal{G} = pG, pG_x, G_{xy}$ with G the Green function. These results are necessary for the analysis of the inverse problems. The estimates of integrals of Green functions are stated in Theorems 2.1 and 2.5. Although these theorems were already proved in [23, 24], in view of the shortness of the proofs, we repeat them. Further, the asymptotics of \mathcal{Q} in the cases $\mathcal{G} = pG, G_{xy}$ was already proved in [23, 24]. Therefore, we cite these results without proofs in the form of Theorems 2.2, 2.3, 2.6, 2.7. The case $\mathcal{G} = pG_x$ is not presented in the literature. We prove the asymptotics in this case in Theorems 2.4 and 2.8. Technically complicated proofs of the latter theorems and the preparation for these proofs fill a large part of Chapter 2.

Chapter 3 contains the study of the inverse problem with temperature observations. In Section 3.1 we reduce this problem to a fixed-point form. To this end, we consider the asymptotics of the problem in the process $\operatorname{Re} p \to +\infty$. This asymptotics gives us a linear system (3.1.3) for the initial values of the functions n_k . This system is overdetermined. For the existence of a solution, the consistency condition (3.1.4) must hold. Assuming this condition, it is possible to extract a

second order asymptotics from the problem to get the desired fixed point system for unknown functions (formulas (3.1.9), (3.1.10)). More precisely, the obtained system is written in terms of the Laplace transforms of the functions n'_k and m_k . The principal part of this system contains a matrix Γ (formula (3.1.7)) that is assumed to be regular. The proved equivalence of the inverse problem in the Laplace domain and the fixed-point system is formulated in Proposition 3.1.

The fixed-point system (3.1.9), (3.1.10) contains operators of the Laplace transforms of the functions n_k' and m_k that correspond to the solution of the direct problem and its derivative. In Section 3.2 auxiliary estimates for these operators are deduced. After that, in Section 3.3 main results about the inverse problem are proved. Firstly, Theorem 3.2 establishes existence and uniqueness for the system (3.1.9), (3.1.10). The proof of this theorem uses fixed-point technique in halfplanes $\operatorname{Re} p > \sigma$ with sufficiently large σ , because the right-hand side of this system is of lower order in the process $\operatorname{Re} p \to +\infty$. Further, in Corollary 3.1 by means of Theorem 3.2 and Proposition 3.1 the existence and uniqueness results are extended to the generalized inverse problem in the time domain. Chapter 3 is finished by Section 3.4 where the assumptions of Theorem 3.2 and Corollary 3.1 are interpreted. Namely, we write sufficient conditions in the time domain for the assumptions of these sentences and give examples when the matrix Γ is regular.

Chapter 4 contains the study of the inverse problem with flux observations. It has the same structure as Chapter 3. In Section 4.1 the inverse problem is transformed to the fixed-point system (4.1.6), (4.1.7) with the matrix Γ of the form (4.1.4), where the unknowns are the laplace transforms of n_k and m_k . Further, in Section 4.2 the operators corresponding to the direct problem are analysed and Section 4.3 contains the main results. In Theorem 4.2 of Section 4.3 we prove the existence and uniqueness for the fixed-point system and in Corollary 4.1 extend these results to the generalized inverse problem in the time domain. Finally, Section 4.4 contains interpretation of assumptions. There sufficient conditions in the time domain for the assumptions of the existence theorem are given and examples for regular Γ are provided. However, it is shown that in some important cases Γ is singular.

Chapter 4 is complemented with the analysis of the case of singular Γ , too. In Section 4.5 we describe the procedure of extracting higher order principal parts from the inverse problems to get the fixed-point systems in cases of singular Γ . However, sometimes even this procedure doesn't work. An example is the problem with flux observations in the homogeneous case. It is shown that then the inverse problem is severely ill-posed.

1. Formulation of direct and inverse problems

1.1 Problems in time domain

In the linear theory of heat flow in a rigid nonhomogeneous bar consisting of a material with thermal memory, the following system of constitutive relations holds [11, 26, 20]

$$e(x,t) = \beta(x)u(x,t) + \int_0^t n(x,t-\tau)u(x,\tau)d\tau,$$
 (1.1.1)

$$q(x,t) = -\lambda(x)u_x(x,t) + \int_0^t m(x,t-\tau)u_x(x,\tau)d\tau.$$
 (1.1.2)

Here u is the temperature that is assumed to be 0 for t < 0, e is the internal energy, q is the heat flux, β is the product of the specific caloric coefficient and the mass density and λ is the heat conduction coefficient. Moreover, n and m are the *memory kernels* of the internal energy and heat flux, respectively. These constitutive relations can be complemented by the heat balance equation

$$e_t(x,t) + q_x(x,t) = r(x,t), \quad x \in (0,1), \ t > 0,$$
 (1.1.3)

with r being the heat supply. We assume the rod to be of the unit length for a sake of simplicity.

Using (1.1.1) and (1.1.2) in (1.1.3) we arrive at the following integro-differential equation of heat conduction:

$$\beta(x)\frac{\partial}{\partial t}u(x,t) + \frac{\partial}{\partial t}\int_{0}^{t}n(x,t-\tau)u(x,\tau)\,d\tau = \frac{\partial}{\partial x}\left(\lambda(x)u_{x}(x,t)\right)$$
$$-\frac{\partial}{\partial x}\int_{0}^{t}m(x,t-\tau)u_{x}(x,\tau)\,d\tau + r(x,t), \quad x \in (0,1), \ t > 0. \quad (1.1.4)$$

We require that the function u(x,t) satisfies the initial condition

$$u(x,0) = \varphi(x), \quad x \in (0,1)$$
 (1.1.5)

and either the boundary conditions of the first kind

$$u(0,t) = f_1(t), \quad u(1,t) = f_2(t), \quad t > 0$$
 (1.1.6)

1. Formulation of direct and inverse problems

or the boundary conditions of the third kind

$$-q(0,t) = \alpha_1(u(0,t) - f_1(t)), \quad q(1,t) = \alpha_2(u(1,t) - f_2(t)), \quad t > 0.$$

Here φ and f_j , j=1,2 are given functions and $\alpha_j \geq 0$, j=1,2 are given constants. In view of (1.1.2) the boundary conditions of the third kind can be rewritten as

$$\lambda(0)u_{x}(0,t) - \int_{0}^{t} m(0,t-\tau)u_{x}(0,\tau) d\tau = \alpha_{1}(u(0,t) - f_{1}(t)),$$

$$-\lambda(1)u_{x}(1,t) + \int_{0}^{t} m(1,t-\tau)u_{x}(1,\tau) d\tau = \alpha_{2}(u(1,t) - f_{2}(t)), \quad t > 0.$$
(1.1.7)

Equation (1.1.4) with the initial condition (1.1.5) and the boundary conditions (1.1.6) or (1.1.7) form the *direct problem* for the temperature u.

In this thesis we will be concerned with the determination of the kernels n and m. We will be limited to the case, when these kernels have the following degenerate forms

$$n(x,t) = \sum_{k=1}^{K_1} \nu_k(x) n_k(t), \quad m(x,t) = \sum_{k=1}^{K_2} \mu_k(x) m_k(t), \quad (1.1.8)$$

where ν_k , $k=1,\ldots,K_1$, μ_k , $k=1,\ldots,K_2$ are given x-dependent functions and n_k , $k=1,\ldots,K_1$, m_k , $k=1,\ldots,K_2$ are unknown time-dependent coefficients.

There are two important cases when the kernels are of the form (1.1.8). In the first case the rod under consideration is piecewise homogeneous. Then denoting by $J_k = (y_{k-1}, y_k)$ with $0 = y_0 < y_1 < \ldots < y_{K_1} = 1$ the homogeneous pieces of such a rod, we can set $K_1 = K_2$ and ν_k , μ_k to be the characteristic function of J_k . However, in our analysis we have to assume smoothness ν_k and μ_k (see the solvability theorems in next chapters). Therefore, in this case we have to set ν_k and μ_k to be some smooth approximation the characteristic function of J_k . For instance, we can define

$$\nu_{k}, \mu_{k} \in C^{l}(\mathbb{R}),$$

$$\nu_{k}(x) = \mu_{k}(x) = \begin{cases} 1 & \text{if } x \in (y_{k-1} + \epsilon, y_{k} - \epsilon) \\ 0 & \text{if } x \notin (y_{k-1} - \epsilon, y_{k} + \epsilon) \end{cases}, k = 2, \dots, K_{1} - 1,$$

$$\nu_{1}(x) = \mu_{1}(x) = \begin{cases} 1 & \text{if } x \in [0, y_{1} - \epsilon) \\ 0 & \text{if } x \notin [0, y_{1} + \epsilon) \end{cases},$$

$$\nu_{K_{1}}(x) = \mu_{K_{1}}(x) = \begin{cases} 1 & \text{if } x \in (y_{K_{1} - 1} + \epsilon, 1] \\ 0 & \text{if } x \notin (y_{K_{1} - 1} - \epsilon, 1] \end{cases}$$
(1.1.9)

with some $l \in \mathbb{N}$ and small $\epsilon > 0$. In the second case the sums in (1.1.8) are certain finite-dimensional approximations of the actual non-degenerate kernels n

and m. Then ν_k and μ_k form a certain basis of x-dependent functions on [0,1]. For instance, ν_k and μ_k could be the power basis: $\nu_k(x) = \mu_k(x) = x^{k-1}$ or the weighted power basis:

$$\nu_k(x) = a(x)x^{k-1}, \ \mu_k(x) = b(x)x^{k-1}.$$

Here a and b are some smooth functions such that |a(x)|, |b(x)| > 0 for any $x \in [0,1]$. The latter one is equivalent to the usual power expansions for the functions $\frac{n(x,t)}{a(x)}$ and $\frac{m(x,t)}{b(x)}$. Trigonometrical expansions could also be used for n and m. In order to determine the unknown functions n_k and m_k we have to specify

In order to determine the unknown functions n_k and m_k we have to specify $K = K_1 + K_2$ additional conditions. We introduce two different types of additional (observation) conditions.

1. Temperature observations

$$u(x_i, t) = h_i(t), \quad t > 0, \quad i = 1, \dots, K.$$
 (1.1.10)

Here x_i are K different points of measurement in (0, 1).

2. Flux observations

$$q(x_i, t) = -\lambda(x_i)u_x(x_i, t) + \int_0^t m(x_i, t - \tau)u_x(x_i, \tau) d\tau = h_i(t), \quad (1.1.11)$$

$$t > 0, \quad i = 1, \dots, K,$$

where x_i are K different points of measurement in (0,1).

Summing up, our inverse problems are as follows.

- 1. Inverse problem with temperature observations. Find n_k , $k = 1, ..., K_1$ and m_k , $k = 1, ..., K_2$ such that the solution u of the direct problem with kernels n and m of the form (1.1.8) satisfies the conditions (1.1.10).
- 2. Inverse problem with flux observations. Find n_k , $k = 1, ..., K_1$ and m_k , $k = 1, ..., K_2$ such that the solution u of the direct problem with kernels n and m of the form (1.1.8) satisfies the conditions (1.1.11).

1.2 Problems in Laplace domain

1.2.1 Laplace transform and its basic properties

In this subsection we collect some basic properties of the Laplace transform that we will use in the sequel. These properties can be found e.g. in the book [8].

The Laplace transform of a function z(t), t > 0, is given by the formula

$$Z(p) = \mathcal{L}_{t \to p} z(t) = \int_0^\infty e^{-pt} z(t) dt, \quad p \in \mathbb{C}.$$
 (1.2.1)

1. Formulation of direct and inverse problems

A sufficient condition of existence for the Laplace transform of z is

$$z \in \mathscr{E} = \{z : z(t) - \text{piecewise continuous in } (0, \infty) \text{ and } |z(t)| \le Ce^{\sigma t}, t \ge 0 \text{ with some } C \ge 0 \text{ and } \sigma \in \mathbb{R}\}.$$
 (1.2.2)

In case $z\in \mathscr{E}$ the transformed function Z(p) is holomorphic in the half-plane $\operatorname{Re} p>\sigma$ and satisfies the condition

$$|Z(p)| \le \frac{C}{\operatorname{Re} p - \sigma} \quad \text{for } \operatorname{Re} p > \sigma.$$
 (1.2.3)

Moreover, in this case the inverse Laplace transform is given by the formula

$$z(t) = \mathcal{L}_{p \to t}^{-1} Z(p) = \frac{1}{2\pi i} \int_{\xi - i\infty}^{\xi + i\infty} e^{tp} Z(p) \, dp, \tag{1.2.4}$$

where ξ is an arbitrary number greater than σ .

Let us list some further properties necessary for our analysis in next sections.

1. Let $z^{(j)} \in \mathcal{E}$, $j = 0, \dots, k$, with some $k \in \mathbb{N}$. Then

$$\mathcal{L}_{t \to p} z^{(k)}(t) = p^k Z(p) - p^{k-1} z(0) - p^{k-2} z'(0) - \dots - z^{(k-1)}(0),$$

where $Z(p) = \mathcal{L}_{t \to p} z(t)$. Thus,

$$Z(p) = \frac{z(0)}{p} + \frac{z'(0)}{p^2} + \ldots + \frac{z^{(k-1)}(0)}{p^{k-1}} + V(p),$$

where by (1.2.3)

$$|V(p)| = \left| \frac{1}{p^k} \mathcal{L}_{t \to p} z^{(k)}(t) \right| \le \frac{C}{|p|^k (\operatorname{Re} p - \sigma)} \text{ for } \operatorname{Re} p > \sigma.$$

2. Let $z_1,z_2\in\mathscr{E}$. Then $\int_0^{\cdot}z_1(\cdot-\tau)z_2(\tau)d\tau\in\mathscr{E}$ and

$$\mathcal{L}_{t \to p} \int_0^t z_1(t - \tau) z_2(\tau) d\tau = Z_1(p) Z_2(p)$$
 (1.2.5)

with $Z_j(p) = \mathcal{L}_{t \to p} z_j(t)$, j = 1, 2.

- 3. Let Z(p) be holomorphic in some half-plane $\operatorname{Re} p > \sigma$ and have the form $Z(p) = \frac{c}{p} + V(p)$ where $|V(p)| \leq \operatorname{Const} |p|^{-\alpha}$ for $\operatorname{Re} p > \sigma$ with $\alpha > 1$. Then there exists $z \in C[0,\infty) \cap \mathscr{E}$ such that $Z(p) = \mathcal{L}_{t \to p} z(t)$. Moreover, z(0) = c.
- 4. The inverse Laplace transform is unique. Namely, let the Laplace transforms $Z_j(p)$ of functions $z_j \in \mathscr{E}$, j=1,2 satisfy the equality $Z_1(p)=Z_2(p)$ for $\operatorname{Re} p > \sigma$. Then $z_1(t)=z_2(t)$ a.e. $t\in (0,\infty)$.

Let us deduce a further property. This is related to the fractional derivative of a function z(t) defined by the formula

$$\frac{d^{\varkappa}}{dt^{\varkappa}}z(t) = \frac{d}{dt}\frac{1}{\Gamma(1-\varkappa)}\int_0^t (t-\tau)^{-\varkappa}z(\tau)d\tau, \quad 0 < \varkappa < 1.$$

From (1.2.1) we easily derive the relation $\mathcal{L}_{t\to p}\frac{1}{\Gamma(1-\varkappa)}t^{-\varkappa}=p^{\varkappa-1}$ for $\operatorname{Re} p>0$ containing the main branch of the power function. Thus, from the properties 1, 2 and (1.2.3) we infer

5. Let
$$v, \frac{d^{\varkappa}}{dt^{\varkappa}}v \in \mathscr{E}$$
 with some $\varkappa \in (0,1)$. Then $\mathcal{L}_{t \to p} \frac{d^{\varkappa}}{dt^{\varkappa}}v(t) = p^{\varkappa}V(p)$ and
$$|V(p)| \leq \frac{C}{|p|^{\varkappa}(\operatorname{Re} p - \sigma)} \quad \text{for} \quad \operatorname{Re} p > \sigma.$$

1.2.2 Direct problem in the case of boundary conditions of the first kind

Assume a priori that

$$n_k, m_k, r, f_1, f_2$$
 as functions of t belong to \mathscr{E} (1.2.6)

and suppose that u solving the direct problem has the properties

$$U, U_x, U_t, U_{xx}, U_{tt}$$
 as functions of t belong to \mathscr{E} . (1.2.7)

Let us apply the Laplace transform to the equation (1.1.4). Denoting

$$U = \mathcal{L}_{t \to p} u$$
, $R = \mathcal{L}_{t \to p} r$, $M_k = \mathcal{L}_{t \to p} m_k$, $N_k = \mathcal{L}_{t \to p} n_k$,

observing the initial condition (1.1.5), the representation of the kernels (1.1.8) and the properties 1 and 2 of the Laplace transform from Section 1.2.1 we obtain

$$\beta(x) \left[pU(x,p) - \varphi(x) \right] + p \sum_{k=1}^{K_1} N_k(p) \nu_k(x) U(x,p)$$
 (1.2.8)

$$= \frac{\partial}{\partial x} \left(\lambda(x) U_x(x, p) \right) - \sum_{k=1}^{K_2} M_k(p) \frac{\partial}{\partial x} \left(\mu_k(x) U_x(x, p) \right) + R(x, p).$$

The first kind boundary conditions (1.1.6) are transformed to

$$U(0,p) = F_1(p), \quad U(1,p) = F_2(p).$$
 (1.2.9)

Here $F_j = \mathcal{L}_{t \to p} f_j$, j = 1, 2.

We are going to reduce the direct problem for U into a system of integral equations. To this end we represent the equation (1.2.8) in the form

$$(LU)(x,p) = p \sum_{k=1}^{K_1} N_k(p) \nu_k(x) U(x,p)$$

$$+ \sum_{k=1}^{K_2} M_k(p) \frac{\partial}{\partial x} (\mu_k(x) U_x(x,p)) - R(x,p) - \beta(x) \varphi(x)$$

$$(1.2.10)$$

1. Formulation of direct and inverse problems

with the differential operator

$$(LU)(x,p) = \frac{\partial}{\partial x} (\lambda(x)U_x(x,p)) - \beta(x)pU(x,p), \quad x \in (0,1).$$
 (1.2.11)

Further, we introduce the green function G(x, y, p) of the operator L corresponding to the first kind boundary conditions. This is the function satisfying the problem

$$L_y G(x, y, p) = \delta(y - x), \quad x \in (0, 1), y \in (0, 1),$$

$$G(x, 0, p) = G(x, 1, p) = 0,$$
(1.2.12)

where L_y stands for the operator L with respect to the variable y.

Then, the solution of (1.2.8), (1.2.9) is given by

$$U(x,p) = \sum_{k=1}^{K_1} N_k(p) \int_0^1 G(x,y,p) \nu_k(y) p U(y,p) \, dy$$

$$+ \sum_{k=1}^{K_2} M_k(p) \int_0^1 G(x,y,p) \frac{\partial}{\partial y} (\mu_k(y) U_y(y,p)) \, dy - Q(x,p),$$
(1.2.13)

where

$$Q(x,p) = \int_{0}^{1} G(x,y,p) \left[\beta(y)\varphi(y) + R(y,p) \right] dy$$

$$+ \lambda(0)G_{y}(x,0,p)F_{1}(p) - \lambda(1)G_{y}(x,1,p)F_{2}(p) .$$
(1.2.14)

Integrating the integrals in the second sum of (1.2.13) by parts and observing the homogeneous boundary conditions of G we obtain the following equation for U:

$$U(x,p) = \sum_{k=1}^{K_1} N_k(p) \int_0^1 pG(x,y,p) \nu_k(y) U(y,p) dy$$

$$- \sum_{k=1}^{K_2} M_k(p) \int_0^1 G_y(x,y,p) \mu_k(y) U_y(y,p) dy - Q(x,p).$$
(1.2.15)

It contains U_x in the right-hand side. Therefore, we have to derive an additional equation for U_x , too. To this end we differentiate (1.2.13) with respect to x:

$$U_{x}(x,p) = \sum_{k=1}^{K_{1}} N_{k}(p) \int_{0}^{1} pG_{x}(x,y,p) \nu_{k}(y) U(y,p) dy$$

$$+ \sum_{k=1}^{K_{2}} M_{k}(p) \int_{0}^{1} G_{x}(x,y,p) \frac{\partial}{\partial y} (\mu_{k}(y) U_{y}(y,p)) dy - Q_{x}(x,p).$$
(1.2.16)

Thereupon we split the second integral in (1.2.16) into two parts, from 0 to x and from x to 1, and integrate them by parts. Taking into consideration the equalities $G_x(x,0,p)=G_x(x,1,p)=0,\ 0< x<1$, following from (1.2.12), and the jump relation

$$G_x(x, x - 0, p) - G_x(x, x + 0, p) = \frac{1}{\lambda(x)}, \quad 0 < x < 1$$
 (1.2.17)

(see [39], p. 169) we obtain the following equation for U_x :

$$U_{x}(x,p) = \frac{1}{\lambda(x)} \sum_{k=1}^{K_{2}} M_{k}(p)\mu_{k}(x)U_{x}(x,p)$$

$$+ \sum_{k=1}^{K_{1}} N_{k}(p) \int_{0}^{1} pG_{x}(x,y,p)\nu_{k}(y)U(y,p) dy \qquad (1.2.18)$$

$$- \sum_{k=1}^{K_{2}} M_{k}(p) \int_{0}^{1} G_{xy}(x,y,p)\mu_{k}(y)U_{y}(y,p) dy - Q_{x}(x,p).$$

1.2.3 Direct problem in the case of boundary conditions of the third kind

As in the previous subsection we assume (1.2.6), (1.2.7), deduce the equation (1.2.8) in the Laplace domain and rewrite it in the form (1.2.10) where L is given by (1.2.11). The third kind boundary conditions (1.1.7) are transformed to

$$\lambda(0)U_{x}(0,p) = \alpha_{1}[U(0,p) - F_{1}(p)] + \sum_{k=1}^{K_{2}} \mu_{k}(0)M_{k}(p)U_{x}(0,p) -\lambda(1)U_{x}(1,p) = \alpha_{2}[U(1,p) - F_{2}(p)] - \sum_{k=1}^{K_{2}} \mu_{k}(1)M_{k}(p)U_{x}(1,p),$$
(1.2.19)

with
$$F_j = \mathcal{L}_{t \to p} f_j$$
, $j = 1, 2$.

Further, let us denote by G(x, y, p) the Green function of operator L with the third kind boundary conditions, i.e.,

$$L_y G(x, y, p) = \delta(y - x), \quad x \in (0, 1), \ y \in (0, 1),$$

$$\lambda(0) G_y(x, 1, p) = \alpha_1 G(x, 0, p), \ -\lambda(1) G_y(x, 0, p) = \alpha_2 G(x, 1, p), \ x \in (0, 1),$$
(1.2.20)

where L_y is the operator L with respect to the variable y, as before.

Then, according to the Green representation in the case of the third kind boundary conditions (see [24]), the solution of (1.2.8) is given by

$$U(x,p) = p \sum_{k=1}^{K_1} N_k(p) \int_0^1 G(x,y,p) \nu_k(y) U(y,p) dy$$

$$+ \sum_{k=1}^{K_2} M_k(p) \int_0^1 G(x,y,p) \frac{\partial}{\partial y} (\mu_k(y) U_y(y,p)) dy - F(x,p),$$
(1.2.21)

1. Formulation of direct and inverse problems

where

$$F(x,p) = \int_{0}^{1} G(x,y,p) \left[\beta(y)\varphi(y) + R(y,p) \right] dy$$

+G(x,1,p)[\lambda(1)U_x(1,p) + \alpha_2U(1,p)] - G(x,0,p)[\lambda(0)U_x(0,p) - \alpha_1U(0,p)].

Due to (1.2.19), the latter formula can be rewritten in the form

$$F(x,p) = \int_{0}^{1} G(x,y,p) \left[\beta(y)\varphi(y) + R(y,p)\right] dy$$

$$+ \left(\alpha_{1}F_{1}(p) - \sum_{k=1}^{K_{2}} \mu_{k}(0)M_{k}(p)U_{x}(0,p)\right) G(0,x,p)$$

$$+ \left(\alpha_{2}F_{2}(p) + \sum_{k=1}^{K_{2}} \mu_{k}(1)M_{k}(p)U_{x}(1,p)\right) G(1,x,p) .$$

$$(1.2.22)$$

Integrating the integrals in the second sum of (1.2.21) by parts and using (1.2.22) as well as the symmetry relations G(x,1,p) = G(1,x,p), G(x,0,p) = G(0,x,p), we obtain the equation (1.2.15) for U, where

$$Q(x,p) = \int_{0}^{1} G(x,y,p) \left[\beta(y)\varphi(y) + R(y,p) \right] dy$$

$$+ \alpha_{1}F_{1}(p)G(0,x,p) + \alpha_{2}F_{2}(p)G(1,x,p).$$
(1.2.23)

To derive the additional equation for U_x we differentiate (1.2.21) with respect to x:

$$U_{x}(x,p) = \sum_{k=1}^{K_{1}} p N_{k}(p) \int_{0}^{1} G_{x}(x,y,p) \nu_{k}(y) U(y,p) dy$$

$$+ \sum_{k=1}^{K_{2}} M_{k}(p) \int_{0}^{1} G_{x}(x,y,p) \frac{\partial}{\partial y} (\mu_{k}(y) U_{y}(y,p)) dy - F_{x}(x,p).$$
(1.2.24)

As in the previous subsection we split the second integral in (1.2.21) into two parts, from 0 to x and from x to 1, and integrate them by parts. Noting that for the Green function with the third kind boundary conditions the same jump relation (1.2.17) holds as for the Green function with the first kind boundary conditions

(see [24]), we get

$$\begin{split} U_x(x,p) &= \frac{1}{\lambda(x)} \sum_{k=1}^{K_2} M_k(p) \mu_k(x) U_x(x,p) \\ &+ \sum_{k=1}^{K_1} p N_k(p) \int_0^1 G_x(x,y,p) \nu_k(y) U(y,p) \, dy \\ &- \sum_{k=1}^{K_2} M_k(p) \Biggl(\int_0^1 (G_{xy}(x,y,p) \mu_k(y) U_y(y,p) \, dy - \mu_k(1) U_x(1,p) G_x(x,1,p) \\ &+ \mu_k(0) U_x(0,p) G_x(x,0,p) \Biggr) - F_x(x,p). \end{split}$$

Replacing here F_x by (1.2.22) and observing (1.2.23) as well as the symmetry relations G(x, 1, p) = G(1, x, p), G(x, 0, p) = G(0, x, p), again, we arrive at the equation (1.2.18) for U_x .

1.2.4 Summary. Generalized formulation of inverse problems

Summing up, the direct problem in the Laplace domain is reduced to the system of equations (1.2.15) and (1.2.18) for the Laplace transform of the temperature function U(x,p). In the case of boundary conditions of the first kind this system contains the Green function corresponding to the first kind boundary conditions and the function Q is given by (1.2.14). In the case of boundary conditions of the third kind this system contains the Green function corresponding to the third kind boundary conditions and the function Q is given by (1.2.23).

The systems for U in the cases of the first and third kind boundary conditions have the same form. The difference occurs only in the functions G and Q. This similarity enables to treat these systems in a common form in the next chapters.

Assuming that h_i are measurable and exponentially bounded, the observation conditions (1.1.10) and (1.1.11) in the Laplace domain take the forms

$$U(x_i, p) = H_i(p), \quad i = 1, ..., K$$
 (1.2.25)

and

$$-\lambda(x_i)U_x(x_i,p) + \sum_{k=1}^{K_2} M_k(p)\mu_k(x_i)U_x(x_i,p) = H_i(p), \ i = 1,\dots,K, \ (1.2.26)$$

respectively, where $H_i(p) = \mathcal{L}_{t \to p} h_i$.

Thus, the inverse problems in the Laplace domain are as follows.

1. Inverse problem with temperature observations in the Laplace domain. Find the functions N_k , $k=1,\ldots,K_1$ and M_k , $k=1,\ldots,K_2$ such that the solution U of (1.2.15), (1.2.18) satisfies (1.2.25).

- 1. Formulation of direct and inverse problems
- 2. Inverse problem with flux observations in the Laplace domain. Find the functions N_k , $k = 1, ..., K_1$ and M_k , $k = 1, ..., K_2$ such that the solution U of (1.2.15), (1.2.18) satisfies (1.2.26).

On the basis of the inverse problems in the Laplace domain we can formulate the generalized inverse problems in time domain.

- 1. Generalized inverse problem with temperature observations in the time domain. Find t-dependent functions $n_k, k = 1, \ldots, K_1$ and $m_k, k = 1, \ldots, K_2$ such that their Laplace transforms satisfy the inverse problem with temperature observations in the Laplace domain.
- 2. Generalized inverse problem with flux observations in the time domain. Find t-dependent functions n_k , $k=1,\ldots,K_1$ and m_k , $k=1,\ldots,K_2$ such that their Laplace transforms satisfy the inverse problem with flux observations in the Laplace domain.

From the discussions of this chapter it follows, that in case of sufficiently regular data which guarantee the conditions (1.2.7) for the solution u of the direct problem, any solution n_k , $k=1,\ldots,K_1$, m_k , $k=1,\ldots,K_2$ of an inverse problems posed in Section 1.1 solves the corresponding generalized inverse problem, too. The converse assertion is not right, if we consider the direct problem formulated in Section 1.1 in the classical sense. Namely, the system (1.2.15), (1.2.18) related to the generalized inverse problems doesn't require the existence of second derivative U_{xx} in contrast to the equation (1.1.4) related to the classical inverse problems.

2.1 Functional spaces

To analyse the direct and inverse problems we define the spaces

$$\mathcal{A}_{\gamma,\sigma} = \{V : V(p) \text{ is holomorphic on } \operatorname{Re} p > \sigma, \|V\|_{\gamma,\sigma} < \infty\},$$
 (2.1.1)

 $\gamma, \sigma \geq 0$, where

$$||V||_{\gamma,\sigma} = \sup_{\text{Re}p>\sigma} |p|^{\gamma} |V(p)|$$

and

$$(\mathcal{A}_{\gamma,\sigma})^K = \{ V = (V_1, \dots, V_K) : V_k(p) \in \mathcal{A}_{\gamma,\sigma}, k = 1, \dots, K \}$$
 (2.1.2)

with the norms

$$||V||_{\gamma,\sigma} = \sum_{k=1}^{K} ||V_k||_{\gamma,\sigma}, \quad V \in (\mathcal{A}_{\gamma,\sigma})^K.$$

These spaces are complete due to the completeness in supremum-norm of sets of holomorphic functions. Thus, they are Banach spaces. Moreover, we note that $\mathcal{A}_{\gamma,\sigma} \subset \mathcal{A}_{\gamma,\sigma'}, \ (\mathcal{A}_{\gamma,\sigma})^K \subset \left(\mathcal{A}_{\gamma,\sigma'}\right)^K$ and $\|\cdot\|_{\gamma,\sigma'} \leq \|\cdot\|_{\gamma,\sigma}$ if $\sigma' > \sigma$. Further, let us introduce the following spaces of K-component vector functions.

Further, let us introduce the following spaces of K-component vector functions:

$$\mathcal{M}_{c,\alpha,\sigma} = \left\{ Z : Z = \frac{c}{p} + V(p), V \in (\mathcal{A}_{\alpha,\sigma})^K \right\}.$$

Here $c=(c_1,\ldots,c_K)\in\mathbb{R}^K$ is a given constant vector and $\alpha>1$ is a fixed number. We remark that any vector function $Z\in\mathcal{M}_{c,\alpha,\sigma}$ has the unique original $z(t)=\mathcal{L}_{p\to t}^{-1}Z(p)$ in the time domain which is continuous for $t\in[0,\infty)$ and z(0)=c (see property 3 in Section 1.2.1).

In the next chapters we will seek for the solutions of the inverse problems such that the vector $(Z_k|_{k=1,\dots,K_1}, M_k|_{k=1,\dots,K_2})$ with Z_k being either N_k or a Laplace transform of a derivative n_k belongs to $\mathcal{M}_{c,\alpha,\sigma}$. We will always assume that the parameter α in the definition of $\mathcal{M}_{c,\alpha,\sigma}$ satisfies the condition

$$1 < \alpha < \frac{3}{2} \,. \tag{2.1.3}$$

Recall that in the previous chapter we reduced the direct problem to a system of integral equations for the pair $(U(x,p),U_x(x,p))$. Therefore, we need suitable

spaces for pairs of x- and p-dependent functions. Let us define the following Banach spaces of single functions

$$\hat{\mathcal{B}}_{\gamma,\sigma} = \left\{ F(x,p) : F(x,\cdot) \in \mathcal{A}_{\gamma,\sigma} \text{ for } x \in [0,1], \ F(\cdot,p) \in C[0,1] \text{ for } \mathrm{Re}p > \sigma \right\},$$
 where $\gamma,\sigma \geq 0$ with the norms

$$||F||_{\gamma,\sigma} = \max_{0 \le x \le 1} \sup_{\text{Re} p > \sigma} |p|^{\gamma} |F(x,p)|, \quad F \in \hat{\mathcal{B}}_{\gamma,\sigma},$$

and the Banach spaces of pairs of functions:

$$\mathcal{B}_{\gamma,\sigma} = \hat{\mathcal{B}}_{\gamma,\sigma} imes \hat{\mathcal{B}}_{\gamma-rac{1}{2},\sigma}\,, \quad \gamma \geq rac{1}{2}\,, \ \ \sigma \geq 0,$$

with the norms

$$||F||_{\gamma,\sigma} = ||F_1||_{\gamma,\sigma} + ||F_2||_{\gamma-\frac{1}{2},\sigma}, \quad F = (F_1, F_2) \in \mathcal{B}_{\gamma,\sigma}.$$
 (2.1.4)

2.2 Asymptotical representation of Green function.

2.2.1 Asymptotical representation of a solution of Cauchy problem

In this section we prove asymptotical properties of the solution of the Cauchy problem for the operator L defined by (1.2.11). Let $\psi = \psi(x,p)$ solve the Cauchy problem

$$(L\psi)(x,p) = 0, \ x \in (0,1), \ \psi(0,p) = \theta_0, \ \psi_x(0,p) = \theta_1$$
 (2.2.1)

with the differential operator L defined in (1.2.11) and some given numbers $\theta_0, \theta_1 \in \mathbb{C}$ independent of p. If $\lambda \in C^1[0,1], \ \beta \in C[0,1]$ and $\lambda(x)>0, \ x \in [0,1]$, then by the Cauchy theorem the problem (2.2.1) has the unique solution $\psi(\cdot,p) \in C^2[0,1]$.

Lemma 2.1. Let

$$\lambda, \beta \in C^2[0,1], \ \lambda(x), \beta(x) > 0, \ x \in [0,1].$$
 (2.2.2)

Then the solution $\psi(x,p)$ of (2.2.1) and its derivative $\psi_x(x,p)$ are holomorphic in $\operatorname{Re} p > 0$ for any $x \in [0,1]$. Moreover, the following asymptotical relations are valid:

$$\psi(x,p) = \frac{a(0)}{a(x)} \theta_0 ch sz + \frac{a(0)}{a(x)} \left(\frac{\zeta(x)}{2} + l\bar{b}(0)a'(0) \right) \theta_0 \frac{1}{s} sh sz + \frac{l}{a(x)b(0)} \theta_1 \frac{1}{s} sh sz + \frac{l}{2a(x)b(0)} \theta_1 \frac{1}{s^2} ch sz + \left(|\theta_0| + \frac{|\theta_1|}{|s|} \right) O\left(\frac{e^{sz}}{s^2} \right) ,$$

$$\psi_x(x,p) = \frac{b(x)a(0)}{l} \theta_0 s sh sz + \frac{b(x)a(0)}{l} \left(\frac{\zeta(x)}{2} + l\bar{b}(0)a'(0) - l\bar{b}(x)a'(x) \right) \theta_0 ch sz + \frac{b(x)}{b(0)} \theta_1 ch sz + \frac{b(x)}{b(0)} \left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) \right) \theta_1 \frac{1}{s} sh sz + \left(|\theta_0| + \frac{|\theta_1|}{|s|} \right) O\left(\frac{e^{sz}}{s} \right)$$

$$for Re p \to +\infty \quad uniformly \text{ with respect to } x \in [0,1] \quad and \quad \text{Im } p.$$

Here

$$l = \int_0^1 \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} \, d\tau \,, \tag{2.2.4}$$

 $s = l\sqrt{p}$ for $\operatorname{Re} p > 0$, where \sqrt{p} is the main branch of square root, i.e. it satisfies $\sqrt{p} > 0$ for real p > 0.

$$a(x) = (\beta(x)\lambda(x))^{1/4}, b(x) = \beta(x)^{1/4}\lambda(x)^{-3/4}, \bar{b}(x) = \lambda(x)^{1/4}\beta(x)^{-3/4}, (2.2.6)$$

$$z = z(x) = \frac{1}{l} \int_0^x \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau$$
 (2.2.7)

and

$$\zeta(x) = \frac{1}{l} \int_0^x a_1(\tau) \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau$$
 (2.2.8)

with

$$a_1(x) = \frac{l^2}{\beta(x)} \left[a(x)\lambda(x) \frac{d}{dx} \left(\frac{a'(x)}{a^2(x)} \right) + \frac{\lambda'(x)a'(x)}{a(x)} \right]. \tag{2.2.9}$$

Proof. We apply the Liouville transform replacing the argument x by z using the formula (2.2.7) and the unknown ψ by v using the relation

$$v(z,p) = a(x)\psi(x,p),$$
 (2.2.10)

where a(x) is given by (2.2.6). Then the equation $L\psi=0$ with $x\in[0,1]$ is equivalent to the equation

$$v_{zz}(z,p) - s^2 v(z,p) = c(z)v(z,p), z \in [0,1],$$
 (2.2.11)

where $c(z) = a_1(x)$ with a_1 defined in (2.2.9). The assumptions (2.2.2) imply $c \in C[0,1]$. Note that (2.2.10) with (2.2.6) and (2.2.7) yields the formulas

$$\psi(x,p) = \frac{1}{a(x)}v(z,p), \quad \psi_x(x,p) = \frac{b(x)}{l}v_z(z,p) - \frac{a'(x)}{a^2(x)}v(z,p).$$
 (2.2.12)

Thus, the initial conditions $\psi(0,p) = \theta_0$, $\psi_x(0,p) = \theta_1$ in terms of v take the form

$$v(0,p) = \kappa_0 := a(0)\theta_0, \ v_z(0,p) = \kappa_1 := \frac{l}{b(0)} \left(\frac{a'(0)}{a(0)} \theta_0 + \theta_1 \right).$$
 (2.2.13)

Let us solve the equation (2.2.11) with respect to the left-hand side subject to the initial conditions (2.2.13). We obtain the following Volterra integral equation of the second kind for v:

$$v(z,p) = \kappa_0 ch sz + \kappa_1 \frac{1}{s} sh sz + \frac{1}{s} \int_0^z sh s(z-\tau) c(\tau) v(\tau,p) d\tau,$$

 $z \in [0,1].$ (2.2.14)

We are going to prove the assertions of the lemma by means of this relation and the connections (2.2.12) between ψ and v.

We start by proving the holomorphy assertion. Since (2.2.14) is a Volterra equation of the first kind with a bounded kernel, its solution can be expressed in terms of the Neumann series

$$v(z,p) = f(z,p) + \sum_{i=1}^{\infty} \int_{0}^{z} g_{i}(z,\tau,p) f(\tau,p) d\tau, \qquad (2.2.15)$$

where

$$f(z,p) = \kappa_0 ch sz + \kappa_1 \frac{1}{s} sh sz, \quad g_1(z,\tau,p) = \frac{1}{s} sh s(z-\tau) c(\tau),$$

$$g_i(z,\tau,p) = \int_{\tau}^{z} g_1(z,y,p) g_{i-1}(y,\tau,p) dy, \quad i = 2,3,\dots.$$
(2.2.16)

Let us fix some $0 < \gamma_1 < \gamma_2$. Observing the definition of s (2.2.5) and (2.2.16) we see that

$$\sup_{\substack{0 \leq z \leq 1 \\ \gamma_1 < \operatorname{Re}\sqrt{p} < \gamma_2}} |f(z,p)| = T_f < \infty, \ \sup_{\substack{0 \leq z \leq 1 \\ \gamma_1 < \operatorname{Re}\sqrt{p} < \gamma_2}} |g_1(z,p)| = T_g < \infty.$$

Making use of the standard technique of estimation of the Neumann series we deduce the following estimate for the remainder of this series:

$$\sup_{\substack{0 \le z \le 1 \\ \gamma_1 \le \text{Re } \sqrt{p} \le \gamma_2}} \left| \sum_{i=l+1}^{\infty} \int_0^z g_i(z,\tau,p) f(\tau,p) d\tau \right| \le T_f \sum_{i=l+1}^{\infty} \frac{T_g^i}{i!} \to 0 \quad \text{as} \quad l \to +\infty.$$

This estimate shows that the series (2.2.15) is uniformly convergent in the set $0 \le z \le 1$, $\gamma_1 < \operatorname{Re} \sqrt{p} < \gamma_2$. Further, observing (2.2.16) we see that every addend in (2.2.15) is a holomorphic function in $\gamma_1 < \operatorname{Re} \sqrt{p} < \gamma_2$ for any $z \in [0,1]$. Since a limit of uniformly convergent sequence of holomorphic functions is also holomorphic, the function v(z,p) is holomorphic in $\gamma_1 < \operatorname{Re} \sqrt{p} < \gamma_2$ for any $z \in [0,1]$. From (2.2.14) we have the relation for the derivative

$$v_z(z,p) = \kappa_0 s \, sh \, sz + \kappa_1 ch \, sz + \int_0^z ch \, s(z-\tau) \, c(\tau) v(\tau,p) d\tau, \, z \in [0,1].$$

By the proven holomorphy of v and other terms in the right-hand side of this formula, the function $v_z(z,p)$ is holomorphic in $\gamma_1<\mathrm{Re}\,\sqrt{p}<\gamma_2$ for any $z\in[0,1]$. Finally, since $0<\gamma_1<\gamma_2$ where chosen arbitrarily and $\{p:\mathrm{Re}\,p>0\}\subset\{p:$

Re $\sqrt{p} > 0$ }, by the proven holomorphy of v and v_z and the relation (2.2.12) we get the holomorphy of $\psi(x, p)$ and $\psi_x(x, p)$ in Re p > 0 for any $x \in [0, 1]$.

Let us continue proving the relations (2.2.3). To this end we derive an estimate for v. Multiplying (2.2.14) by e^{-sz} we have

$$e^{-sz}v(z,p) = \kappa_0 e^{-sz} ch \, sz + \kappa_1 \frac{1}{s} e^{-sz} sh \, sz + \frac{1}{s} \int_0^z e^{-s(z-\tau)} sh \, s(z-\tau) \, c(\tau) e^{-s\tau} v(\tau,p) d\tau$$
(2.2.17)

for $z \in [0, 1]$. Making use of the elementary relations

$$|e^{sz}| = e^{\operatorname{Re} sz}, |ch sz| \le ch (\operatorname{Re} sz), |sh sz| \le ch (\operatorname{Re} sz)$$
 (2.2.18)

and denoting $||v||_z = \max_{\tau \in [0,z]} |e^{-s\tau}v(\tau,p)|$ from (2.2.17) we get

$$|e^{-sz}v(z,p)| \leq \left(|\kappa_{0}| + \frac{|\kappa_{1}|}{|s|}\right) e^{-\operatorname{Re} sz} ch \left(\operatorname{Re} sz\right)$$

$$+ \frac{\|c\|_{C[0,1]}}{|s|} \int_{0}^{z} e^{-\operatorname{Re} s(z-\tau)} ch \left(\operatorname{Re} s(z-\tau)\right) d\tau \|v\|_{z} = \left(|\kappa_{0}| + \frac{|\kappa_{1}|}{|s|}\right)$$

$$\times e^{-\operatorname{Re} sz} ch \left(\operatorname{Re} sz\right) + \frac{\|c\|_{C[0,1]}}{2|s|} \left[1 + \frac{1}{2\operatorname{Re} s} \left(1 - e^{-2\operatorname{Re} s}\right)\right] \|v\|_{z}.$$
(2.2.19)

Note that due to the inequality

$$\operatorname{Re}\sqrt{p} > \frac{1}{\sqrt{2}}\sqrt{|p|} \quad \text{for} \quad \operatorname{Re}p > 0$$
 (2.2.20)

and the definition (2.2.5) of s, the formula

$$|s| \ge \operatorname{Re} s > \frac{|s|}{\sqrt{2}} = \frac{l}{\sqrt{2}} \sqrt{|p|} \ge \frac{l}{\sqrt{2}} \sqrt{\operatorname{Re} p} \quad \text{for} \quad \operatorname{Re} p > 0 \quad (2.2.21)$$

is valid. Thus, from (2.2.19) in case of a sufficiently large $\sigma_c > 0$ depending on c we obtain

$$|e^{-sz}v(z,p)| \le 2\left(|\kappa_0| + \frac{|\kappa_1|}{|s|}\right) + \frac{1}{2}||v||_z \text{ for } z \in [0,1], \text{ Re } p > \sigma_c.$$

This implies $\|v\|_z=\max_{\tau\in[0,z]}|e^{-s\tau}v(\tau,p)|\leq 2\left(|\kappa_0|+\frac{|\kappa_1|}{|s|}\right)+\frac{1}{2}\|v\|_z$, hence $\|v\|_z=\max_{\tau\in[0,z]}|e^{-s\tau}v(\tau,p)|\leq 4\left(|\kappa_0|+\frac{|\kappa_1|}{|s|}\right)$ for $z\in[0,1]$ and $\operatorname{Re} p>\sigma_c$. Thus, we arrive at the estimate

$$|v(z,p)| \le 4\left(|\kappa_0| + \frac{|\kappa_1|}{|s|}\right)e^{\operatorname{Re} sz} \quad \text{for } z \in [0,1], \ \operatorname{Re} p > \sigma_c. \quad (2.2.22)$$

Now we are ready to prove (2.2.3). To this end we plug the iterative formula (2.2.14) for v into itself to get

$$v(z,p) = \kappa_0 ch sz + \frac{\kappa_1}{s} sh sz + \frac{\kappa_0}{s} \underbrace{\int_0^z sh s(z-\tau) c(\tau) ch s\tau d\tau}_{I_1} + \frac{\kappa_1}{s^2} \underbrace{\int_0^z sh s(z-\tau) c(\tau) sh s\tau d\tau}_{I_2} + \frac{1}{s^2} \underbrace{\int_0^z sh s(z-\tau) c(\tau) \int_0^\tau sh s(\tau-y) c(y) v(y,p) dy d\tau}_{I_3}.$$
(2.2.23)

Differentiating this relation we deduce the formula for the derivative of v, too:

$$v_{z}(z,p) = \kappa_{0} s sh sz + \kappa_{1} ch sz + \kappa_{0} \underbrace{\int_{0}^{z} ch s(z-\tau) c(\tau) ch s\tau d\tau}_{I_{4}} + \frac{\kappa_{1}}{s} \underbrace{\int_{0}^{z} ch s(z-\tau) c(\tau) sh s\tau d\tau}_{I_{5}} + \frac{1}{s} \underbrace{\int_{0}^{z} ch s(z-\tau) c(\tau) \int_{0}^{\tau} sh s(\tau-y) c(y) v(y,p) dy d\tau}_{I_{6}}.$$

$$(2.2.24)$$

Let us represent I_1 , I_2 , I_4 and I_5 in the form

$$I_{1} = \frac{1}{2} \int_{0}^{z} c(\tau)d\tau \, sh \, sz \, + \frac{1}{2} \underbrace{\int_{0}^{z} sh \, s(z - 2\tau) \, c(\tau)d\tau}_{I_{2}}, \qquad (2.2.25)$$

$$I_2 = \frac{1}{2} \int_0^z c(\tau) d\tau \, ch \, sz \, -\frac{1}{2} \underbrace{\int_0^z ch \, s(z - 2\tau)(\tau) d\tau}_{I_2}, \qquad (2.2.26)$$

$$I_4 = \frac{1}{2} \int_0^z c(\tau) d\tau \, ch \, sz \, + \, \frac{1}{4} I_8 \,, \quad I_5 = \frac{1}{2} \int_0^z c(\tau) d\tau \, sh \, sz \, - \, \frac{1}{4} I_7 \,.$$
 (2.2.27)

Observing (2.2.18) and (2.2.21) we estimate:

$$|I_{7}|, |I_{8}| \leq ||c||_{C[0,1]} \int_{0}^{z} \left[e^{\operatorname{Re} s(z-2\tau)} + e^{\operatorname{Re} s(2\tau-z)} \right] d\tau$$

$$= \frac{2||c||_{C[0,1]}}{\operatorname{Re} s} sh(\operatorname{Re} sz) \leq \frac{\operatorname{Const}}{|s|} e^{\operatorname{Re} sz}$$
(2.2.28)

for any $z \in [0, 1]$ and Re p > 0. Moreover, due to (2.2.18), (2.2.21) and (2.2.22),

we get

$$|I_{3}|, |I_{6}| \leq 4 \|c\|_{C[0,1]}^{2} \left(|\kappa_{0}| + \frac{|\kappa_{1}|}{|s|} \right)$$

$$\times \int_{0}^{z} ch \left(\operatorname{Re} s(z - \tau) \right) \int_{0}^{\tau} ch \left(\operatorname{Re} s(\tau - y) \right) e^{\operatorname{Re} sy} dy d\tau$$

$$= 4 \|c\|_{C[0,1]}^{2} \left(|\kappa_{0}| + \frac{|\kappa_{1}|}{|s|} \right) \frac{\operatorname{Re} sz(1 + \operatorname{Re} sz) e^{\operatorname{Re} sz} + (2\operatorname{Re} sz - 1) sh \left(\operatorname{Re} sz \right)}{8(\operatorname{Re} s)^{2}}$$

$$\leq \operatorname{Const} \left(|\kappa_{0}| + \frac{|\kappa_{1}|}{|s|} \right) e^{\operatorname{Re} sz}$$

$$(2.2.29)$$

for any $z \in [0,1]$ and ${\rm Re} \, p > \sigma_c$. Using (2.2.28) in (2.2.25) - (2.2.27) and thereupon (2.2.25) - (2.2.27), (2.2.29) in (2.2.23), (2.2.24) we deduce the following asymptotical relations for v and v_z :

$$v(z,p) = \kappa_0 \left(ch \, sz + \frac{1}{2} \int_0^z c(\tau) d\tau \frac{1}{s} \, sh \, sz \right)$$

$$+ \kappa_1 \left(\frac{1}{s} sh \, sz + \frac{1}{2} \int_0^z c(\tau) d\tau \frac{1}{s^2} \, ch \, sz \right) + \left(|\kappa_0| + \frac{|\kappa_1|}{|s|} \right) O\left(\frac{e^{sz}}{s^2} \right) ,$$

$$v_z(z,p) = \kappa_0 \left(s \, sh \, sz + \frac{1}{2} \int_0^z c(\tau) d\tau \, ch \, sz \right)$$

$$+ \kappa_1 \left(ch \, sz + \frac{1}{2} \int_0^z c(\tau) d\tau \frac{1}{s} \, sh \, sz \right) + \left(|\kappa_0| + \frac{|\kappa_1|}{|s|} \right) O\left(\frac{e^{sz}}{s} \right)$$
for $Re \, p \to +\infty$ uniformly with respect to $z \in [0,1]$ and $Im \, p$.

Plugging (2.2.30) into (2.2.12), using the formulas (2.2.13), collecting the terms with θ_0 and θ_1 and simplifying by means of the relation $\frac{1}{a^2b} = \bar{b}$ (see (2.2.6)) we deduce

$$\begin{split} \psi(x,p) &= \frac{a(0)}{a(x)} \left[ch \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \frac{1}{s} \; sh \; sz \right. \\ &+ l\bar{b}(0)a'(0) \left(\frac{1}{s} sh \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \frac{1}{s^{2}} \; ch \; sz \right) \right] \theta_{0} \\ &+ \frac{1}{a(x)} \frac{l}{b(0)} \left(\frac{1}{s} sh \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \frac{1}{s^{2}} \; ch \; sz \right) \theta_{1} + \left(|\theta_{0}| + \frac{|\theta_{0}| + |\theta_{1}|}{|s|} \right) O\left(\frac{e^{sz}}{s^{2}} \right) \; , \\ \psi_{x}(x,p) &= \frac{b(x)a(0)}{l} \left[s \; sh \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \; ch \; sz \right. \\ &+ \left(l\bar{b}(0)a'(0) - l\bar{b}(x)a'(x) \right) \left(ch \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \frac{1}{s} \; sh \; sz \right) \\ &- l^{2}\bar{b}(0)a'(0)\bar{b}(x)a'(x) \left(\frac{1}{s} sh \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \frac{1}{s^{2}} \; ch \; sz \right) \right] \theta_{0} \\ &+ \frac{b(x)}{b(0)} \left[ch \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \frac{1}{s} \; sh \; sz \right. \\ &- l\bar{b}(x)a'(x) \left(\frac{1}{s} sh \; sz + \frac{1}{2} \int_{0}^{z} c(\tau) d\tau \frac{1}{s^{2}} \; ch \; sz \right) \right] \theta_{1} \\ &+ \left. \left(|\theta_{0}| + \frac{|\theta_{0}| + |\theta_{1}|}{|s|} \right) \left[O\left(\frac{e^{sz}}{s} \right) + O\left(\frac{e^{sz}}{s^{2}} \right) \right] \end{split}$$

for $Re \ p \to +\infty$ uniformly with respect to $x \in [0,1]$ and $\operatorname{Im} p$.

Let us extract the terms of first and second order in the process $\operatorname{Re} p \to +\infty$ inside the factors before θ_0 and θ_1 and shift the remaining parts into the *O*-terms. Observing in addition the relations (2.2.21),

$$sh sz, ch sz = O(e^{sz}), \quad \left|O\left(\frac{e^{sz}}{s}\right)\right| \le \frac{\operatorname{Const} e^{\operatorname{Re} sz}}{|s|} = O(e^{sz}) \quad (2.2.31)$$

and
$$\int_0^z c(\tau)d\tau = \zeta(x)$$
 we obtain (2.2.3). The lemma is proved.

2.2.2 Basic properties and representation of Green function in case of the first kind boundary conditions

Recall that the Green function of the operator L given by (1.2.11) in the interval $x \in (0,1)$ with the first kind boundary conditions is the function G that satisfies the problem (1.2.12). This function is given by the formula (cf. [39] Section 24, [23])

$$G(x,y;p) = \begin{cases} \frac{1}{\Delta_0(p)\lambda(0)} \psi_2(y,p)\psi_1(x,p) & \text{for } 0 \le x \le y \le 1\\ \frac{1}{\Delta_0(p)\lambda(0)} \psi_1(y,p)\psi_2(x,p) & \text{for } 0 \le y \le x \le 1, \end{cases}$$
(2.2.32)

where $\psi_j(x,p), j=1,2$ are the solutions of $L\psi=0$ satisfying the initial conditions

$$\psi_1(0,p) = 0$$
, $\psi_{1,x}(0,p) = c_1$, $\psi_2(1,p) = 0$, $\psi_{2,x}(1,p) = c_2$ (2.2.33)

with arbitrarily chosen numbers $c_1, c_2 \neq 0$ and $\Delta_0(p)$ is the Wronski determinant at zero:

$$\Delta_0(p) = \psi_1(0,p)\psi_{2,x}(0,p) - \psi_{1,x}(0,p)\psi_2(0,p) = -\psi_{1,x}(0,p)\psi_2(0,p).$$
 (2.2.34)

Here $\Delta(p) \neq 0$ for Re p > 0 due to the linear independence of ψ_1 and ψ_2 [39].

We note that ψ_1 satisfies the Cauchy problem (2.2.1) studied in the previous subsection. The function ψ_2 satisfies also the problem (2.2.1) provided we change its argument $\xi=1-x$. Thus, using the smoothness property $\psi(\cdot,p)\in C^2[0,1]$, $\operatorname{Re} p>0$, and the assertions about holomorphy of the solution of (2.2.1) in Lemma 2.1 as well as the formula (2.2.32) we deduce the following basic properties of the Green function:

Lemma 2.2. Assume that (2.2.2) holds and G is the Green function of the operator L in [0,1] with first kind boundary conditions. Then G(x,y,p), $G_x(x,y,p)$, $G_y(x,y,p)$ and $G_{xy}(x,y,p)$ are are holomorphic in $\operatorname{Re} p > 0$ for any $(x,y) \in [0,1]^2$, $x \neq y$, and continuous and bounded in every strip

$$S_{\sigma} = \{(x, y) \in [0, 1]^2 : x \neq y\} \times \{p \in \mathbb{C} : 0 < \operatorname{Re} p < \sigma\}, \quad \sigma > 0.$$

Moreover, for any function V(x,p) such that $V(\cdot,p) \in C[0,1]$, $\operatorname{Re} p > \sigma$ and V(x,p) - holomorphic in $\operatorname{Re} p > \sigma$ for any $x \in [0,1]$ with some $\sigma \geq 0$, the functions

$$\int_{0}^{1} G(x,y,p)V(y,p)dy, \quad \int_{0}^{1} G_{x}(x,y,p)V(y,p)dy,$$
$$\int_{0}^{1} G_{y}(x,y,p)V(y,p)dy, \quad \int_{0}^{1} G_{xy}(x,y,p)V(y,p)dy$$

belong to C[0,1] for any $\operatorname{Re} p > \sigma$ and are holomorphic in $\operatorname{Re} p > \sigma$ for any $x \in [0,1]$.

Let us set $c_1 = \frac{b(0)}{l}$ with l and b given by (2.2.4) and (2.2.6). Then ψ_1 solves (2.2.1) with $\theta_0 = 0$, $\theta_1 = \frac{b(0)}{l}$, hence by Lemma 2.1 the function ψ_1 and its first derivative possess the following asymptotic behavior:

$$\psi_{1}(x,p) = \frac{1}{a(x)} \frac{1}{s} sh \ sz + \frac{1}{a(x)} \frac{\zeta(x)}{2} \frac{1}{s^{2}} ch \ sz + O\left(\frac{e^{sz}}{s^{3}}\right)$$

$$= \frac{1}{a(x)} \frac{1}{s} sh \ sz + O\left(\frac{e^{sz}}{s^{2}}\right) ,$$

$$\psi_{1,x}(x,p) = \frac{b(x)}{l} ch \ sz + \frac{b(x)}{l} \left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x)\right) \frac{1}{s} sh \ sz + O\left(\frac{e^{sz}}{s^{2}}\right)$$

$$= \frac{b(x)}{l} ch \ sz + O\left(\frac{e^{sz}}{s}\right)$$
for $Re \ p \to +\infty$ uniformly with respect to $x \in [0,1]$ and $Im \ p$

where a, \bar{b} and ζ are defined in (2.2.6) and (2.2.8) with (2.2.9). In (2.2.33) we set $c_2 = \frac{b(1)}{l}$. Then using the change of variables

$$\xi = 1 - x, \ \psi^{\dagger}(\xi, p) = \psi_2(x, p), \ \lambda^{\dagger}(\xi) = \lambda(x), \ \beta^{\dagger}(\xi) = \beta(x)$$
 (2.2.36)

the problem for ψ_2 is transformed to the following Cauchy problem for ψ^{\dagger} :

$$(L\psi^{\dagger})(\xi,p) = 0, \ \xi \in (0,1), \quad \psi^{\dagger}(0,p) = 0, \ \psi^{\dagger}_{\xi}(0,p) = -\frac{b(1)}{l}.$$
 (2.2.37)

Using Lemma 2.1 we write the asymptotical formulas (2.2.3) for $\psi^{\dagger}(\xi,p)$ and $\psi^{\dagger}_{\xi}(\xi,p)$. These formulas contain instead of the quantities x,a,b,\bar{b},z and ζ the quantities $\xi,a^{\dagger},b^{\dagger},\bar{b}^{\dagger},z^{\dagger}$ and ζ^{\dagger} , respectively, where

$$a^{\dagger}(\xi) = (\beta^{\dagger}(\xi)\lambda^{\dagger}(\xi))^{1/4}, \ b^{\dagger}(\xi) = \beta^{\dagger}(\xi)^{1/4}\lambda^{\dagger}(\xi)^{-3/4},$$

$$\bar{b}^{\dagger}(\xi) = \lambda^{\dagger}(\xi)^{1/4}\beta^{\dagger}(\xi)^{-3/4},$$

$$z^{\dagger} = \frac{1}{l} \int_{0}^{\xi} \sqrt{\frac{\beta^{\dagger}(\tau)}{\lambda^{\dagger}(\tau)}} d\tau, \ \zeta^{\dagger}(\xi) = \frac{1}{l} \int_{0}^{\xi} a_{1}^{\dagger}(\tau)\sqrt{\frac{\beta^{\dagger}(\tau)}{\lambda^{\dagger}(\tau)}} d\tau$$

$$(2.2.38)$$

and a_1^{\dagger} is defined by (2.2.9) with a, λ, β replaced by $a^{\dagger}, \lambda^{\dagger}, \beta^{\dagger}$. Observing that

$$a^{\dagger}(\xi) = a(x), \ a^{\dagger'}(\xi) = -a'(x), \ b^{\dagger}(\xi) = b(x), \ \bar{b}^{\dagger}(\xi) = \bar{b}(x),$$

 $z^{\dagger}(\xi) = 1 - z(x), \ \zeta^{\dagger}(\xi) = \zeta(1) - \zeta(x)$ (2.2.39)

from these formulas we obtain

$$\psi_{2}(x,p) = \frac{1}{a(x)} \frac{1}{s} sh \ s(z-1) - \frac{1}{a(x)} \frac{\zeta(1) - \zeta(x)}{2} \frac{1}{s^{2}} ch \ s(1-z)$$

$$+ O\left(\frac{e^{s(1-z)}}{s^{3}}\right) = \frac{1}{a(x)} \frac{1}{s} sh \ s(z-1) + O\left(\frac{e^{s(1-z)}}{s^{2}}\right) ,$$

$$\psi_{2,x}(x,p) = \frac{b(x)}{l} ch \ s(1-z) - \frac{b(x)}{l} \left(\frac{\zeta(1) - \zeta(x)}{2} + l\bar{b}(x)a'(x)\right) \frac{1}{s} sh \ s(z-1)$$

$$+ O\left(\frac{e^{s(1-z)}}{s^{2}}\right) = \frac{b(x)}{l} ch \ s(1-z) + O\left(\frac{e^{s(1-z)}}{s}\right)$$
for $Re \ p \to +\infty$ uniformly with respect to $x \in [0,1]$ and $Im \ p$.

Further, defining

$$d_0(p) = s\lambda(0)\Delta_0(p) \tag{2.2.41}$$

from (2.2.34) by means of (2.2.35), (2.2.40), (2.2.31) and the relation $\lambda(0) = \frac{a(0)}{b(0)}$ we derive the following asymptotic formula for d_0 :

$$d_0(p) = -s\lambda(0) \left[\frac{b(0)}{l} + O\left(\frac{1}{s}\right) \right] \left[-\frac{1}{a(0)} \frac{1}{s} sh \, s + O\left(\frac{e^s}{s^2}\right) \right]$$

$$= \frac{1}{l} sh \, s + O\left(\frac{e^s}{s}\right) \text{ for } \operatorname{Re} p \to +\infty \text{ uniformly in } \operatorname{Im} p.$$

$$(2.2.42)$$

Now we are in the situation to deduce the asymptotical relations for G and its derivatives. To this end we plug (2.2.35), (2.2.40) and (2.2.41) into (2.2.32). In case we take into consideration only the first-order asymptotics, we get the following representation for G:

$$G(x,y;p) = \frac{1}{d_0(p)} \frac{1}{a(x)a(y)} \frac{1}{s} \begin{cases} sh \, sz \cdot sh \, s(w-1) + O_1 & \text{for } x \leq y \\ sh \, sw \cdot sh \, s(z-1) + O_2 & \text{for } y \leq x, \end{cases} (2.2.43)$$

where

$$w = w(y) = \frac{1}{l} \int_0^y \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau$$
 (2.2.44)

and

$$O_1 = O\left(\frac{e^{s(1-w+z)}}{s}\right), \quad O_2 = O\left(\frac{e^{s(w+1-z)}}{s}\right)$$
 for $\operatorname{Re} p \to +\infty$ uniformly in $0 \le y < x \le 1$ and $\operatorname{Im} p$.

Similarly, from (2.2.32) by means of (2.2.35), (2.2.40) and (2.2.41) for the derivatives G_x , G_y and G_{xy} we have the representations

$$G_x(x, y; p) = \frac{1}{ld_0(p)} \frac{b(x)}{a(y)} \begin{cases} ch \ sz \cdot sh \ s(w-1) + O_3 & \text{for} \quad x < y \\ sh \ sw \cdot ch \ s(z-1) + O_4 & \text{for} \quad y < x, \end{cases} (2.2.46)$$

$$G_y(x, y; p) = \frac{1}{ld_0(p)} \frac{b(y)}{a(x)} \begin{cases} sh sz \cdot ch s(w-1) + O_5 & \text{for } x < y \\ ch sw \cdot sh s(z-1) + O_6 & \text{for } y < x, \end{cases} (2.2.47)$$

$$G_{xy}(x, y; p) = \frac{b(x)b(y)}{l^2d_0(p)} s \begin{cases} ch \ sz \cdot ch \ s(w-1) + O_7 & \text{for } x < y \\ ch \ sw \cdot ch \ s(z-1) + O_8 & \text{for } y < x, \end{cases} (2.2.48)$$

where the terms O_3, O_5, O_7 behave like O_1 and the terms O_4, O_6, O_8 like O_2 , respectively, as $\text{Re } p \to +\infty$.

To deal with the inverse problem with flux observations we have to prove certain properties of G_x that require the second-order asymptotic relation for this function. Taking the second order asymptotics in (2.2.35), (2.2.40) into account, from (2.2.32) with (2.2.41) we deduce the formula

$$G_{x}(x,y;p) = \begin{cases} \frac{1}{ld_{0}(p)} \frac{b(x)}{a(y)} \left[ch \ sz \cdot sh \ s(w-1) + \left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) \right) \frac{1}{s}sh \ sz \cdot sh \ s(w-1) - \frac{\zeta(1) - \zeta(y)}{2} \frac{1}{s}ch \ sz \cdot ch \ s(w-1) + \frac{1}{s}O_{9} \right] & \text{for} \quad x < y, \\ \frac{1}{ld_{0}(p)} \frac{b(x)}{a(y)} \left[sh \ sw \cdot ch \ s(z-1) - \left(\frac{\zeta(1) - \zeta(x)}{2} + l\bar{b}(x)a'(x) \right) \frac{1}{s}sh \ sw \cdot sh \ s(z-1) + \frac{\zeta(y)}{2} \frac{1}{s}ch \ sw \cdot ch \ s(z-1) + \frac{1}{s}O_{10} \right] & \text{for} \quad y < x \end{cases}$$

$$(2.2.49)$$

where O_9 and O_{10} behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$.

2.2.3 Basic properties and representation of Green function in case of the third kind boundary conditions

The Green function G of the operator L given by (1.2.11) in the interval $x \in (0,1)$ with the third kind boundary conditions is a solution to the problem (1.2.20). This function is given by the formula (2.2.32), where $\psi_j(x,p)$, j=1,2 are the solutions of $L\psi=0$ satisfying the initial conditions

$$\psi_1(0,p) = c_1 \lambda(0), \ \psi'_{1,x}(0,p) = c_1 \alpha_1,
\psi_2(1,p) = c_2 \lambda(1), \ \psi'_{2,p}(1,p) = -c_2 \alpha_2$$
(2.2.50)

with arbitrarily chosen numbers c_1 and c_2 and $\Delta_0(p)$ again is the Wronski determinant at zero, i.e.

$$\Delta_0(p) = \psi_1(0, p)\psi_{2,x}(0, p) - \psi_{1,x}(0, p)\psi_2(0, p)$$
(2.2.51)

(cf. [39] Section 24, [13] Chapter 7.2 and [24]). The assumed inequalities $\alpha_1, \alpha_2 \ge 0$ (Section 1.1) imply that 0 is not an eigenvalue of the operator L with the third kind boundary conditions in case $\operatorname{Re} p > 0$. This in turn yields that ψ_1 and ψ_2 are linearly independent [39]. Thus, $\Delta(p) \ne 0$ for $\operatorname{Re} p > 0$.

As in the previous subsection we deduce the following basic properties of G:

Lemma 2.3. Assume that (2.2.2) holds and G is the Green function of the operator L in [0,1] with the third kind boundary conditions. Then the assertions of Lemma 2.2 are valid for G.

Let us set $c_1 = \frac{1}{a(0)\lambda(0)}$ with a and b given by (2.2.6). Then using the assertion (2.2.3) of Lemma 2.1 and observing the relation $\frac{1}{ab} = a\bar{b}$ we derive for the function ψ_1 and its first derivative the following asymptotical formulas:

$$\psi_{1}(x,p) = \frac{1}{a(x)}ch \, sz + \frac{1}{a(x)} \left[\frac{\zeta(x)}{2} + \frac{l\bar{b}(0)}{\lambda(0)} \left(\lambda(0)a'(0) + \alpha_{1}a(0) \right) \right] \frac{1}{s} \, sh \, sz \\ + O\left(\frac{e^{sz}}{s^{2}} \right) = \frac{1}{a(x)}ch \, sz + O\left(\frac{e^{sz}}{s} \right) \,,$$

$$\psi_{1,x}(x,p) = \frac{b(x)}{l} \, s \, sh \, sz \\ + \frac{b(x)}{l} \left[\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) + \frac{l\bar{b}(0)}{\lambda(0)} \left(\lambda(0)a'(0) + \alpha_{1}a(0) \right) \right] ch \, sz \\ + O\left(\frac{e^{sz}}{s} \right) = \frac{b(x)}{l} \, s \, sh \, sz + O\left(e^{sz} \right)$$
for $Re \, p \to +\infty$ uniformly with respect to $x \in [0,1]$ and $Im \, p$.

Further, let us set and $c_2 = \frac{1}{a(1)\lambda(1)}$. Using again the change of variables (2.2.36) we get the following Cauchy problem for ψ^{\dagger} :

$$L(\psi^{\dagger})(\xi, p) = 0, \ \xi \in (0, 1), \quad \psi^{\dagger}(0, p) = \frac{1}{a(0)}, \ \psi^{\dagger}_{\xi}(0, p) = \frac{\alpha_2}{a(1)\lambda(1)}.$$
 (2.2.53)

Writing the asymptotical formulas (2.2.3) for ψ^{\dagger} and ψ^{\dagger}_{ξ} in terms of the quantities (2.2.38) and observing the relations (2.2.39) and $\frac{1}{ab} = a\bar{b}$, again, we arrive at the following representations:

$$\psi_{2}(x,p) = \frac{1}{a(x)}ch\ s(1-z)$$

$$-\frac{1}{a(x)}\left[\frac{\zeta(1)-\zeta(x)}{2} + \frac{l\bar{b}(1)}{\lambda(1)}\left(-\lambda(1)a'(1) + \alpha_{2}a(1)\right)\right] \frac{1}{s}\ sh\ s(z-1)$$

$$+O\left(\frac{e^{s(1-z)}}{s^{2}}\right) = \frac{1}{a(x)}ch\ s(1-z) + O\left(\frac{e^{s(1-z)}}{s}\right)\ ,$$

$$\psi_{2,x}(x,p) = \frac{b(x)}{l}\ s\ sh\ s(z-1) - \frac{b(x)}{l}\left[\frac{\zeta(1)-\zeta(x)}{2}\right]$$

$$+l\bar{b}(x)a'(x) + \frac{l\bar{b}(1)}{\lambda(1)}\left(-\lambda(1)a'(1) + \alpha_{2}a(1)\right)\right]ch\ s(1-z)$$

$$+O\left(\frac{e^{s(1-z)}}{s}\right) = \frac{b(x)}{l}\ s\ sh\ s(z-1) + O\left(e^{s(1-z)}\right)$$
for $Re\ p \to +\infty$ uniformly with respect to $x \in [0,1]$ and $Im\ p$.

We define

$$d_1(p) = -\frac{1}{s}\lambda(0)\Delta_0(p). \tag{2.2.55}$$

Then from (2.2.51) by means of (2.2.52), (2.2.54) with the first-order asymptotics, (2.2.31) and the relation $\lambda(0) = \frac{a(0)}{b(0)}$ we obtain

$$d_{1}(p) = \frac{1}{s}\lambda(0) \left[\frac{1}{a(0)} + O\left(\frac{1}{s}\right) \right] \left[-\frac{b(0)}{l}s \, sh \, s + O\left(e^{s}\right) \right]$$

$$-\frac{1}{s}\lambda(0) \, O\left(1\right) \left[\frac{1}{a(0)}ch \, s + O\left(\frac{e^{s}}{s}\right) \right]$$

$$= -\frac{1}{l}sh \, s + O\left(\frac{e^{s}}{s}\right) \quad \text{for } \operatorname{Re} p \to +\infty \quad \text{uniformly in } \operatorname{Im} p \, .$$

$$(2.2.56)$$

Now we can formulate the asymptotical relations for G and its derivatives. Formulas (2.2.32), (2.2.52), (2.2.54) with the first-order asymptotics and (2.2.55) yield

$$G(x,y;p) = \frac{1}{d_1(p)} \frac{1}{a(x)a(y)} \frac{1}{s} \begin{cases} ch \ sz \cdot ch \ s(w-1) + \hat{O}_1 & \text{for } x \le y \\ ch \ sw \cdot ch \ s(z-1) + \hat{O}_2 & \text{for } y \le x, \end{cases} (2.2.57)$$

$$G_x(x, y; p) = \frac{1}{ld_1(p)} \frac{b(x)}{a(y)} \begin{cases} sh sz \cdot ch s(w-1) + \hat{O}_3 & \text{for } x < y \\ ch sw \cdot sh s(z-1) + \hat{O}_4 & \text{for } y < x, \end{cases} (2.2.58)$$

$$G_{y}(x, y; p) = \frac{1}{ld_{1}(p)} \frac{b(y)}{a(x)} \begin{cases} ch \ sz \cdot sh \ s(w-1) + \hat{O}_{5} & \text{for} \quad x < y \\ sh \ sw \cdot ch \ s(z-1) + \hat{O}_{6} & \text{for} \quad y < x, \end{cases} (2.2.59)$$

$$G_{xy}(x, y; p) = \frac{b(x)b(y)}{l^2d_1(p)}s \begin{cases} sh sz \cdot sh s(w-1) + \hat{O}_7 & \text{for } x < y \\ sh sw \cdot sh s(z-1) + \hat{O}_8 & \text{for } y < x, \end{cases} (2.2.60)$$

where w=w(y) is defined by (2.2.44) and the terms $\hat{O}_1,\hat{O}_3,\hat{O}_5,\hat{O}_7$ behave like O_1 and the terms $\hat{O}_2,\hat{O}_4,\hat{O}_6,\hat{O}_8$ like O_2 , respectively, in the process $\operatorname{Re} p \to +\infty$ (see (2.2.45)). Moreover, using the second order asymptotics in (2.2.52), (2.2.54) we derive the following representation for G_x :

$$G_{x}(x,y;p) = \begin{cases} \frac{1}{ld_{1}(p)} \frac{b(x)}{a(y)} \left[sh \, sz \cdot ch \, s(w-1) \right. \\ + \left. \left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) + \vartheta_{0} \right) \frac{1}{s}ch \, sz \cdot ch \, s(w-1) \right. \\ - \left. \left(\frac{\zeta(1) - \zeta(y)}{2} - \vartheta_{1} \right) \frac{1}{s}sh \, sz \cdot sh \, s(w-1) + \frac{1}{s}\hat{O}_{9} \right] \text{ for } x < y, \\ \frac{1}{ld_{1}(p)} \frac{b(x)}{a(y)} \left[ch \, sw \cdot sh \, s(z-1) \right. \\ - \left. \left(\frac{\zeta(1) - \zeta(x)}{2} + l\bar{b}(x)a'(x) - \vartheta_{1} \right) \frac{1}{s}ch \, sw \cdot ch \, s(z-1) \right. \\ + \left. \left(\frac{\zeta(y)}{2} + \vartheta_{0} \right) \frac{1}{s}sh \, sw \cdot sh \, s(z-1) + \frac{1}{s}\hat{O}_{10} \right] \text{ for } y < x, \end{cases}$$

where

$$\vartheta_0 = \frac{l\bar{b}(0)}{\lambda(0)} \left(\lambda(0)a'(0) + \alpha_1 a(0) \right), \ \vartheta_1 = \frac{l\bar{b}(1)}{\lambda(1)} \left(\lambda(1)a'(1) - \alpha_2 a(1) \right) (2.2.62)$$

and \hat{O}_9 and \hat{O}_{10} behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$.

2.3 Auxiliary results

Lemma 2.4. Let $\lambda, \beta \in C[0,1]$ and $\lambda, \beta > 0$ in [0,1] and \mathcal{G} be a function that is bounded in every strip S_{σ} , $\sigma > 0$, and has the following asymptotical representation:

$$\mathcal{G}(x,y;p) = \frac{1}{\widetilde{d}_0(p)} \begin{cases} \rho_1(x,y)\chi_1(sz)\chi_2(s(w-1)) + \mathcal{O}_1 & \text{for } x < y \\ \rho_2(x,y)\chi_3(sw)\chi_4(s(z-1)) + \mathcal{O}_2 & \text{for } y < x, \end{cases} (2.3.1)$$

where ρ_1 , ρ_2 are some bounded functions, χ_1 , χ_2 , χ_3 and χ_4 are either sh or ch and the terms \mathcal{O}_1 and \mathcal{O}_2 behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$. Moreover, let

$$\widetilde{d}_0(p) = \alpha \, sh \, s + O\left(\frac{e^s}{s}\right) \quad \text{for } \operatorname{Re} p \to +\infty \quad \textit{uniformly in } \operatorname{Im} p \quad (2.3.2)$$

with some $\alpha \neq 0$. Then the estimate

$$\sup_{\substack{0 \le x \le 1 \\ \text{Re } p > 0}} \sqrt{|p|} \int_0^1 |\mathcal{G}(x, y; p)| dy < \infty \tag{2.3.3}$$

holds.

Proof. First, let us estimate $\widetilde{d}_0(p)$ from below. From (2.3.2) due to (2.2.18) and the relation $|sh\,s| \ge sh(\operatorname{Re} s)$ we have

$$|\widetilde{d}_0(p)| \geq |\alpha| \; sh(\operatorname{Re} s) - O\left(\frac{e^{\operatorname{Re} s}}{|s|}\right) \quad \text{for } \operatorname{Re} p \to +\infty \quad \text{uniformly in } \operatorname{Im} p \, .$$

Since $\operatorname{Re} p \to +\infty \Rightarrow \operatorname{Re} s \to +\infty$ (see (2.2.21)), the term $|\alpha| sh(\operatorname{Re} s)$ dominates over the term $O\left(\frac{e^{\operatorname{Re} s}}{|s|}\right)$ in the process $\operatorname{Re} p \to +\infty$. Thus, in case of sufficiently large σ the estimate

$$|\widetilde{d}_0(p)| \ge \frac{|\alpha|}{2} sh(\operatorname{Re} s)$$
 for $\operatorname{Re} p > \sigma$

is valid. Due to this estimate, (2.2.18) and the relation $s=l\sqrt{p}$ from (2.3.1) we obtain

$$\begin{split} \sqrt{|p|} \int_0^1 |\mathcal{G}(x,y;p)| dy &\leq \operatorname{Const} \frac{|s|}{sh(\operatorname{Re} s)} \Big[\int_0^x ch(\operatorname{Re} s \cdot w) ch \operatorname{Re} s(z-1) dy \\ &+ \int_x^1 ch(\operatorname{Re} s \cdot z) ch \operatorname{Re} s(w-1) dy \Big] \\ &= \operatorname{Const} \frac{|s|}{\operatorname{Re} s \cdot sh(\operatorname{Re} s)} \left[\int_0^x \frac{1}{w'(y)} \frac{d}{dy} sh(\operatorname{Re} s \cdot w) dy \cdot ch \operatorname{Re} s(z-1) \\ &+ ch(\operatorname{Re} s \cdot z) \int_x^1 \frac{1}{w'(y)} \frac{d}{dy} sh \operatorname{Re} s(w-1) dy \right], \quad x \in [0,1], \operatorname{Re} p > \sigma. \end{split}$$

We note that the relation

$$w'(y) \ge \varkappa > 0, \quad y \in [0, 1]$$
 (2.3.4)

holds due to (2.2.44) and the assumptions imposed on λ and β . Using (2.3.4), the positivity of $\frac{d}{dy}sh(\operatorname{Re} s\cdot w), \frac{d}{dy}sh\operatorname{Re} s(w-1)$, the definitions (2.2.7), (2.2.44) of z,w and the relation $s=l\sqrt{p}$ we further obtain

$$\sqrt{|p|} \int_0^1 |\mathcal{G}(x,y;p)| dy \le \operatorname{Const} \frac{|s|}{\operatorname{Re} s \cdot sh(\operatorname{Re} s)}$$

$$\times \left[\int_0^x \frac{d}{dy} sh(\operatorname{Re} s \cdot w) \, dy \cdot ch \, \operatorname{Re} s(z-1) + ch(\operatorname{Re} s \cdot z) \int_x^1 \frac{d}{dy} sh \, \operatorname{Re} s(w-1) \, dy \right]$$

$$= \operatorname{Const} \frac{|s|}{\operatorname{Re} s} \frac{1}{sh(\operatorname{Re} s)} \left[sh(\operatorname{Re} s \cdot z) \, ch \, \operatorname{Re} s(1-z) + ch(\operatorname{Re} s \cdot z) \, sh \, \operatorname{Re} s(1-z) \right]$$

$$= \operatorname{Const} \frac{|s|}{\operatorname{Re} s} \quad \text{for} \quad x \in [0,1] \,, \, \operatorname{Re} p > \sigma \,.$$

Due to (2.2.21) we deduce the inequality

$$\sup_{\substack{0 \le x \le 1 \\ \text{Re } p > \sigma}} \sqrt{|p|} \int_0^1 |\mathcal{G}(x, y; p)| dy < \infty.$$

Finally, observing that $\mathcal{G}(x, y, p)$ is bounded in S_{σ} , this supremum can be extended to $0 \le x \le 1$, Re p > 0. This proves (2.3.3).

Lemma 2.5. Let $\lambda, \beta \in C^1[0,1]$ and $\lambda, \beta > 0$ in [0,1]. Let \mathcal{G} be a function that has the following asymptotical representation:

$$\mathcal{G}(x,y;p) = \begin{cases} q_1(x)\rho_1(y)\chi_1(sz)\chi_2'(s(w-1)) + \mathcal{O}_1 & \text{for } x < y \\ q_2(x)\rho_2(y)\chi_3'(sw)\chi_4(s(z-1)) + \mathcal{O}_2 & \text{for } y < x, \end{cases}$$
(2.3.5)

where q_1, q_2, χ_1, χ_4 are some continuous functions, $\rho_1, \rho_2, \chi_2, \chi_3$ are some differentiable functions and the terms \mathcal{O}_1 and \mathcal{O}_2 behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$. Then for any $V \in C^1[0,1]$ the relation

$$\int_{0}^{1} \mathcal{G}(x, y; p) V(y) dy$$

$$= \frac{1}{s} \left[q_{1}(x) \frac{\rho_{1}(y)}{w'(y)} \chi_{1}(sz) \chi_{2}(s(w-1)) + \mathcal{O}_{3}(x, y, p) \right] V(y) \Big|_{y=x}^{y=1}
+ \frac{1}{s} \left[q_{2}(x) \frac{\rho_{2}(y)}{w'(y)} \chi_{3}(sw) \chi_{4}(s(z-1)) + \mathcal{O}_{4}(x, y, p) \right] V(y) \Big|_{y=0}^{y=x}
- \int_{0}^{1} \mathcal{T}[V](x, y; p) dy$$
(2.3.6)

holds, where

$$T[V](x,y,p)$$

$$= \frac{1}{s} \begin{cases} q_1(x) \left(\frac{\rho_1(y)V(y)}{w'(y)}\right)' \chi_1(sz)\chi_2(s(w-1)) + V'(y)\mathcal{O}_3 & \text{for } x < y \\ q_2(x) \left(\frac{\rho_2(y)V(y)}{w'(y)}\right)' \chi_3(sw)\chi_4(s(z-1)) + V'(y)\mathcal{O}_4 & \text{for } y < x, \end{cases}$$
(2.3.7)

$$\mathcal{O}_3(x,y,p) = s \int_1^y \mathcal{O}_1(x,\eta,p) d\eta \,, \quad \mathcal{O}_4(x,y,p) = s \int_0^y \mathcal{O}_2(x,\eta,p) d\eta \,\,(2.3.8)$$

and \mathcal{O}_3 , \mathcal{O}_4 behave like \mathcal{O}_1 , \mathcal{O}_2 , respectively, as $\operatorname{Re} p \to +\infty$.

Proof. Let ρ, χ be some differentiable functions and $0 \le c_1, c_2 \le 1$. By means of integration by parts we get the following auxiliary relation:

$$\int_{c_1}^{c_2} \rho(y) \chi'(sw) dy = \frac{1}{s} \int_{c_1}^{c_2} \frac{\rho(y)}{w'(y)} \frac{d}{dy} \chi(sw) dy$$

$$= \frac{1}{s} \frac{\rho(y)}{w'(y)} \chi(sw) \Big|_{y=c_1}^{y=c_2} - \frac{1}{s} \int_{c_1}^{c_2} \left(\frac{\rho(y)}{w'(y)} \right)' \chi(sw) dy. \tag{2.3.9}$$

Further, in view of the definition (2.3.8) of \mathcal{O}_3 and \mathcal{O}_4 we have $\mathcal{O}_1 = \frac{1}{s} \frac{d}{dy} \mathcal{O}_3$ and $\mathcal{O}_2 = \frac{1}{s} \frac{d}{dy} \mathcal{O}_4$. Thus, due to (2.3.5), we can rewrite the integral $\int_0^1 \mathcal{G}(x,y,p) V(y) dy$ in the form

$$\int_{0}^{1} \mathcal{G}(x,y,p)V(y)dy
= q_{1}(x)\chi_{1}(sz) \int_{x}^{1} \rho_{1}(y)V(y)\chi'_{2}(s(w-1))dy + \frac{1}{s} \int_{x}^{1} V(y)\frac{d}{dy}\mathcal{O}_{3}(x,y,p)dy
+q_{2}(x)\chi_{4}(s(z-1)) \int_{0}^{x} \rho_{2}(y)V(y)\chi'_{3}(sw)dy + \frac{1}{s} \int_{0}^{x} V(y)\frac{d}{dy}\mathcal{O}_{4}(x,y,p)dy.$$

In this relation we apply to the integrals containing χ'_2 and χ'_3 the formula (2.3.9) and integrate the integrals containing \mathcal{O}_3 and \mathcal{O}_4 by parts. As a result we immediately obtain (2.3.6) with (2.3.7).

To complete the proof it remains to show that \mathcal{O}_3 and \mathcal{O}_4 behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$. To this end let us define

$$\xi = \xi(\eta) = \frac{1}{l} \int_0^{\eta} \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau.$$
 (2.3.10)

Observing (2.3.8), the behavior of \mathcal{O}_1 , \mathcal{O}_2 , (2.2.45) and the definition (2.2.44) of w(y) and (2.3.10) of $\xi(\eta)$ we compute:

$$\left| \frac{e^{s(1-w+z)}}{s} \right|^{-1} |\mathcal{O}_3(x,y,p)| = \left[\frac{e^{\operatorname{Re} s(1-w+z)}}{|s|} \right]^{-1} |s| \left| \int_1^y \mathcal{O}_1(x,\eta,p) d\eta \right|$$

$$\leq \operatorname{Const} \left[\frac{e^{\operatorname{Re} s(1-w+z)}}{|s|} \right]^{-1} |s| \int_y^1 \frac{e^{\operatorname{Re} s(1-\xi+z)}}{|s|} d\eta = \operatorname{Const} |s| \int_y^1 e^{\operatorname{Re} s(w-\xi)} d\eta$$

$$= -\frac{\operatorname{Const} |s|}{\operatorname{Re} s} \int_y^1 \frac{1}{\xi'(\eta)} \frac{d}{d\eta} e^{\operatorname{Re} s(w-\xi)} d\eta \quad \text{for } 0 \leq x < y \leq 1, \ \operatorname{Re} p > \sigma,$$

$$\left| \frac{e^{s(w+1-z)}}{s} \right|^{-1} |\mathcal{O}_4(x,y,p)| = \left[\frac{e^{\operatorname{Re} s(w+1-z)}}{|s|} \right]^{-1} |s| \left| \int_0^y \mathcal{O}_2(x,\eta,p) d\eta \right|
\leq \operatorname{Const} \left[\frac{e^{\operatorname{Re} s(w+1-z)}}{|s|} \right]^{-1} |s| \int_y^1 \frac{e^{\operatorname{Re} s(\xi+1-z)}}{|s|} d\eta = \operatorname{Const} |s| \int_0^y e^{\operatorname{Re} s(\xi-w)} d\eta
= \frac{\operatorname{Const} |s|}{\operatorname{Re} s} \int_0^y \frac{1}{\xi'(\eta)} \frac{d}{d\eta} e^{\operatorname{Re} s(\xi-w)} d\eta \quad \text{for } 0 \leq y < x \leq 1, \operatorname{Re} p > \sigma$$

2.4. Asymptotical properties of Green function in case of the first kind boundary conditions

with some sufficiently large $\sigma>0$ independent of x,y. Since $\frac{1}{\xi'(\eta)}$ is bounded, $\frac{d}{d\eta}e^{\operatorname{Re} s(w-\xi)}\leq 0$ and $\frac{d}{d\eta}e^{\operatorname{Re} s(\xi-w)}\geq 0$ using the inequality (2.2.21) we obtain

$$\begin{split} \left| \frac{e^{s(1-w+z)}}{s} \right|^{-1} |\mathcal{O}_3(x,y,p)| &\leq -\frac{\operatorname{Const}|s|}{\operatorname{Re}\,s} \int_y^1 \frac{d}{d\eta} e^{\operatorname{Re}\,s(w-\xi)} d\eta \\ &= \frac{\operatorname{Const}|s|}{\operatorname{Re}\,s} \left[1 - e^{\operatorname{Re}\,s(w-1)} \right] \leq \operatorname{Const} \quad \text{for } \ 0 \leq x < y \leq 1 \,, \ \operatorname{Re}\,p > \sigma \,, \\ \left| \frac{e^{s(w+1-z)}}{s} \right|^{-1} |\mathcal{O}_4(x,y,p)| &\leq \frac{\operatorname{Const}|s|}{\operatorname{Re}\,s} \int_0^y \frac{d}{d\eta} e^{\operatorname{Re}\,s(\xi-w)} d\eta \\ &\qquad \qquad \frac{\operatorname{Const}|s|}{\operatorname{Re}\,s} \left[1 - e^{-\operatorname{Re}\,sw} \right] \leq \operatorname{Const} \quad \text{for } \ 0 \leq y < x \leq 1 \,, \ \operatorname{Re}\,p > \sigma \,. \end{split}$$

This proves that \mathcal{O}_3 and \mathcal{O}_4 behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$ (cf. (2.2.45)).

2.4 Asymptotical properties of Green function in case of the first kind boundary conditions

Let us start by proving some estimates of integrals of G and its derivatives.

Theorem 2.1. Let λ , β satisfy (2.2.2). Then the Green function G of the operator L in [0,1] with the first kind boundary conditions satisfies the following estimates:

$$C_1 = \sup_{\substack{0 \le x \le 1 \\ \text{Re } p > 0}} |p| \int_0^1 |G(x, y; p)| dy < \infty, \tag{2.4.1}$$

$$C_2 = \sup_{\substack{0 \le x \le 1 \\ \text{Re } p > 0}} \sqrt{|p|} \int_0^1 |G_x(x, y; p)| dy < \infty, \tag{2.4.2}$$

$$C_3 = \sup_{\substack{0 \le x \le 1 \\ \text{Re } p > 0}} \sqrt{|p|} \int_0^1 |G_y(x, y; p)| dy < \infty, \tag{2.4.3}$$

$$C_4 = \sup_{\substack{0 \le x \le 1 \\ \text{Re } p > 0}} \int_0^1 |G_{xy}(x, y; p)| dy < \infty .$$
 (2.4.4)

Theorem 2.1 was proved in [23], but since the proof is short, we will present it here.

Proof. Estimates (2.4.1) - (2.4.4) follow from Lemma 2.4 if we take the relations (2.2.43), (2.2.46) - (2.2.48) for G, G_x , G_y , G_{xy} , the positivity and boundendness of a and b following from (2.2.6), (2.2.2), the assumptions of Theorem 2.1 and the equality $s = l\sqrt{p}$ into account.

Next we formulate without proofs two theorems proved also in [23].

2. Functional spaces and properties of Green function

Theorem 2.2. Let λ, β satisfy (2.2.2) and G be the Green function of the operator L in [0,1] with the first kind boundary conditions. If $V \in C^1[0,1]$, then the estimate

$$\sup_{\text{Re }p>0} \left| \sqrt{|p|} \left(\int_0^1 G_{xy}(x,y,p) V(y) dy - \frac{V(x)}{\lambda(x)} \right) \right| \le C_5 \|V'\|_{C[0,1]} \quad (2.4.5)$$

holds for any $x \in [0,1]$, where C_5 is a positive constant.

Theorem 2.3. Let λ, β satisfy (2.2.2) and G be the Green function of the operator L in [0,1] with the first kind boundary conditions. If $V \in C^1[0,1]$, then the estimate

$$\sup_{\text{Re }p>0} \left| \sqrt{|p|} \left(\int_0^1 pG(x,y,p)V(y)dy + \frac{V(x)}{\beta(x)} \right) \right| \le C_6(x) \|V\|_{C^1[0,1]}$$
 (2.4.6)

holds for any $x \in (0,1)$, where $C_6(x)$ is some function bounded in every compact subinterval of (0,1).

Finally, we prove a similar theorem about the asymptotics of the derivative of $\int_0^1 G(x,y,p)V(y)dy$.

Theorem 2.4. Let λ, β satisfy (2.2.2) and G be the Green function of the operator L in [0,1] with the first kind boundary conditions. If $V \in C^2[0,1]$, then the estimate

$$\sup_{\text{Re }p>0}\left|\sqrt{|p|}\left[\int_{0}^{1}pG_{x}(x,y,p)V(y)dy+\left(\frac{V(x)}{\beta(x)}\right)'\right]\right|\leq C_{7}(x)\|V\|_{C^{2}[0,1]}(2.4.7)$$

is valid for any $x \in (0,1)$, where $C_7(x)$ is some function bounded in every compact subinterval of (0,1).

Proof. We begin the proof by deducing some auxiliary relations. Similarly to the estimation of \widetilde{d}_0 in the proof of Lemma 2.4, from (2.2.42) we get $|d_0(p)| \ge \frac{1}{2l} sh(\operatorname{Re} s)$ for $\operatorname{Re} p > \sigma$ with some sufficiently large $\sigma > 0$. This relation implies

$$\left| \frac{e^{sz}}{d_0(p)} \right|, \left| \frac{sh \, sz}{d_0(p)} \right|, \left| \frac{ch \, sz}{d_0(p)} \right| \leq \operatorname{Const} \frac{|e^{sz}|}{sh(\operatorname{Re} s)} = \operatorname{Const} e^{\operatorname{Re} s(z-1)}$$

$$= \operatorname{Const} e^{-\operatorname{Re} \sqrt{p} \int_x^1 \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau} \quad \text{for } \operatorname{Re} p > \sigma,$$

$$\left| \frac{e^{s(1-z)}}{d_0(p)} \right|, \left| \frac{sh \, s(z-1)}{d_0(p)} \right|, \left| \frac{ch \, s(z-1)}{d_0(p)} \right| \leq \operatorname{Const} \frac{|e^{s(1-z)}|}{sh(\operatorname{Re} s)} = \operatorname{Const} e^{-\operatorname{Re} sz}$$

$$= \operatorname{Const} e^{-\operatorname{Re} \sqrt{p} \int_0^x \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau} \quad \text{for } \operatorname{Re} p > \sigma.$$
(2.4.8)

Due to (2.2.20) the following inequalities

$$e^{-\operatorname{Re}\sqrt{p}\int_{x}^{1}\sqrt{\frac{\beta(\tau)}{\lambda(\tau)}}d\tau}, \ e^{-\operatorname{Re}\sqrt{p}\int_{0}^{x}\sqrt{\frac{\beta(\tau)}{\lambda(\tau)}}d\tau} \le \frac{C(\kappa, x)}{|p|^{\kappa}}$$
for any $x \in (0, 1)$, $\operatorname{Re} p > 0$, $\kappa > 0$

2.4. Asymptotical properties of Green function in case of the first kind boundary conditions

are valid with some function $C(\kappa,x)$ of $\kappa>0$ and x. Note that the function $C(\kappa,\cdot)$ is bounded in every compact subinterval of (0,1). Moreover, by (2.2.42) and (2.2.21) we have

$$\frac{sh \, s}{d_0(p)} = \frac{l \, sh \, s}{sh \, s + O\left(\frac{e^s}{s}\right)} = \frac{l}{1 + O\left(\frac{1}{s}\right)} = l + O\left(\frac{1}{s}\right) = l + O\left(\frac{1}{\sqrt{|p|}}\right)$$

$$\frac{ch \, s}{d_0(p)} = \frac{l \, ch \, s}{sh \, s + O\left(\frac{e^s}{s}\right)} = \frac{l + O\left(e^{-2s}\right)}{1 + O\left(\frac{1}{s}\right)} = l + O\left(\frac{1}{s}\right) = l + O\left(\frac{1}{\sqrt{|p|}}\right)$$
(2.4.10)

as $\operatorname{Re} p \to +\infty$ uniformly in $\operatorname{Im} p$.

Due to the representation (2.2.49), we can decompose G_x as follows:

$$G_x(x, y, p) = \mathcal{G}_1(x, y, p) + \mathcal{G}_2(x, y, p) + \mathcal{G}_3(x, y, p),$$
 (2.4.11)

where

$$\mathcal{G}_1(x,y,p) = \frac{1}{ld_0(p)} \frac{b(x)}{a(y)} \begin{cases} ch \ sz \cdot sh \ s(w-1) & \text{for } x < y \\ sh \ sw \cdot ch \ s(z-1) & \text{for } x > y, \end{cases}$$
(2.4.12)

$$\mathcal{G}_2(x, y, p) = \frac{1}{ld_0(p)} \frac{b(x)}{a(y)} \frac{1}{s}$$

$$\times \left\{ \begin{array}{ll} \left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x)\right)sh\ sz \cdot sh\ s(w-1) & \text{for } x < y\\ \left(\frac{\zeta(x) - \zeta(1)}{2} - l\bar{b}(x)a'(x)\right)sh\ sw \cdot sh\ s(z-1) & \text{for } x > y, \end{array} \right. \tag{2.4.13}$$

$$\mathcal{G}_3(x,y,p) = \frac{1}{d_0(p)} \frac{1}{s}$$

$$\times \begin{cases} \frac{b(x)}{l} \frac{\zeta(y) - \zeta(1)}{2} ch \ sz \cdot ch \ s(w - 1) + \mathcal{O}_1 & \text{for } x < y \\ \frac{b(x)}{l} \frac{\zeta(y)}{2} ch \ sw \cdot ch \ s(z - 1) + \mathcal{O}_2 & \text{for } x > y \end{cases}$$
(2.4.14)

and $\mathcal{O}_1 = \frac{b(x)}{l\,a(y)}O_9$ and $\mathcal{O}_2 = \frac{b(x)}{l\,a(y)}O_{10}$ behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$.

We apply twice Lemma 2.5 to the integral $\int_0^1 \mathcal{G}_1(x,y,p)V(y)dy$ and use the relation $\frac{1}{aw'}=l\bar{b}$ following from (2.2.6), (2.2.44) to get

$$\int_{0}^{1} \mathcal{G}_{1}(x,y,p)V(y)dy = \mathcal{D}_{1}[V](x,p) + \int_{0}^{1} \mathcal{T}_{1}[V](x,y,p)dy \qquad (2.4.15)$$

2. Functional spaces and properties of Green function

with

$$\mathcal{D}_{1}[V](x,p) = \frac{b(x)}{d_{0}(p)} \left\{ \frac{1}{s} \left[ch \, sz \cdot ch \, s(w-1)\bar{b}(y)V(y) \Big|_{y=x}^{y=1} \right. \right. \\ \left. + ch \, sw \cdot ch \, s(z-1)\bar{b}(y)V(y) \Big|_{y=0}^{y=x} \right] \\ \left. - \frac{1}{s^{2}} \left[ch \, sz \cdot sh \, s(w-1) \frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)} \Big|_{y=x}^{y=1} \right. \\ \left. + sh \, sw \cdot ch \, s(z-1) \frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)} \Big|_{y=0}^{y=x} \right] \right\},$$

$$(2.4.16)$$

$$\mathcal{T}_{1}[V](x, y.p) = \frac{b(x)}{s^{2}d_{0}(p)} \left(\frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)}\right)' \begin{cases} ch \ sz \cdot sh \ s(w-1) & \text{for } x < y \\ sh \ sw \cdot ch \ s(z-1) & \text{for } x > y. \end{cases}$$
(2.4.17)

Applying once Lemma 2.5 to the integrals $\int_0^1 \mathcal{G}_2(x,y,p)V(y)dy$ and $\int_0^1 \mathcal{G}_3(x,y,p)V(y)dy$ we similarly get

$$\int_0^1 \mathcal{G}_k(x,y,p)V(y)dy = \mathcal{D}_k[V](x,p) - \int_0^1 \mathcal{T}_k[V](x,y,p)dy, \ k = 2, 3, (2.4.18)$$

where

$$\mathcal{D}_{2}[V](x,p) = \frac{b(x)}{s^{2}d_{0}(p)} \left\{ \left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) \right) sh \, sz \cdot ch \, s(w-1) \, \bar{b}(y)V(y) \Big|_{y=x}^{y=1} + \left(\frac{\zeta(x) - \zeta(1)}{2} - l\bar{b}(x)a'(x) \right) ch \, sw \cdot sh \, s(z-1) \, \bar{b}(y)V(y) \Big|_{y=0}^{y=x} \right\},$$
(2.4.19)

$$\mathcal{D}_{3}[V](x,p) = \frac{b(x)}{s^{2}d_{0}(p)} \left\{ \left[\frac{\zeta(y) - \zeta(1)}{2} \bar{b}(y) ch sz \cdot sh s(w-1) + \mathcal{O}_{3}(x,y,p) \right] V(y) \Big|_{y=x}^{y=1} \right. \\ \left. + \left[\frac{\zeta(y)}{2} \bar{b}(y) sh sw \cdot ch s(z-1) + \mathcal{O}_{4}(x,y,p) \right] V(y) \Big|_{y=0}^{y=x} \right\},$$

$$\mathcal{T}_{2}[V](x,y,p) = \frac{b(x)}{s^{2}d_{0}(p)} \times \begin{cases}
\left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x)\right) \left(\bar{b}(y)V(y)\right)'sh \, sz \cdot ch \, s(w-1), & x < y \\
\left(\frac{\zeta(x) - \zeta(1)}{2} - l\bar{b}(x)a'(x)\right) \left(\bar{b}(y)V(y)\right)'ch \, sw \cdot sh \, s(z-1), & x > y
\end{cases} (2.4.21)$$

2.4. Asymptotical properties of Green function in case of the first kind boundary conditions

$$\mathcal{T}_{3}[V](x,y,p) = \frac{1}{s^{2}d_{0}(p)} \times \begin{cases}
b(x) \left(\frac{\zeta(y) - \zeta(1)}{2} \bar{b}(y)V(y)\right)' ch \, sz \cdot sh \, s(w-1) + V'(y)\mathcal{O}_{3}, & x < y \\
b(x) \left(\frac{\zeta(y)}{2} \bar{b}(y)V(y)\right)' sh \, sw \cdot ch \, s(z-1) + V'(y)\mathcal{O}_{4}, & x > y
\end{cases} (2.4.22)$$

and \mathcal{O}_3 , \mathcal{O}_4 given by (2.3.8) behave like O_1 , O_2 , respectively as $\operatorname{Re} p \to +\infty$. Summing up, by (2.4.11), (2.4.15), (2.4.18) we have

$$\int_0^1 G_x(x,y,p)V(y)dy = \mathcal{D}[V](x,p) + \int_0^1 \mathcal{T}[V](x,y,p)dy \qquad (2.4.23)$$

where $\mathcal{D}[V] = \mathcal{D}_1[V] + \mathcal{D}_2[V] + \mathcal{D}_3[V]$ and $\mathcal{T}[V] = \mathcal{T}_1[V] - \mathcal{T}_2[V] - \mathcal{T}_3[V]$. Let us estimate the quantity $\int_0^1 \mathcal{T}[V](x,y,p)dy$ using Lemma 2.4 for the integrals of functions $\mathcal{T}_1,\mathcal{T}_2,\mathcal{T}_3$ given by (2.4.17), (2.4.21), (2.4.22). This lemma with the relation $s = l\sqrt{p}$ implies

$$\left| \int_0^1 \mathcal{T}[V](x,y,p) dy \right| \le \frac{\operatorname{Const} ||V||_{C^2[0,1]}}{|p|^{3/2}}, \quad x \in [0,1], \ \operatorname{Re} p > 0. (2.4.24)$$

Further, computing $\mathcal{D}=\mathcal{D}_1+\mathcal{D}_2+\mathcal{D}_3$ from (2.4.16), (2.4.19), (2.4.20), simplifying by means of the relations w(0)=0, w(1)=1, w(x)=z and reordering the terms we obtain

$$\mathcal{D}[V] = D_{right}[V] + D_{left}[V] + D_{diag}[V] + \widetilde{\mathcal{D}}[V], \qquad (2.4.25)$$

where

$$D_{right}[V](x,p) = \frac{b(x)}{sd_0(p)}\bar{b}(1)V(1)ch \ sz ,$$

$$D_{left}[V](x,p) = -\frac{b(x)}{sd_0(p)}\bar{b}(0)V(0)ch \ s(z-1),$$
(2.4.26)

$$D_{diag}[V](x,p) = \frac{b(x)}{s^2 d_0(p)} sh s \left[l(\bar{b}(x))^2 a'(x) V(x) - \frac{(\bar{b}(x)V(x))'}{w'(x)} \right], (2.4.27)$$

$$\widetilde{\mathcal{D}}[V](x,p) = \frac{b(x)}{s^{2}d_{0}(p)} \left\{ \left[\left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) \right) sh \, sz \, \bar{b}(1) + \mathcal{O}_{3}(x,1,p) \right] V(1) - \left[\left(\frac{\zeta(x) - \zeta(1)}{2} - l\bar{b}(x)a'(x) \right) sh \, s(z-1) \, \bar{b}(0) + \mathcal{O}_{4}(x,0,p) \right] V(0) \right. \\
\left. + \left(\mathcal{O}_{4}(x,x,p) - \mathcal{O}_{3}(x,x,p) \right) V(x) \right\}.$$

2. Functional spaces and properties of Green function

Let $x \in (0,1)$. From the formulas (2.4.26) by means of the inequalities (2.4.8) and (2.4.9) with $\kappa = 1$ we obtain

$$|D_{right}[V](x,p)|, |D_{left}[V](x,p)| \le \frac{C^1(x) \|V\|_{C[0,1]}}{|p|^{3/2}} \text{ for } \operatorname{Re} p > \sigma(2.4.29)$$

with a function C^1 bounded in every compact subinterval of (0,1). Further, by the behavior of \mathcal{O}_3 and \mathcal{O}_4 (cf. (2.2.45)) we have $\mathcal{O}_3(x,x,p), \mathcal{O}_4(x,x,p) = O\left(\frac{e^s}{s}\right)$ as $\operatorname{Re} p \to +\infty$ uniformly in x and $\operatorname{Im} p$. Using this relation and (2.4.8), (2.4.9) in (2.4.28) and observing the equality $s = l\sqrt{p}$ and (2.4.9) with $\kappa = 1/2$ we deduce the estimate

$$\begin{split} |\widetilde{\mathcal{D}}[V](x,p)| \\ &\leq \frac{\operatorname{Const}}{|p|} \left[e^{-\operatorname{Re}\sqrt{p} \int_{x}^{1} \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau} |V(1)| + e^{-\operatorname{Re}\sqrt{p} \int_{0}^{x} \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau} |V(0)| \right. \\ &\left. + \frac{1}{\sqrt{|p|}} |V(x)| \right] \leq \frac{\operatorname{C}^{2}(x) \, \|V\|_{C[0,1]}}{|p|^{3/2}} \quad \text{for } \operatorname{Re} p > \sigma, \end{split}$$

where C^2 bounded in every compact subinterval of (0, 1).

Finally, let us consider $D_{diag}[V](x,p)$. Due to the definitions (2.2.6), (2.2.7), (2.2.44) of a,b,\bar{b},z,w we can further simplify (2.4.27) as follows: $D_{diag}[V](x,p) = -\frac{l\,sh\,s}{s^2d_0(p)}\left(\frac{V(x)}{\beta(x)}\right)'$. Thus, by (2.4.10) we have

$$D_{diag}[V](x,p)$$

$$=-\frac{l}{s^2}\left[l+O\left(\frac{1}{\sqrt{|p|}}\right)\right]\left(\frac{V(x)}{\beta(x)}\right)'=-\frac{1}{p}\left(\frac{V(x)}{\beta(x)}\right)'+O\left(\frac{1}{|p|^{3/2}}\right)\|V\|_{C^1[0,1]}, (2.4.31)$$
 as $\operatorname{Re} p\to +\infty$ uniformly in $x\in[0,1]$ and $\operatorname{Im} p$.

Let us sum up. Using (2.4.24), (2.4.25) and (2.4.29) - (2.4.31) in (2.4.23) we obtain the estimate (2.4.7) with the inequality $\operatorname{Re} p > \sigma$ instead of $\operatorname{Re} p > 0$ under the sign of supremum. But we can replace the inequality $\operatorname{Re} p > \sigma$ by $\operatorname{Re} p > 0$ there, because owing to the boundedness of G_x in the strip S_σ , the estimate (2.4.7) hold trivially in case the range of the supremum is $0 < \operatorname{Re} p < \sigma$. The theorem is proved.

2.5 Asymptotical properties of Green function in case of the third kind boundary conditions

First we prove estimates of G and its derivatives.

Theorem 2.5. Let λ , β satisfy (2.2.2). Then the Green function G of the operator L in [0,1] with the third kind boundary conditions satisfies the estimates (2.4.1) - (2.4.4). Moreover, for the boundary values of G the relations

$$\sup_{\text{Re }p>0} |p|^{\kappa} |G(x,1,p)|, \ \sup_{\text{Re }p>0} |p|^{\kappa} |G(x,0,p)| \le C_{\kappa}^{1}(x)$$
 (2.5.1)

2.5. Asymptotical properties of Green function in case of the third kind boundary conditions

hold for any $x \in (0,1)$ and $\kappa > 0$, where $C_{\kappa}^{1}(x)$ is a function bounded in every compact subinterval of (0,1).

Proof. The estimates (2.4.1) - (2.4.4) were proved in [24]. They follow from Lemma 2.4, if we take into account the relations (2.2.57) - (2.2.60) for G, G_x , G_y , G_{xy} , the positivity and boundendness of a and b (cf. (2.2.6) and assumptions of Theorem 2.5) as well as the equality $s = l\sqrt{p}$.

To prove (2.5.1) we use the representation (2.2.57) for G. Since w(1) = 1, w(0) = 0, from this representation we get

$$G(x,1,p) = \frac{1}{sd_1(p)a(x)a(1)} \left[ch \, sz + \hat{O}_1(x,1,p) \right] ,$$

$$G(x,0,p) = \frac{1}{sd_1(p)a(x)a(0)} \left[ch \, s(z-1) + \hat{O}_2(x,0,p) \right] .$$
(2.5.2)

Comparing (2.2.56) with (2.2.42) we see that $d_1(p)$ has the same asymptotical behavior as $-d_0(p)$. This implies that the inequalities (2.4.8) hold with $d_0(p)$ replaced by $d_1(p)$ and instead of (2.4.10) we have the formulae

$$\frac{sh\,s}{d_1(p)},\,\frac{ch\,s}{d_1(p)}=-l+O\left(\frac{1}{\sqrt{|p|}}\right)\ \text{ as } \operatorname{Re} p\to +\infty \text{ uniformly in } \operatorname{Im} p.\ (2.5.3)$$

Observing the relations $\hat{O}_1(x,1,p) = O\left(\frac{e^{sz}}{s}\right)$, $\hat{O}_2(x,0,p) = O\left(\frac{e^{s(1-z)}}{s}\right)$ (cf. (2.2.45)), the inequalities (2.4.8) with $d_1(p)$ instead of $d_0(p)$ and the relation $s = l\sqrt{p}$ from (2.5.2) we deduce

$$|G(x,1,p)| \le \frac{\operatorname{Const}}{\sqrt{|p|}} e^{-\operatorname{Re}\sqrt{p} \int_{x}^{1} \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau},$$

$$|G(x,0,p)| \le \frac{\operatorname{Const}}{\sqrt{|p|}} e^{-\operatorname{Re}\sqrt{p} \int_{0}^{x} \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau}$$
(2.5.4)

for any $x \in (0,1)$ and $\operatorname{Re} p > \sigma$ where $\sigma > 0$ is sufficiently large. These relations with (2.4.9) and boundedness of G in S_{σ} imply (2.5.1).

Next, let us formulate two theorems proved in [24].

Theorem 2.6. Let λ , β satisfy (2.2.2) and G be the Green function of the operator L in [0,1] with the third kind boundary conditions. If $V \in C^1[0,1]$, then the estimate

$$\sup_{\text{Re }p>0} \left| \sqrt{|p|} \left(\int_0^1 G_{xy}(x,y,p) V(y) dy - \frac{V(x)}{\lambda(x)} \right) \right| \le C_8(x) \|V\|_{C^1[0,1]} \quad (2.5.5)$$

holds for any $x \in (0,1)$, where $C_8(x)$ is some function bounded in every compact subinterval of (0,1).

2. Functional spaces and properties of Green function

Theorem 2.7. Let λ , β satisfy (2.2.2) and G be the Green function of the operator L in [0,1] with the third kind boundary conditions. If $V \in C^1[0,1]$, then the estimate

$$\sup_{\text{Re }p>0} \left| \sqrt{|p|} \left(\int_0^1 pG(x,y,p)V(y)dy + \frac{V(x)}{\beta(x)} \right) \right| \le C_9 ||V||_{C^1[0,1]}$$
 (2.5.6)

holds for any $x \in [0, 1]$, where C_9 is a positive constant.

Finally, we prove the following theorem.

Theorem 2.8. Let λ, β satisfy (2.2.2) and G be the Green function of the operator L in [0,1] with the third kind boundary conditions. If $V \in C^2[0,1]$, then the estimate

$$\sup_{\text{Re }p>0}\left|\sqrt{|p|}\left[\int_{0}^{1}pG_{x}(x,y,p)V(y)dy+\left(\frac{V(x)}{\beta(x)}\right)'\right]\right|\leq C_{10}(x)\|V\|_{C^{2}[0,1]} \ (2.5.7)$$

is valid for any $x \in (0,1)$, where $C_{10}(x)$ is some function bounded in every compact subinterval of (0,1).

Proof is similar to the proof of Theorem 2.4. Due to the representation (2.2.61) we can decompose G_x as (2.4.11), where

$$\mathcal{G}_1(x, y, p) = \frac{1}{ld_1(p)} \frac{b(x)}{a(y)} \begin{cases} sh sz \cdot ch s(w-1) & \text{for } x < y \\ ch sw \cdot sh s(z-1) & \text{for } x > y, \end{cases}$$
(2.5.8)

$$\mathcal{G}_{2}(x,y,p) = \frac{1}{ld_{1}(p)} \frac{b(x)}{a(y)} \frac{1}{s}$$

$$\times \begin{cases}
\left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) + \vartheta_{0}\right) ch \, sz \cdot ch \, s(w-1) & \text{for } x < y \\
\left(\frac{\zeta(x) - \zeta(1)}{2} - l\bar{b}(x)a'(x) + \vartheta_{1}\right) ch \, sw \cdot ch \, s(z-1) & \text{for } x > y,
\end{cases} (2.5.9)$$

$$\mathcal{G}_{3}(x,y,p) = \frac{1}{d_{1}(p)} \frac{1}{s}$$

$$\times \begin{cases}
\frac{b(x)}{l \, a(y)} \left(\frac{\zeta(y) - \zeta(1)}{2} + \vartheta_{1}\right) sh \, sz \cdot sh \, s(w-1) + \mathcal{O}_{1} & \text{for } x < y \\
\frac{b(x)}{l \, a(y)} \left(\frac{\zeta(y)}{2} + \vartheta_{0}\right) sh \, sw \cdot sh \, s(z-1) + \mathcal{O}_{2} & \text{for } x > y,
\end{cases} (2.5.10)$$

the numbers ϑ_0 , ϑ_1 are given by (2.2.62) and $\mathcal{O}_1 = \frac{b(x)}{l\,a(y)}\hat{O}_9$ and $\mathcal{O}_2 = \frac{b(x)}{l\,a(y)}\hat{O}_{10}$ behave like O_1 and O_2 , respectively, as $\operatorname{Re} p \to +\infty$.

Applying twice Lemma 2.5 to the integral $\int_0^1 \mathcal{G}_1(x,y,p)V(y)dy$ and using the

2.5. Asymptotical properties of Green function in case of the third kind boundary conditions

relation $\frac{1}{aw'} = l\bar{b}$ we obtain (2.4.15) with

$$\mathcal{D}_{1}[V](x,p) = \frac{b(x)}{d_{1}(p)} \left\{ \frac{1}{s} \left[sh \, sz \cdot sh \, s(w-1)\bar{b}(y)V(y) \Big|_{y=x}^{y=1} \right. \right. \\ \left. + sh \, sw \cdot sh \, s(z-1)\bar{b}(y)V(y) \Big|_{y=0}^{y=x} \right] \\ \left. - \frac{1}{s^{2}} \left[sh \, sz \cdot ch \, s(w-1) \frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)} \Big|_{y=x}^{y=1} \right. \\ \left. + ch \, sw \cdot sh \, s(z-1) \frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)} \Big|_{y=0}^{y=x} \right] \right\},$$

$$\left. + ch \, sw \cdot sh \, s(z-1) \frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)} \Big|_{y=0}^{y=x} \right] \right\},$$

$$\left. (2.5.11)$$

$$\mathcal{T}_1[V](x,y,p)$$

$$= \frac{b(x)}{s^2 d_1(p)} \left(\frac{\left(\overline{b}(y)V(y)\right)'}{w'(y)} \right)' \begin{cases} sh sz \cdot ch s(w-1) & \text{for } x < y \\ ch sw \cdot sh s(z-1) & \text{for } x > y. \end{cases} (2.5.12)$$

Using once Lemma 2.5 for integrals $\int_0^1 \mathcal{G}_k(x,y,p)V(y)dy$, k=2,3 we get (2.4.18), where

$$\mathcal{D}_{2}[V](x,p) = \frac{b(x)}{s^{2}d_{1}(p)} \times \left\{ \left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) + \vartheta_{0} \right) ch \ sz \cdot sh \ s(w-1) \ \bar{b}(y)V(y) \Big|_{y=x}^{y=1} \right.$$

$$\left. + \left(\frac{\zeta(x) - \zeta(1)}{2} - l\bar{b}(x)a'(x) + \vartheta_{1} \right) sh \ sw \cdot ch \ s(z-1) \ \bar{b}(y)V(y) \Big|_{y=0}^{y=x} \right\},$$

$$(2.5.13)$$

$$\mathcal{D}_{3}[V](x,p) = \frac{b(x)}{s^{2}d_{1}(p)} \times \left\{ \left[\left(\frac{\zeta(y) - \zeta(1)}{2} + \vartheta_{1} \right) \bar{b}(y) sh \, sz \cdot ch \, s(w-1) + \mathcal{O}_{3}(x,y,p) \right] V(y) \Big|_{y=x}^{y=1} \right. \\ \left. + \left[\left(\frac{\zeta(y)}{2} + \vartheta_{0} \right) \bar{b}(y) ch \, sw \cdot sh \, s(z-1) + \mathcal{O}_{4}(x,y,p) \right] V(y) \Big|_{y=0}^{y=x} \right\},$$

$$\mathcal{T}_{2}[V](x,y,p) = \frac{b(x)}{s^{2}d_{1}(p)} \times \begin{cases}
\left(\frac{\zeta(x)}{2} - l\bar{b}(x)a'(x) + \vartheta_{0}\right) \left(\bar{b}(y)V(y)\right)'ch\,sz \cdot sh\,s(w-1), & x < y \\
\left(\frac{\zeta(x) - \zeta(1)}{2} - l\bar{b}(x)a'(x) + \vartheta_{1}\right) \left(\bar{b}(y)V(y)\right)'sh\,sw \cdot ch\,s(z-1), & x > y,
\end{cases} (2.5.15)$$

$$\mathcal{T}_{3}[V](x,y.p) = \frac{b(x)}{s^{2}d_{1}(p)}$$

$$\times \begin{cases} \left(\left(\frac{\zeta(y)-\zeta(1)}{2} + \vartheta_{1}\right)\bar{b}(y)V(y)\right)'sh\ sz \cdot ch\ s(w-1) + \frac{V'(y)}{b(x)}\mathcal{O}_{3}, \ x < y \\ \left(\left(\frac{\zeta(y)}{2} + \vartheta_{0}\right)\bar{b}(y)V(y)\right)'ch\ sw \cdot sh\ s(z-1) + \frac{V'(y)}{b(x)}\mathcal{O}_{4}, \qquad x > y \end{cases}$$

$$(2.5.16)$$

2. Functional spaces and properties of Green function

and \mathcal{O}_3 , \mathcal{O}_4 given by (2.3.8) behave like \mathcal{O}_1 , \mathcal{O}_2 , respectively as $\operatorname{Re} p \to +\infty$.

By (2.4.11), (2.4.15), (2.4.18) we have the relation (2.4.23) with $\mathcal{D}[V] = \mathcal{D}_1[V] + \mathcal{D}_2[V] + \mathcal{D}_3[V]$ and $\mathcal{T}[V] = \mathcal{T}_1[V] - \mathcal{T}_2[V] - \mathcal{T}_3[V]$. Applying Lemma 2.4 for the integrals of functions $\mathcal{T}_1, \mathcal{T}_2, \mathcal{T}_3$ given by (2.4.17), (2.4.21), (2.4.22) and using the relation $s = l\sqrt{p}$ we get for $\mathcal{T}[V]$ the estimate (2.4.24).

Further, computing $\mathcal{D}=\mathcal{D}_1+\mathcal{D}_2+\mathcal{D}_3$ from (2.5.11), (2.5.13), (2.5.14), simplifying by means of the relations w(0)=0, w(1)=1, w(x)=z and $\zeta(0)=0$ (see (2.2.44), (2.2.8)) and reordering the terms we obtain

$$\mathcal{D}[V] = D_{diag}[V] + \widetilde{\mathcal{D}}[V], \qquad (2.5.17)$$

where

$$D_{diag}[V](x,p) = -\frac{b(x)}{s^2 d_1(p)} sh s \left[l(\bar{b}(x))^2 a'(x) V(x) - \frac{(\bar{b}(x)V(x))'}{w'(x)} \right], (2.5.18)$$

$$\widetilde{\mathcal{D}}[V](x,p) = -\frac{b(x)}{s^{2}d_{1}(p)} \times \left\{ -\frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)} \Big|_{y=1} sh \, sz + \frac{\left(\bar{b}(y)V(y)\right)'}{w'(y)} \Big|_{y=0} sh \, s(z-1) + \left[\vartheta_{1}\bar{b}(1) \, sh \, sz + \mathcal{O}_{3}(x,1,p)\right] V(1) - \left[\vartheta_{0}\bar{b}(0) sh \, s(z-1) - \mathcal{O}_{4}(x,0,p)\right] V(0) + (\mathcal{O}_{4}(x,x,p) - \mathcal{O}_{3}(x,x,p))V(x) \right\}.$$
(2.5.19)

Let $x\in(0,1)$. Due to the behavior of \mathcal{O}_3 and \mathcal{O}_4 (cf. (2.2.45)), we have $\mathcal{O}_3(x,1,p)=O\left(\frac{e^{sz}}{s}\right)$, $\mathcal{O}_4(x,0,p)=O\left(\frac{e^{s(1-z)}}{s}\right)$, $\mathcal{O}_3(x,x,p)$, $\mathcal{O}_4(x,x,p)=O\left(\frac{e^s}{s}\right)$. Using these relations and (2.4.8) (with $d_1(p)$ instead of $d_0(p)$) in (2.5.19) and observing the equality $s=l\sqrt{p}$ and (2.4.9) with $\kappa=1/2$ we deduce the estimate

$$|\widetilde{\mathcal{D}}[V](x,p) \leq \frac{\text{Const}}{|p|} \left[e^{-\text{Re}\sqrt{p} \int_{x}^{1} \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau} (|V(1)| + |V'(1)|) + e^{-\text{Re}\sqrt{p} \int_{0}^{x} \sqrt{\frac{\beta(\tau)}{\lambda(\tau)}} d\tau} (|V(0)| + |V'(0)|) + \frac{1}{\sqrt{|p|}} |V(x)| \right] \leq \frac{C^{3}(x) \|V\|_{C^{1}[0,1]}}{|p|^{3/2}} \quad \text{for } \text{Re } p > \sigma,$$

where C^3 bounded in every compact subinterval of (0, 1).

Finally, we consider $D_{diag}[V](x,p)$. As in the proof of Theorem 2.4 we simplify the expression of $D_{diag}[V](x,p)$. From (2.5.18) we get $D_{diag}[V](x,p) = \frac{l \, sh \, s}{s^2 d_1(p)} \left(\frac{V(x)}{\beta(x)}\right)'$. Using here (2.5.3) we obtain (2.4.31).

Summing up, the relations (2.4.24) and (2.5.17), (2.5.20), (2.4.31) with (2.4.23) and the boundedness of G_x in S_σ yield (2.5.7). The proof is complete.

In this chapter we study the generalized inverse problem with temperature observations.

3.1 Reduction of the inverse problem to a fixed-point form

Here we deduce a fixed-point system for the inverse problem in the Laplace domain and transform further the system for U and U_x . Recall that this problem was formulated as the system (1.2.25) with the additional equations (1.2.15), (1.2.18).

Using the relation (1.2.15) for U in the left-hand side of the conditions (1.2.25) we obtain

$$\sum_{k=1}^{K_1} p N_k(p) \int_0^1 p G(x_i, y, p) \nu_k(y) p U(y, p) \, dy$$

$$-\sum_{k=1}^{K_2} \sqrt{p} M_k(p) \sqrt{p} \int_0^1 G_y(x_i, y, p) \mu_k(y) p U_y(y, p) \, dy$$

$$= p^2 [H_i(p) + Q(x_i, p)], \quad i = 1, \dots, K.$$
(3.1.1)

Conversely, using (1.2.15) in (3.1.1) we deduce (1.2.25). Therefore, the system (1.2.25) with (1.2.15), (1.2.18), is equivalent to the system (3.1.1) with (1.2.15), (1.2.18).

The latter system (3.1.1) is more convenient for our analysis because we can rewrite it in a fixed-point form. The main idea is to separate the principal part of the system in the process $\operatorname{Re} p \to +\infty$. In case such a the principal part contains unknowns N and M, we leave it to the left hand side of the system bringing the remainder to the right-hand side. The remainder is already a term of a lower order, i.e. a contractive mapping of the unknowns in some half-plane $\operatorname{Re} p > \sigma$ with sufficiently large σ . We note that transforming the system (3.1.1) into the fixed-point form is rather a formal procedure and doesn't require regularity assumptions on the data and solutions (except for (3.1.4), $\det \Gamma \neq 0$ and the decomposition (3.1.16) of the function Φ^0 below). However, ideas of extracting this principal part are based on certain proper asymptotics of the system in the process $\operatorname{Re} p \to +\infty$. Such an asymptotics is achieved under the conditions that the system (3.1.1) with (1.2.15), (1.2.18) has a solution $N_k|_{k=1,\ldots,K_1}$, $M_k|_{k=1,\ldots,K_2}$, U that satisfies the following properties:

- (1) $n_k(t) = \mathcal{L}_{p\to t}^{-1} N_k(p)$ and n_k' belong to $\mathscr E$ implying $pN_k(p) \to n_k(0)$, as $\operatorname{Re} p \to +\infty$ (cf. property 1 of the Laplace transform in Section 1.2.1);
- (2) $\sqrt{p}M_k(p) \to 0$, as $\text{Re}p \to +\infty$ (strengthened form of (1.2.3));
- (3) for $u(x,t) = \mathcal{L}_{p\to t}^{-1}U(x,p)$ the relations $u(x,\cdot), u_t(x,\cdot), u_x(x,\cdot), u_{xt}(x,\cdot) \in \mathscr{E}$ hold with the initial condition (1.1.5) implying

$$pU(x,p) \to \varphi(x)$$
, $pU_x(x,p) \to \varphi'(x)$, as $\operatorname{Re} p \to \infty$. (3.1.2)

We emphasize that (1) - (3) are not assumptions. We only use them to explain the ideas of the transformation of the system (3.1.1).

In case (1) - (3) hold by the assertions of Theorems 2.1, 2.3, 2.5, 2.7 we see that the first sum on the left-hand side of (3.1.1) dominates with respect to the second sum in the process $\operatorname{Re} p \to +\infty$. As a result in the process $\operatorname{Re} p \to +\infty$ from (3.1.1) we obtain the following system for $n_k^0 = n_k(0)$:

$$-\sum_{k=1}^{K_1} n_k^0 \frac{\nu_k(x_i)\varphi(x_i)}{\beta(x_i)} = \lim_{\text{Re}p\to\infty} p^2 \left[H_i(p) + Q(x_i, p) \right], \ i = 1, \dots, K. \ (3.1.3)$$

Since this system doesn't contain the full unknowns N(p) and M(p), we have to compute the second approximation of the system (3.1.1) in the process $\operatorname{Re} p \to +\infty$, too. To this end we have to assume that (3.1.3) is solvable. According to the Kronecker theorem the system (3.1.1) has a solution provided

$$\mathcal{R} := \operatorname{rank} \left(\left(\frac{\nu_k(x_i)\varphi(x_i)}{\beta(x_i)} \right)_{k=1,\dots,K_1}, \lim_{\operatorname{Re}p \to \infty} p^2 \left[H_i(p) + Q(x_i, p) \right] \right)_{i=1,\dots,K}$$

$$= \operatorname{rank} \left(\frac{\nu_k(x_i)\varphi(x_i)}{\beta(x_i)} \right)_{k=1,\dots,K_1 \atop i=1,\dots,K}. \tag{3.1.4}$$

Here the matrix in the left-hand side is formed by placing two matrices left to right. In the sequel we assume that the solvability condition (3.1.4) for the system (3.1.3) is satisfied. In case $\mathcal{R} = K_1$ the solution of (3.1.3) is even unique.

Let us continue the transformation of (3.1.1). As we remarked, under the conditions (1) - (3) the first sum on the left-hand side of (3.1.1) dominates with respect to the second sum in the process $\operatorname{Re} p \to +\infty$. This suggests that the kernels n_k and m_k can be determined simultaneously with higher smoothness in n_k than in m_k . Therefore, we introduce the new unknowns

$$Z = (Z_1, \dots, Z_K), \ Z_k(p) = \begin{cases} pN_k(p) - n_k^0, & k = 1, \dots, K_1, \\ M_{k-K_1}(p), & k = K_1 + 1, \dots, K, \end{cases}$$
(3.1.5)

where n_k^0 , $k = 1, ..., K_1$ solves (3.1.3). To find the second-order approximation of (3.1.1) we multiply it by p. Then, in view of (3.1.3) and (3.1.5), (1) - (3) and

Theorems 2.1, 2.3, 2.5, 2.7 we see that the system (3.1.1) is in the process $\operatorname{Re} p \to +\infty$ is asymptotically equivalent to the system

$$\sum_{k=1}^{K_1} Z_k(p) \frac{1}{\beta(x_i)} \nu_k(x_i) \varphi(x_i) + \sum_{k=K_1+1}^{K} Z_k(p) \frac{1}{\beta(x_i)} \left(\mu_{k-K_1}(x) \varphi'(x) \right)' \Big|_{x=x_i} = 0.$$

The left-hand side of this system represents the principal part of (3.1.1). Thus, separating the principal part we after some elementary transformations rewrite (3.1.1) in the following equivalent form:

$$\sum_{k=1}^{K_{1}} Z_{k}(p) \frac{1}{\beta(x_{i})} \nu_{k}(x_{i}) \varphi(x_{i}) + \sum_{k=K_{1}+1}^{K} Z_{k}(p) \frac{1}{\beta(x_{i})} \left(\mu_{k-K_{1}}(x)\varphi'(x)\right)' \Big|_{x=x_{i}}$$

$$= \sum_{k=1}^{K_{1}} Z_{k}(p) \left[\int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \left[pU(y, p) - \varphi(y) \right] dy \right]$$

$$+ \int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \varphi(y) dy + \frac{1}{\beta(x_{i})} \nu_{k}(x_{i}) \varphi(x_{i}) \right]$$

$$- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left[\int_{0}^{1} pG_{y}(x_{i}, y, p) \mu_{k-K_{1}}(y) \left[pU_{y}(y, p) - \varphi'(y) \right] dy$$

$$- \int_{0}^{1} pG(x_{i}, y, p) \left(\mu_{k-K_{1}}(y) \varphi'(y) \right)' dy - \frac{1}{\beta(x_{i})} \left(\mu_{k-K_{1}}(x) \varphi'(x) \right)' \Big|_{x=x_{i}}$$

$$+ \mu_{k-K_{1}}(0) \varphi'(0) pG(x_{i}, 0, p) - \mu_{k-K_{1}}(1) \varphi'(1) pG(x_{i}, 1, p) \right]$$

$$+ \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \left[pU(y, p) - \varphi(y) \right] dy$$

$$- p^{2} \left[H_{i}(p) + Q(x_{i}, p) \right] + \lim_{Ren \to \infty} q^{2} \left[H_{i}(q) + Q(x_{i}, q) \right], i = 1, \dots, K. \quad (3.1.6)$$

The main reason of transformations in the right-hand side is to bring pU(x,p) and $pU_x(x,p)$ to the form $pU(x,p)-\varphi(x)$ and $pU_x(x,p)-\varphi'(x)$ that converge to zero in the process $\operatorname{Re} p \to +\infty$ (see (3.1.2)).

Further, let us introduce the matrix

$$\Gamma = (\gamma_{ik})_{i,k=1,\dots,K}, \ \gamma_{ik} = \begin{cases} \frac{\nu_k(x_i)\varphi(x_i)}{\beta(x_i)}, & k = 1,\dots,K_1, \\ \frac{(\mu_{k-K_1}(y)\varphi'(y))'}{\beta(x_i)}|_{y=x_i}, & k = K_1 + 1,\dots,K \end{cases}$$
(3.1.7)

related to the principal part and assume $\det \Gamma \neq 0$. Moreover, we introduce the functions

$$B^{0}[Z](x,p) = pU[Z](x,p) - \varphi(x), \ B^{1}[Z](x,p) = pU_{x}[Z](x,p) - \varphi'(x), (3.1.8)$$

where U[Z](x,p) is the solution of (1.2.15), (1.2.18) with the given vector Z in the form (3.1.5).

Now we are ready to rewrite system (3.1.6) in the equivalent fixed-point form

$$Z = \Gamma^{-1} \mathcal{F}(Z) \,, \tag{3.1.9}$$

where $\mathcal{F}(Z) = (\mathcal{F}_1(Z), \dots, \mathcal{F}_K(Z)),$

$$\mathcal{F}_{i}[Z](p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \left[\int_{0}^{1} pG(x_{i}, y, p)\nu_{k}(y)B^{0}[Z](y, p) dy + \int_{0}^{1} pG(x_{i}, y, p)\nu_{k}(y)\varphi(y) dy + \frac{1}{\beta(x_{i})}\nu_{k}(x_{i})\varphi(x_{i}) \right] + \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left[-\int_{0}^{1} pG_{y}(x_{i}, y, p)\mu_{k-K_{1}}(y)B^{1}[Z](y, p) dy + \int_{0}^{1} pG(x_{i}, y, p) \left(\mu_{k-K_{1}}(y)\varphi'(y)\right)' dy + \frac{1}{\beta(x_{i})} \left(\mu_{k-K_{1}}(x)\varphi'(x)\right)' \Big|_{x=x_{i}} + \mu_{k-K_{1}}(0)\varphi'(0)pG(x_{i}, 0, p) - \mu_{k-K_{1}}(1)\varphi'(1)pG(x_{i}, 1, p) \right] + \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} pG(x_{i}, y, p)\nu_{k}(y)B^{0}[Z](y, p) dy + \widehat{\Psi}_{i}(p), i = 1, \dots, K \quad (3.1.10)$$

and

$$\widehat{\Psi}_{i}(p) = \sum_{k=1}^{K_{1}} n_{k}^{0} \left[\int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \varphi(y) \, dy + \frac{1}{\beta(x_{i})} \nu_{k}(x_{i}) \varphi(x_{i}) \right] - p^{2} \left[H_{i}(p) + Q(x_{i}, p) \right] + \lim_{\text{Re} a \to \infty} q^{2} \left[H_{i}(q) + Q(x_{i}, q) \right].$$
(3.1.11)

Since the system (3.1.10) contains the quantities $B^0[Z]$ and $B^1[Z]$ we have to deduce additional equations for these quantities, too. To this end we simply rewrite the system (1.2.15), (1.2.18) for U in terms of these quantities. In view of

the definitions of Z, $B^0[Z]$ and $B^1[Z]$ from (1.2.15) we have

$$B^{0}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} G(x,y,p)\nu_{k}(y) \left[B^{0}[Z](y,p) + \varphi(y)\right] dy$$

$$- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \int_{0}^{1} G_{y}(x,y,p)\mu_{k-K_{1}}(y) \left[B^{1}[Z](y,p) + \varphi'(y)\right] dy \quad (3.1.12)$$

$$+ \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} G(x,y,p)\nu_{k}(y)B^{0}[Z](y,p) dy + \Phi^{0}(x,p)$$

with

$$\Phi^{0}(x,p) = \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} G(x,y,p) \nu_{k}(y) \varphi(y) \, dy - pQ(x,p) - \varphi(x) \quad (3.1.13)$$

and from (1.2.18) we get

$$B^{1}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} G_{x}(x,y,p)\nu_{k}(y) \left[B^{0}[Z](y,p) + \varphi(y)\right] dy$$

$$+ \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left[\frac{\mu_{k-K_{1}}(x)}{\lambda(x)} B^{1}[Z](x,p) - \int_{0}^{1} G_{xy}(x,y,p)\mu_{k-K_{1}}(y) \left[B^{1}[Z](y,p) + \varphi'(y)\right] dy\right]$$

$$+ \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} G_{x}(x,y,p)\nu_{k}(y)B^{0}[Z](y,p) dy + \sum_{k=K_{1}+1}^{K} Z_{k}(p) \frac{\mu_{k-K_{1}}(x)\varphi'(x)}{\lambda(x)}$$

$$+ \Phi^{1}(x,p)$$

with

$$\Phi^{1}(x,p) = \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} G_{x}(x,y,p) \nu_{k}(y) \varphi(y) dy - pQ_{x}(x,p) - \varphi'(x).$$
(3.1.15)

For the function $B^0[Z]$, which in contrast to $B^1[Z]$ doesn't contain a space derivative of U, we need a certain higher regularity in the time variable. To this end we assume that the data-dependent term Φ^0 given by (3.1.13) can be decomposed as follows:

$$\Phi^{0}(x,p) = \frac{c_{\Phi}(x)}{p} + \widetilde{\Phi}^{0}(x,p)$$
(3.1.16)

where c_{Φ} is some function in C[0,1]. The meaning of this decomposition is that the term $\frac{c_{\Phi}(x)}{p}$ forms the higher-order part of $\Phi^0(x,p)$ in the process $\operatorname{Re} p \to +\infty$. Indeed, in the forthcoming sections we will assume that $\widetilde{\Phi}^0 \in \hat{\mathcal{B}}_{\alpha,\sigma_0}$ with some $\alpha > 1$, $\sigma_0 \geq 0$ implying the estimate $|\widetilde{\Phi}^0(x,p)| \leq \frac{\operatorname{Const}}{|p|^{\alpha}}$, $x \in [0,1]$ $\operatorname{Re} p > \sigma_0$. But this moment such properties of $|\widetilde{\Phi}^0(x,p)|$ are not necessary to assume.

Let us split $B^0[Z]$ into the sum

$$B^{0}[Z](x,p) = \frac{c_{\Phi}(x)}{p} + \widehat{B}^{0}[Z](x,p). \tag{3.1.17}$$

From (3.1.12) and (3.1.14) in view of (3.1.13), (3.1.15) and the definitions of Z and $B^1[Z]$ we deduce the following fixed-point equation for the vector $B[Z] = (\widehat{B}^0[Z], B^1[Z])$:

$$B[Z] = A[Z]B[Z] + b[Z],$$
 (3.1.18)

where $A[Z] = (A^0[Z], A^1[Z])$ is the Z-dependent linear operator of B and $b[Z] = (b^0[Z], b^1[Z])$ is the Z-dependent B-free term with with the following components:

$$(A^{0}[Z]B)(x,p) = \sum_{k=1}^{K_{1}} (Z_{k}(p) + n_{k}^{0}) \int_{0}^{1} G(x,y,p)\nu_{k}(y)\widehat{B}^{0}(y,p) dy$$

$$- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \int_{0}^{1} G_{y}(x,y,p)\mu_{k-K_{1}}(y)B^{1}(y,p) dy, \qquad (3.1.19)$$

$$(A^{1}[Z]B)(x,p) = \sum_{k=1}^{K_{1}} (Z_{k}(p) + n_{k}^{0}) \int_{0}^{1} G_{x}(x,y,p)\nu_{k}(y)\widehat{B}^{0}(y,p) dy$$

$$+ \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left[\frac{\mu_{k-K_{1}}(x)}{\lambda(x)} B^{1}(x,p) - \int_{0}^{1} G_{xy}(x,y,p)\mu_{k-K_{1}}(y)B^{1}(y,p) dy \right], \qquad (3.1.20)$$

$$b^{0}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} G(x,y,p) \nu_{k}(y) \left[\frac{c_{\Phi}(y)}{p} + \varphi(y) \right] dy$$
$$- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \int_{0}^{1} G_{y}(x,y,p) \mu_{k-K_{1}}(y) \varphi'(y) dy + \widehat{\Phi}^{0}(x,p), \quad (3.1.21)$$

$$b^{1}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} G_{x}(x,y,p) \nu_{k}(y) \left[\frac{c_{\Phi}(y)}{p} + \varphi(y) \right] dy$$

$$+ \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left[\frac{\mu_{k-K_{1}}(x)\varphi'(x)}{\lambda(x)} \right]$$

$$- \int_{0}^{1} G_{xy}(x,y,p) \mu_{k-K_{1}}(y)\varphi'(y) dy + \widehat{\Phi}^{1}(x,p)$$
(3.1.22)

and

$$\widehat{\Phi}^{0}(x,p) = \frac{1}{p} \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} G(x,y,p) \nu_{k}(y) c_{\Phi}(y) dy + \widetilde{\Phi}^{0}(x,p), \quad (3.1.23)$$

$$\widehat{\Phi}^{1}(x,p) = \frac{1}{p} \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} G_{x}(x,y,p) \nu_{k}(y) c_{\Phi}(y) dy + \Phi^{1}(x,p).$$

Summing up, we have proved

Proposition 3.1. Let the condition (3.1.4) hold implying the existence of the solution n_k^0 , $k = 1, ..., K_1$ to the system (3.1.3). Moreover, let $\det \Gamma \neq 0$ and Φ^0 given by (3.1.13) have the representation (3.1.16). Then the inverse problem in the Laplace domain with temperature observations is equivalent to equation (3.1.9) with \mathcal{F} given by (3.1.10). The solutions of these two equivalent problems are related by (3.1.5) with (3.1.8). Moreover, the term $B^0[Z]$ in (3.1.8) has the form (3.1.17) and the vector $B[Z] = (\widehat{B}^0[Z], B^1[Z])$ consisting of the function $\widehat{B}^0[Z]$ from (3.1.17) and the term $B^1[Z]$ from (3.1.8) satisfies the system (3.1.18) with (3.1.19) - (3.1.24).

Remark. In the case of boundary conditions of the first kind the addends containing the terms $G(x_i, 1, p)$ and $G(x_i, 0, p)$ in (3.1.10) vanish because in this case $G(x_i, 1, p) = G(x_i, 0, p) = 0$ (see (1.2.12)).

3.2 Analysis of direct problem system

In this section we study the equation (3.1.18) whose solution B[Z] is connected with the solution U of the direct problem in Laplace domain.

Let us introduce the following basic assumptions on the data:

$$\Phi^0 \text{ given by } (3.1.13) \text{ admits the decomposition } (3.1.16) \text{ where } c_{\Phi} \in C[0,1] \text{ and } \widetilde{\Phi}^0 \in \hat{\mathcal{B}}_{\alpha,\sigma_0} \text{ with some } \\ \sigma_0 > 0 \text{ and } \alpha \text{ satisfying } (2.1.3); \\ \Phi^1 \text{ given by } (3.1.15) \text{ belongs to } \hat{\mathcal{B}}_{\alpha-\frac{1}{2},\sigma_0}; \\ \nu_k \in C[0,1], \ k=1,\ldots,K_1\,, \ \mu_k \in C[0,1], \ k=1,\ldots,K_2; \\ \varphi \in C^1[0,1]. \end{cases}$$
 (3.2.1)

We note that without restriction of generality we can replace the condition $\sigma_0 > 0$ in (3.2.1) by $\sigma_0 \in \mathbb{R}$. Indeed, since $\hat{\mathcal{B}}_{\gamma,\sigma_1} \subset \hat{\mathcal{B}}_{\gamma,\sigma_2}$ for $\sigma_1 < \sigma_2$, the assumption (3.2.1) holds for any $\sigma_0 > 0$ provided it holds for some $\sigma_0 \leq 0$. However, the condition $\sigma_0 > 0$ simplifies the treatment of terms of the form $\frac{1}{\sigma^{\gamma}}$ for $\sigma \geq \sigma_0$ in forthcoming estimations (see e.g. (3.2.7)).

We start by proving some properties of the free term b of the equation (3.1.18).

Lemma 3.1. Let the assumptions (2.2.2), (3.2.1) hold. If $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma}$, then the vector function $b[Z] = (b^0[Z], b^1[Z])$, given by (3.1.21), (3.1.22), belongs to $\mathcal{B}_{\alpha,\sigma_0}$ and satisfies the estimate

$$||b[Z]||_{\alpha,\sigma} \le \operatorname{Const}\left[1 + \frac{1}{\sigma^{\frac{3}{2} - \alpha}} \left(|c| + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha - 1}}\right)\right]$$
 (3.2.2)

with any $\sigma \geq \sigma_0$ and some constant, where

$$|c| = \sum_{k=1}^{K} |c_k|. \tag{3.2.3}$$

Moreover, for every $\sigma \geq \sigma_0$ and $Z^1 = \frac{c}{p} + V^1$, $Z^2 = \frac{c}{p} + V^2 \in \mathcal{M}_{c,\alpha,\sigma}$ the difference $b[Z^1] - b[Z^2]$ fulfils the estimate

$$\left\|b[Z^1] - b[Z^2]\right\|_{\alpha,\sigma} \le \operatorname{Const} \frac{1}{\sqrt{\sigma}} \left\|V^1 - V^2\right\|_{\alpha,\sigma} \tag{3.2.4}$$

with some constant.

Proof. Let us start with the estimation of $b^0[Z]$. Substituting $\frac{c}{p} + V$ for Z in

(3.1.21), multiplying by $|p|^{\alpha}$ and estimating we have

$$|p|^{\alpha} |b^{0}[Z](x,p)| \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{|p|^{\alpha} |V_{k}(p)|}{|p|^{\alpha-1}} \right) |p| \int_{0}^{1} \left| G(x,y,p) \right| dy$$

$$\times \|\nu_{k}\|_{C[0,1]} \left[\frac{1}{|p|^{3-\alpha}} \max_{0 \leq y \leq 1} |c_{\Phi}(y)| + \frac{\|\varphi\|_{C[0,1]}}{|p|^{2-\alpha}} \right]$$

$$+ \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{|p|^{\alpha} |V_{k}(p)|}{|p|^{\alpha-1}} \right) \sqrt{|p|} \int_{0}^{1} |G_{y}(x,y,p)| dy \|\mu_{k-K_{1}}\|_{C[0,1]} \frac{\|\varphi'\|_{C[0,1]}}{|p|^{\frac{3}{2}-\alpha}}$$

$$+ |p|^{\alpha} \left| \widehat{\Phi}^{0}(x,p) \right|.$$

$$(3.2.5)$$

From (3.1.23) in view of the assumption (3.2.1), Lemmas 2.2 and 2.3 and the assertion (2.4.1) of Theorems 2.1 and 2.5 we get $\widehat{\Phi}^0 \in \mathcal{B}_{\alpha,\sigma_0}$ for the function $\widehat{\Phi}^0$. Using this relation, the assertions (2.4.1), (2.4.3) of Theorems 2.1, 2.5, the assumption (3.2.1) and the definitions of the norms $\|\cdot\|_{\gamma,\sigma}$ in $\mathcal{A}_{\alpha,\sigma}$ and $\widehat{\mathcal{B}}_{\alpha,\sigma}$ we obtain from (3.2.5)

$$|p|^{\alpha} |b^{0}[Z](x,p)| \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{\|V_{k}\|_{\alpha,\sigma}}{|p|^{\alpha-1}} \right) C_{1} \|\nu_{k}\|_{C[0,1]} \left[\frac{\|c_{\Phi}\|_{C[0,1]}}{|p|^{3-\alpha}} + \frac{\|\varphi\|_{C[0,1]}}{|p|^{2-\alpha}} \right]$$

$$+ \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{\|V_{k}\|_{\alpha,\sigma}}{|p|^{\alpha-1}} \right) C_{3} \|\mu_{k-K_{1}}\|_{C[0,1]} \frac{\|\varphi'\|_{C[0,1]}}{|p|^{\frac{3}{2}-\alpha}} + \|\widehat{\Phi}^{0}\|_{\alpha,\sigma_{0}}$$

for $\operatorname{Re} p > \sigma$, $\sigma \geq \sigma_0$, $x \in [0,1]$. Taking here the supremum over $\operatorname{Re} p > \sigma$, $x \in [0,1]$ and observing the relation $|p|^{\gamma} > \sigma^{\gamma}$ for $\operatorname{Re} p > \sigma$, which holds in the cases $\gamma = \alpha - 1$, $3 - \alpha$, $2 - \alpha$, $3/2 - \alpha$ due to (2.1.3), we have

$$||b^{0}[Z]||_{\alpha,\sigma} \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha-1}} \right) C_{1} ||\nu_{k}||_{C[0,1]} \left[\frac{||c_{\Phi}||_{C[0,1]}}{\sigma^{3-\alpha}} + \frac{||\varphi||_{C[0,1]}}{\sigma^{2-\alpha}} \right]$$

$$+ \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha-1}} \right) C_{3} ||\mu_{k-K_{1}}||_{C[0,1]} \frac{||\varphi'||_{C[0,1]}}{\sigma^{\frac{3}{2}-\alpha}} + ||\widehat{\Phi}^{0}||_{\alpha,\sigma_{0}}$$
(3.2.6)

for $\sigma \geq \sigma_0$. Finally, observing that

$$\sigma^{\gamma'} \ge \kappa \sigma^{\gamma}$$
 for $\sigma \ge \sigma_0$ in case $\gamma' \ge \gamma$ with $\kappa = \sigma_0^{\gamma' - \gamma} > 0$ (3.2.7)

because $\sigma_0 > 0$, we can take the minimal exponent of σ in the nominators of the estimate (3.2.6). Thus, in view of the definition of the norm in $(\mathcal{A}_{\alpha,\sigma})^K$ and (3.2.3), we arrive at the relation

$$\left\|b^{0}[Z]\right\|_{\alpha,\sigma} \le \frac{\operatorname{Const}}{\sigma^{\frac{3}{2}-\alpha}} \left(|c| + \frac{\|V\|_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) + \|\widehat{\Phi}^{0}\|_{\alpha,\sigma_{0}}, \quad \sigma \ge \sigma_{0}$$
 (3.2.8)

with some constant depending on the data of the problem.

Next, we perform similar transformations with $b^1[Z]$ in (3.1.22) multiplying by $|p|^{\alpha-\frac{1}{2}}$ instead of $|p|^{\alpha}$. We have

$$|p|^{\alpha - \frac{1}{2}} |b^{1}[Z](x,p)| \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{|p|^{\alpha} |V_{k}(p)|}{|p|^{\alpha - 1}} \right) \sqrt{|p|} \int_{0}^{1} |G_{x}(x,y,p)| \, dy$$

$$\times \|\nu_{k}\|_{C[0,1]} \left[\frac{1}{|p|^{3-\alpha}} \max_{0 \leq y \leq 1} |c_{\Phi}(y)| + \frac{\|\varphi\|_{C[0,1]}}{|p|^{2-\alpha}} \right]$$

$$+ \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{|p|^{\alpha} |V_{k}(p)|}{|p|^{\alpha - 1}} \right) \frac{1}{|p|^{\frac{3}{2} - \alpha}}$$

$$\times \left| \frac{\mu_{k-K_{1}}(x)\varphi'(x)}{\lambda(x)} - \int_{0}^{1} G_{xy}(x,y,p)\mu_{k-K_{1}}(y)\varphi'(y) \, dy \right| + |p|^{\alpha - \frac{1}{2}} \left| \widehat{\Phi}^{1}(x,p) \right|.$$

From (3.1.24) in view of the assumption (3.2.1), Lemmas 2.2 and 2.3 and the assertion (2.4.2) of Theorems 2.1, 2.5 we get $\widehat{\Phi}^1 \in \mathcal{B}_{\alpha-\frac{1}{2},\sigma_0}$ for the function $\widehat{\Phi}^1$. Thus, using the assertions (2.4.2), (2.4.4) of Theorems 2.1, 2.5 and taking the supremum over $\operatorname{Re} p > \sigma$, $x \in [0,1]$ we obtain

$$||b^{1}[Z]||_{\alpha-\frac{1}{2},\sigma} \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) C_{2}||\nu_{k}||_{C[0,1]} \left[\frac{||c_{\Phi}||_{C[0,1]}}{\sigma^{3-\alpha}} + \frac{||\varphi||_{C[0,1]}}{\sigma^{2-\alpha}}\right] + \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) \left(\frac{1}{\min\limits_{x\in[0,1]}\lambda(x)} + C_{4}\right) \frac{||\mu_{k-K_{1}}\varphi'||_{C[0,1]}}{\sigma^{\frac{3}{2}-\alpha}} + ||\widehat{\Phi}^{1}||_{\alpha-\frac{1}{2},\sigma_{0}}$$

for $\sigma \geq \sigma_0$. Since $\min_{x \in [0,1]} \lambda(x) > 0$ by assumption, this estimate due to (3.2.7) yields

$$||b^{1}[Z]||_{\alpha-\frac{1}{2},\sigma} \leq \frac{\text{Const}}{\sigma^{\frac{3}{2}-\alpha}} \left(|c| + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) + ||\widehat{\Phi}^{1}||_{\alpha-\frac{1}{2},\sigma_{0}}, \quad \sigma \geq \sigma_{0} \quad (3.2.9)$$

with a constant depending on the data.

In particular, observing the estimates (3.2.8) and (3.2.9) and applying Lemmas 2.2 and 2.3 for the components in (3.1.21), (3.1.22) we see that $b[Z]=(b^0[Z],b^1[Z])\in\mathcal{B}_{\alpha,\sigma}$ for $\sigma\geq\sigma_0$. Moreover, we get (3.2.2). To prove (3.2.4) we denote $Z=Z^1-Z^2$. Observe that in this case $b[Z]=b[Z^1]-b[Z^2]$, where the components $b^0[Z]$ and $b^1[Z]$ of the vector b[Z] are expressed by the formulas (3.1.21) with $\widehat{\Phi}^0=0$ and (3.1.22) with $\widehat{\Phi}^1=0$, respectively. Using the proved estimates (3.2.8) and (3.2.9) for the components of b[Z] and observing that Z has the form $Z=\frac{c}{p}+V$, where c=0 and $V=V^1-V^2$, we deduce (3.2.4). The proof is complete.

We continue proving properties of the operator A[Z] of the equation (3.1.18).

Lemma 3.2. Let the assumptions (2.2.2), (3.2.1) hold. If $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma}$, then the linear operator $A[Z] = (A^0[Z], A^1[Z])$, defined by (3.1.19), (3.1.20), is well-defined and bounded in $\mathcal{B}_{\alpha,\sigma}$ and satisfies the estimate

$$||A[Z]||_{\mathscr{L}(\mathcal{B}_{\alpha,\sigma})} \le C_0 \left[\frac{1+|c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}} \right]$$
(3.2.10)

for any $\sigma \geq \sigma_0$ with some constant C_0 . Moreover, taking $Z^1 = \frac{c}{p} + V^1$, $Z^2 = \frac{c}{n} + V^2 \in \mathcal{M}_{c,\alpha,\sigma}$, the estimate for difference

$$||A[Z^1] - A[Z^2]||_{\mathcal{L}(\mathcal{B}_{\alpha,\sigma})} \le \widehat{C}_0 \frac{1}{\sigma^{\alpha}} ||V^1 - V^2||_{\alpha,\sigma}$$
 (3.2.11)

holds for any $\sigma \geq \sigma_0$ with some constant \widehat{C}_0 .

Proof. First we show that the linear operator A[Z], is well-defined and bounded in $\mathcal{B}_{\alpha,\sigma}$. From (3.1.19) by $Z=\frac{c}{p}+V$ we get

$$|p|^{\alpha} \left| \left(A^{0}[Z]B \right)(x,p) \right| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|^{2}} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha+1}} + \frac{|n_{k}^{0}|}{|p|} \right)$$

$$\times |p| \int_{0}^{1} |G(x,y,p)| \, dy \cdot ||\nu_{k}||_{C[0,1]} \, |p|^{\alpha} \max_{0 \leq y \leq 1} \left| \widehat{B}^{0}(y,p) \right|$$

$$+ \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha+1}} \right) \sqrt{|p|} \int_{0}^{1} |G_{y}(x,y,p)| \, dy$$

$$\times ||\mu_{k-K_{1}}||_{C[0,1]} \, |p|^{\alpha-\frac{1}{2}} \max_{0 \leq y \leq 1} \left| B^{1}(y,p) \right| .$$

Using Theorems 2.1, 2.5 and taking the supremum over $\operatorname{Re} p > \sigma, \ x \in [0,1]$ we deduce

$$\begin{aligned} \|A^{0}[Z]B\|_{\alpha,\sigma} &\leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{\sigma^{2}} + \frac{\|V_{k}\|_{\alpha,\sigma}}{\sigma^{\alpha+1}} + \frac{|n_{k}^{0}|}{\sigma}\right) C_{1} \|\nu_{k}\|_{C[0,1]} \|\widehat{B}^{0}\|_{\alpha,\sigma} + \\ &+ \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{\sigma} + \frac{\|V_{k}\|_{\alpha,\sigma}}{\sigma^{\alpha}}\right) C_{3} \|\mu_{k-K_{1}}\|_{C[0,1]} \|B^{1}\|_{\alpha-\frac{1}{2},\sigma} \end{aligned}$$

for $\sigma \geq \sigma_0$. This, due to the relation (3.2.7), implies

$$||A^{0}[Z]B||_{\alpha,\sigma} \leq \operatorname{Const}\left[\frac{|n^{0}| + |c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}}\right] ||B||_{\alpha,\sigma}, \quad \sigma \geq \sigma_{0} \quad (3.2.12)$$

with some constant and

$$|n^0| = \sum_{k=1}^{K_1} |n_k^0|.$$

Further, from (3.1.20) we derive

$$|p|^{\alpha - \frac{1}{2}} \left| \left(A^{1}[Z]B \right) (x,p) \right| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|^{2}} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha + 1}} + \frac{|n_{k}^{0}|}{|p|} \right)$$

$$\times \sqrt{|p|} \int_{0}^{1} |G_{x}(x,y,p)| \, dy \cdot ||\nu_{k}||_{C[0,1]} |p|^{\alpha} \max_{0 \leq y \leq 1} \left| \widehat{B}^{0}(y,p) \right|$$

$$+ \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha}} \right) ||\mu_{k-K_{1}}||_{C[0,1]}$$

$$\times \left[\frac{1}{\min_{x \in [0,1]} \lambda(x)} + \int_{0}^{1} |G_{xy}(x,y,p)| \, dy \right] |p|^{\alpha - \frac{1}{2}} \max_{0 \leq y \leq 1} \left| B^{1}(y,p) \right| .$$

Using Theorems 2.1, 2.5 and taking the supremum over $\text{Re}p > \sigma$, $x \in [0,1]$ we get

$$||A^{1}[Z]B||_{\alpha-\frac{1}{2},\sigma} \leq \operatorname{Const}\left[\frac{|n^{0}|+|c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}}\right] ||B||_{\alpha,\sigma}, \quad \sigma \geq \sigma_{0} \quad (3.2.13)$$

with some constant.

Putting the estimates (3.2.12) and (3.2.13) together we have

$$||A[Z]B||_{\sigma} \leq \operatorname{Const}\left[\frac{|n^{0}| + |c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}}\right] ||B||_{\alpha,\sigma}, \quad \sigma \geq \sigma_{0}. \quad (3.2.14)$$

Using this relation and Lemmas 2.2, 2.3 for the components of (3.1.19), (3.1.20) we see that A[Z] is well-defined and bounded in $\mathcal{B}_{\alpha,\sigma}$. Moreover, (3.2.14) implies the estimate (3.2.10).

It remains to prove (3.2.11). Denoting $Z=Z^1-Z^2$ the components $A^0[Z]$ and $A^1[Z]$ of the vector $A[Z]=A[Z^1]-A[Z^2]$ are expressed by the formulas (3.1.19) and (3.1.20), respectively, containing $n^0=0$. Using the estimate (3.2.14) for A[Z] and observing that $Z=\frac{c}{p}+V$, where c=0 and $V=V^1-V^2$, we deduce (3.2.11). The lemma is proved.

Now we are in the situation to prove the main result concerning the equation (3.1.18).

Theorem 3.1. Let the assumptions (2.2.2), (3.2.1) hold. Then for any $\sigma \geq \sigma_0$ and $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma}$, satisfying the inequality

$$\eta(Z,\sigma) := \frac{1+|c|}{\sigma} + \frac{\|V\|_{\alpha,\sigma}}{\sigma^{\alpha}} \le \frac{1}{2C_0},$$
(3.2.15)

where C_0 is the constant from (3.2.10), equation (3.1.18) has a unique solution B[Z] in $\mathcal{B}_{\alpha,\sigma}$. This solution satisfies estimate

$$||B[Z]||_{\alpha,\sigma} \le \operatorname{Const}\left[1 + \frac{1}{\sigma^{\frac{3}{2} - \alpha}} \left(|c| + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha - 1}}\right)\right]$$
 (3.2.16)

with some constant. Moreover, for every $\sigma \geq \sigma_0$ and $Z^1 = \frac{c}{p} + V^1$, $Z^2 = \frac{c}{p} + V^2 \in \mathcal{M}_{c,\alpha,\sigma}$ such that $\eta(Z^j,\sigma) \leq \frac{1}{2C_0}$, j=1,2, the difference $B[Z^1] - B[Z^2]$ fulfils estimate

$$||B[Z^{1}] - B[Z^{2}]||_{\alpha,\sigma} \le \operatorname{Const} \left\{ \frac{1}{\sigma^{\alpha}} \left[1 + \frac{1}{\sigma^{\frac{3}{2} - \alpha}} \left(|c| + \frac{||V^{1}||_{\alpha,\sigma}}{\sigma^{\alpha - 1}} \right) \right] + \frac{1}{\sqrt{\sigma}} \right\} ||V^{1} - V^{2}||_{\alpha,\sigma} (3.2.17)$$

with some constant.

Proof. Due to Lemmas 3.1, 3.2 the equation (3.1.18) is well-defined in $\mathcal{B}_{\alpha,\sigma}$ if $\sigma \geq \sigma_0$ and $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma}$. Moreover, in case the relation (3.2.15) holds, from (3.2.10) we get $\|A[Z]\|_{\mathscr{L}(\mathcal{B}_{\alpha,\sigma})} \leq \frac{1}{2}$. Thus, by the contraction principle equation (3.1.18) has a unique solution $B = B[Z] \in \mathcal{B}_{\alpha,\sigma}$.

Furthermore, from the system (3.1.18) we have

$$||B[Z]||_{\alpha,\sigma} \le (1 - ||A[Z]||_{\mathscr{L}(\mathcal{B}_{\alpha,\sigma})})^{-1} ||b[Z]||_{\alpha,\sigma}.$$

This, in view of the inequality $||A[Z]||_{\mathscr{L}(\mathcal{B}_{\alpha,\sigma})} \leq \frac{1}{2}$ and (3.2.2), yields the estimate (3.2.16).

Finally, let us prove the estimate (3.2.17). Let $\sigma \geq \sigma_0$ and $Z^1 = \frac{c}{p} + V^1$, $Z^2 = \frac{c}{p} + V^2$ be such that (3.2.15) is valid for V replaced by V^1 and V^2 , i.e. $\eta(Z^j,\sigma) \leq \frac{1}{2C_0}$, j=1,2. Subtracting equation (3.1.18) for $Z=Z^2$ from the corresponding equation for $Z=Z^1$ we have

$$B[Z^{1}] - B[Z^{2}] = A[Z^{2}] (B[Z^{1}] - B[Z^{2}]) + (A[Z^{1}] - A[Z^{2}]) B[Z^{1}] + b[Z^{1}] - b[Z^{2}].$$

This implies

$$||B[Z^{1}] - B[Z^{2}]||_{\alpha,\sigma} \le (1 - ||A[Z^{2}]||_{\mathscr{L}(\mathcal{B}_{\alpha,\sigma})})^{-1} \times [||A[Z^{1}] - A[Z^{2}]||_{\mathscr{L}(\mathcal{B}_{\alpha,\sigma})} ||B[Z^{1}]||_{\alpha,\sigma} + ||b[Z^{1}] - b[Z^{2}]||_{\alpha,\sigma}].$$

Using in this relation the inequality $||A[Z^2]||_{\mathscr{L}(\mathcal{B}_{\alpha,\sigma})} \leq \frac{1}{2}$ and the estimates (3.2.4), (3.2.11), (3.2.16) we deduce (3.2.17). The proof is complete.

3.3 Existence and uniqueness for inverse problem

In this section we study the fixed-point equation (3.1.9) with (3.1.10) and thereupon infer results for the generalized inverse problem with temperature observations in the time domain.

Using the decomposition (3.1.17) for $B^0[Z]$ in the term containing n_k^0 in (3.1.10) we extract the Z-free addend $\frac{1}{p}\sum_{k=1}^{K_1}n_k^0\int_0^1pG(x_i,y,p)\nu_k(y)c_\Phi(y)dy$ from this term

and add it to $\widehat{\Psi}_i(p)$ to obtain the full Z-free term of the operator $\mathcal{F}=(\mathcal{F}_1,\ldots,\mathcal{F}_N)$ in the following form:

$$\Psi = (\Psi_1, \dots, \Psi_K), \quad \Psi_i(p) = \widehat{\Psi}_i(p) + \frac{1}{p} \sum_{k=1}^{K_1} n_k^0 \int_0^1 pG(x_i, y, p) \nu_k(y) c_{\Phi}(y) dy.$$
(3.3.18)

Recall that $\widehat{\Psi}_i$ is defined by (3.1.11).

Theorem 3.2. Assume that (2.2.2), (3.2.1) hold and

$$\nu_k \in C^1[0,1], k = 1, \dots, K_1, \quad \mu_k \in C^2[0,1], k = 1, \dots, K_2, \quad \varphi \in C^3[0,1].$$
(3.3.19)

Moreover, let $\det \Gamma \neq 0$ for Γ , given by (3.1.7), and

$$\Psi = \frac{d}{p} + Y \in \mathcal{M}_{d,\alpha,\sigma_0} \tag{3.3.20}$$

with some $d \in \mathbb{R}^K$. Then there exists $\sigma_1 \geq \sigma_0$ such that equation (3.1.9) with (3.1.10) has a solution $Z = \frac{c}{p} + V$ in the space $\mathcal{M}_{c,\alpha,\sigma_1}$, where $c = \Gamma^{-1}d$. The solution is unique in the union of spaces $\bigcup_{\substack{\sigma \geq \sigma_1 \\ c \in \mathbb{P}}} \mathcal{M}_{c,\alpha,\sigma}$.

Proof. Firstly, we prove the existence assertion of theorem. Let us set $c = \Gamma^{-1}d$. Observing (3.1.10), (3.3.18), (3.3.20) we see that equation (3.1.9) for $Z = \frac{c}{p} + V$ in $\mathcal{M}_{c,\alpha,\sigma}$ is equivalent to the following equation for V in $(\mathcal{A}_{\alpha,\sigma})^K$:

$$V = F(V), \tag{3.3.21}$$

where $F = \Gamma^{-1}F_1$.

$$F_1(V) = L_0\left(\frac{c}{p} + V, B[Z]\right) + L_1\left(\frac{c}{p} + V\right) + L_2(B[Z]) + Y, \tag{3.3.22}$$

 L_0 is the following bilinear operator of $Z \in \mathcal{M}_{c,\alpha,\sigma}$ and $B = (\widehat{B}^0, B^1) \in \mathcal{B}_{\alpha,\sigma}$:

$$(L_0(Z,B))_i(p) = \sum_{k=1}^{K_1} Z_k(p) \int_0^1 pG(x_i,y,p) \nu_k(y) \widehat{B}^0(y,p) dy$$
$$-\sum_{k=K_1+1}^K Z_k(p) \int_0^1 pG_y(x_i,y,p) \mu_{k-K_1}(y) B^1(y,p) dy, \ i = 1,\dots,K, \ (3.3.23)$$

and L_1 and L_2 are the linear operators of $Z \in \mathcal{M}_{c,\alpha,\sigma}$ and $B = (\widehat{B}^0, B^1) \in \mathcal{B}_{\alpha,\sigma}$,

given by the formulae

$$(L_{1}(Z))_{i}(p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \left\{ \int_{0}^{1} G(x_{i}, y, p) \nu_{k}(y) c_{\Phi}(y) dy + \int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \varphi(y) dy + \frac{\nu_{k}(x_{i}) \varphi(x_{i})}{\beta(x_{i})} \right\}$$

$$+ \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left\{ \int_{0}^{1} pG(x_{i}, y, p) \left(\mu_{k-K_{1}}(y) \varphi'(y) \right)' dy + \frac{\left(\mu_{k-K_{1}}(x) \varphi'(x) \right)'}{\beta(x_{i})} \right|_{x=x_{i}}$$

$$+ \mu_{k-K_{1}}(0) \varphi'(0) pG(x_{i}, 0, p) - \mu_{k-K_{1}}(1) \varphi'(1) pG(x_{i}, 1, p) \right\},$$

$$i = 1, \dots, K,$$

$$(3.3.24)$$

and

$$(L_2(B))_i(p) = \sum_{k=1}^{K_1} n_k^0 \int_0^1 pG(x_i, y, p) \nu_k(y) \widehat{B}^0(y, p) dy, \ i = 1, \dots, K, \ (3.3.25)$$

respectively.

We will make use of the fixed-point argument for the equation (3.3.21) the following balls:

$$D_{\alpha,\sigma}(\rho) = \left\{ V \in (\mathcal{A}_{\alpha,\sigma})^K : \|V\|_{\alpha,\sigma} \le \rho \right\}. \tag{3.3.26}$$

To this end we first deduce some estimates for L_0, L_1 and L_2 . Multiplying by $|p|^{\alpha}$ in (3.3.23) and estimating we have

$$|p|^{\alpha} \left| \left(L_{0} \left(\frac{c}{p} + V, B \right) \right)_{i}(p) \right| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha}} \right) |p| \int_{0}^{1} |G(x_{i}, y, p)| \, dy$$

$$\times \|\nu_{k}(y)\|_{C[0,1]} |p|^{\alpha} \max_{0 \leq y \leq 1} \left| \widehat{B}^{0}(y, p) \right|$$

$$+ \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha-1}} \right) \sqrt{|p|} \int_{0}^{1} |G_{y}(x_{i}, y, p)| \, dy$$

$$\times \|\mu_{k-K_{1}}\|_{C[0,1]} |p|^{\alpha-\frac{1}{2}} \max_{0 \leq y \leq 1} \left| B^{1}(y, p) \right|, \ i = 1, \dots, K.$$

Using assertions (2.4.1), (2.4.3) of Theorems 2.1, 2.5 and the definitions of the

norms $\|\cdot\|_{\gamma,\sigma}$ in $\mathcal{A}_{\gamma,\sigma}$ and $\hat{\mathcal{B}}_{\gamma,\sigma}$ we obtain

$$|p|^{\alpha} \left| \left(L_0 \left(\frac{c}{p} + V, B \right) \right)_i(p) \right| \leq \sum_{k=1}^{K_1} \left(\frac{|c_k|}{|p|} + \frac{\|V_k\|_{\alpha,\sigma}}{|p|^{\alpha}} \right) C_1 \|\nu_k(y)\|_{C[0,1]} \|\widehat{B}^0\|_{\alpha,\sigma} + \sum_{k=K_1+1}^K \left(|c_k| + \frac{\|V_k\|_{\alpha,\sigma}}{|p|^{\alpha-1}} \right) C_3 \|\mu_{k-K_1}\|_{C[0,1]} \|B^1\|_{\alpha-\frac{1}{2},\sigma}, \ i = 1, \dots, K,$$

for $\operatorname{Re} p > \sigma$, $\sigma \geq \sigma_0$. Taking in this relation the supremum over $\operatorname{Re} p > \sigma$ and observing the definition of the norm $\|\cdot\|_{\alpha,\sigma}$ in $(\mathcal{A}_{\gamma,\sigma})^K$ as well as (3.2.3) we have

$$\left\| L_0 \left(\frac{c}{p} + V, B \right) \right\|_{\alpha, \sigma} \le \operatorname{Const} \left[\left(\frac{|c|}{\sigma} + \frac{\|V\|_{\alpha, \sigma}}{\sigma^{\alpha}} \right) \|\widehat{B}^0\|_{\alpha, \sigma} + \left(|c| + \frac{\|V\|_{\alpha, \sigma}}{\sigma^{\alpha - 1}} \right) \|B^1\|_{\alpha - \frac{1}{2}, \sigma} \right], \ \sigma \ge \sigma_0$$

with some constant. Using here the inequality (3.2.7) with $\gamma=0, \gamma'=1$ and $\gamma=\alpha-1, \gamma'=\alpha$ and observing the definition of the norm $\|\cdot\|_{\alpha,\sigma}$ in $\mathcal{B}_{\gamma,\sigma}$ we deduce the estimate

$$\left\| L_0 \left(\frac{c}{p} + V, B \right) \right\|_{\alpha, \sigma} \le \operatorname{Const} \left(|c| + \frac{\|V\|_{\alpha, \sigma}}{\sigma^{\alpha - 1}} \right) \|B\|_{\alpha, \sigma}, \quad \sigma \ge \sigma_0$$
 (3.3.27)

with some constant.

Similarly, multiplying in (3.3.24) by $|p|^{\alpha}$, observing the norm $\|\cdot\|_{\alpha,\sigma}$ in $\mathcal{A}_{\alpha,\sigma}$ and estimating we have

$$\begin{split} |p|^{\alpha} \Big| \Big(L_{1} \Big(\frac{c}{p} + V \Big) \Big)_{i}(p) \Big| \\ &\leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) \left(\frac{1}{\sqrt{|p|}} \|\nu_{k}\|_{C[0,1]} |p| \int_{0}^{1} |G(x_{i}, y, p)| dy \\ &\times \max_{0 \leq y \leq 1} |c_{\Phi}(y)| + \sqrt{|p|} \left| \int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \varphi(y) \, dy + \frac{\nu_{k}(x_{i}) \varphi(x_{i})}{\beta(x_{i})} \right| \right) \\ &+ \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) \left(\sqrt{|p|} \left| \int_{0}^{1} pG(x_{i}, y, p) \left(\mu_{k-K_{1}}(y) \varphi'(y) \right)' dy \right. \\ &\left. + \frac{1}{\beta(x_{i})} \left(\mu_{k-K_{1}}(x) \varphi'(x) \right)' \right|_{x=x_{i}} \right| \\ &+ |\mu_{k-K_{1}}(0)||\varphi'(0)| \, |p|^{3/2} |G(x_{i}, 0, p)| + |\mu_{k-K_{1}}(1)||\varphi'(1)| \, |p|^{3/2} |G(x_{i}, 1, p)| \right), \\ i &= 1, \dots K, \end{split}$$

for $\operatorname{Re} p > \sigma$, $\sigma \geq \sigma_0$. In the case of boundary conditions of the first kind the terms $G(x_i,0,p)$ and $G(x_i,1,p)$ in this estimate are equal to zero (see (1.2.12)). In the

case of boundary conditions of the third kind we make use of the assertion (2.5.1) of Theorem 2.5 with $\kappa=3/2$ and take the assumption $x_i\in(0,1)$ into account. Then we get

$$|p|^{3/2}|G(x_i,0,p)| \le C_{3/2}^1(x_i), \quad |p|^{3/2}|G(x_i,1,p)| \le C_{3/2}^1(x_i), \quad \operatorname{Re} p > 0$$

Using these relations, the assertion (2.4.1) of Theorems 2.1, 2.5 and the assertions (2.4.6) and (2.5.6) of Theorems 2.3 and 2.7, respectively, we deduce

$$|p|^{\alpha} \Big| \Big(L_{1} \Big(\frac{c}{p} + V \Big) \Big)_{i}(p) \Big| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) \\ \times \left(\frac{C_{1}}{\sqrt{|p|}} \|\nu_{k}\|_{C[0,1]} \|c_{\Phi}\|_{C[0,1]} + \widetilde{C}_{i} \|\nu_{k}\varphi\|_{C^{1}[0,1]} \right) \\ + \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) \Big[\widetilde{C}_{i} \| \left(\mu_{k-K_{1}} \varphi' \right)' \|_{C^{1}[0,1]} \\ + \theta \Big(|\mu_{k-K_{1}}(0)\varphi'(0)|C_{3/2}^{1}(x_{i}) + |\mu_{k-K_{1}}(1)\varphi'(1)|C_{3/2}^{1}(x_{i}) \Big) \Big], \ i = 1, \dots K,$$

for $\operatorname{Re} p > \sigma$, $\sigma \geq \sigma_0$ where $\widetilde{C}_i = C_6(x_i)$, $\theta = 0$ in the case of boundary conditions of the first kind and $\widetilde{C}_i = C_9$, $\theta = 1$ in the case of boundary conditions of the third kind. Taking here the supremum over $\operatorname{Re} p > \sigma$ and observing (3.2.7) with $\gamma = 0, \gamma' = 1/2$ we obtain

$$\left\| L_1 \left(\frac{c}{p} + V \right) \right\|_{\alpha, \sigma} \le \operatorname{Const} \left(\frac{|c|}{\sigma^{\frac{3}{2} - \alpha}} + \frac{\|V\|_{\alpha, \sigma}}{\sqrt{\sigma}} \right), \quad \sigma \ge \sigma_0 \tag{3.3.28}$$

with some constant

Finally, from (3.3.25) in view of the assertion (2.2.32) of Theorems 2.1, 2.5 we deduce

$$||L_2(B)||_{\alpha,\sigma} \le \operatorname{Const} ||B||_{\alpha,\sigma}, \ \sigma \ge \sigma_0$$
 (3.3.29)

with some constant.

Let us return to the equation (3.3.21) with the operator $F = \Gamma^{-1}F_1$, where F_1 is given by (3.3.22). By means of (3.3.27), (3.3.28), (3.3.29) and the relation $\|Y\|_{\alpha,\sigma} \leq \|Y\|_{\alpha,\sigma_0}$ we obtain

$$||F_1(V)||_{\alpha,\sigma} \le \operatorname{Const}\left(1 + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) \left[||B[Z]||_{\sigma} + \frac{1}{\sigma^{\frac{3}{2}-\alpha}}\right] + ||Y||_{\alpha,\sigma_0}, \ \sigma > \sigma_0$$
(3.3.30)

with some constant depending on |c|. Let us suppose that $V \in D_{\alpha,\sigma}(\rho)$. Then, using Lemmas 2.2 and 2.3 to the components of (3.3.23) - (3.3.25) and observing (3.3.22) and (3.3.30) we see that $F_1(V) \in (\mathcal{A}_{\alpha,\sigma})^K$ for $\sigma \geq \sigma_0$. If in addition, σ , ρ satisfy the relation

$$\eta_0(\rho, \sigma) := \frac{1 + |c|}{\sigma} + \frac{\rho}{\sigma^{\alpha}} \le \frac{1}{2C_0},$$
(3.3.31)

then (3.2.15) holds for σ and $Z=\frac{c}{p}+V$. Hence, we can apply estimate (3.2.16) of Theorem 3.1 for $\|B[Z]\|_{\alpha,\sigma}$. Plugging (3.2.16) into (3.3.30) and estimating $\|V\|_{\alpha,\sigma}$ by ρ and observing that $\frac{1}{\sigma^{\frac{3}{2}-\alpha}} \leq \operatorname{Const}$ for $\sigma \geq \sigma_0$ we have

$$||F_1(V)||_{\alpha,\sigma} \le \widehat{C}\left(1 + \frac{\rho}{\sigma^{\alpha-1}}\right)\left(1 + \frac{\rho}{\sqrt{\sigma}}\right) + ||Y||_{\alpha,\sigma_0}$$
(3.3.32)

with some constant \widehat{C} independent of ρ and σ . From (3.3.31) and (3.3.32), due to the equality $F = \Gamma^{-1}F_1$ and the relation $\alpha > 1$, we see that for every $\rho > \rho_0 :=$ $|\Gamma^{-1}|(\widehat{C} + ||Y||_{\alpha,\sigma_0})$ there exists $\sigma_2 = \sigma_2(\rho) \geq \sigma_0$ such that the inequalities $\eta_0(\rho,\sigma) \leq \frac{1}{2C_0}$ and $\|FV\|_{\alpha,\sigma} \leq \rho$ hold for any $\sigma \geq \sigma_2(\rho)$. Consequently, by the definition of $D_{\alpha,\sigma}(\rho)$ we have

$$F: D_{\alpha,\sigma}(\rho) \to D_{\alpha,\sigma}(\rho) \quad \text{for} \quad \rho > \rho_0 \quad \text{and} \quad \sigma \ge \sigma_2(\rho).$$
 (3.3.33)

Next, we prove that F is a contraction in $D_{\alpha,\sigma}(\rho)$ with suitable parameters ρ and σ . From (3.3.22), due to the bilinearity of L_0 and the linearity of L_1 , L_2 , we

$$F_1(V) - F_1(\tilde{V}) = L_0(V - \tilde{V}, B[Z]) + L_0(\frac{c}{p} + \tilde{V}, B[Z] - B[\tilde{Z}]) + L_1(V - \tilde{V}) + L_2(B[Z] - B[\tilde{Z}]),$$

where $Z=\frac{c}{p}+V,\ \widetilde{Z}=\frac{c}{p}+\widetilde{V}.$ Using here (3.3.27), (3.3.28) and (3.3.29) we get

$$||F_{1}(V) - F_{1}(\widetilde{V})||_{\alpha,\sigma} \leq \operatorname{Const}\left\{\left(\frac{1}{\sqrt{\sigma}} + \frac{||B[Z]||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) ||V - \widetilde{V}||_{\alpha,\sigma}\right.$$
$$\left. + \left(1 + \frac{||\widetilde{V}||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) ||B[Z] - B[\widetilde{Z}]||_{\alpha,\sigma}\right\}, \ \sigma \geq \sigma_{0}$$

with some constant depending on |c|. Let us suppose that $V, \widetilde{V} \in D_{\alpha,\sigma}(\rho)$, where $\sigma \geq \sigma_0$ and ρ , σ satisfy (3.3.31). Then we can make use of the estimates (3.2.16) and (3.2.17) of Theorem 3.1 to get

$$||F_1(V) - F_1(\widetilde{V})||_{\alpha,\sigma} \leq \operatorname{Const}\left\{\frac{1}{\sqrt{\sigma}} + \frac{1}{\sigma^{\alpha-1}}\left(1 + \frac{1}{\sigma^{\frac{3}{2}-\alpha}}\left(1 + \frac{\rho}{\sigma^{\alpha-1}}\right)\right) + \left(1 + \frac{\rho}{\sigma^{\alpha-1}}\right)\left[\frac{1}{\sigma^{\alpha}}\left(1 + \frac{1}{\sigma^{\frac{3}{2}-\alpha}}\left(1 + \frac{\rho}{\sigma^{\alpha-1}}\right)\right) + \frac{1}{\sqrt{\sigma}}\right]\right\}||V - \widetilde{V}||_{\alpha,\sigma}.$$

In view of the assumed inequalities $1 < \alpha < \frac{3}{2}$ (cf. (3.2.1), (2.1.3)), the coefficient of $||V - V||_{\alpha,\sigma}$ on the right-hand side of this estimate approaches zero, as $\sigma \to \infty$ for any fixed $\rho > 0$. Hence, for any $\rho > 0$ there exists $\sigma_3 = \sigma_3(\rho) \geq \sigma_0$ such that the inequality $\eta_0(\rho,\sigma) \leq \frac{1}{2C_3}$ holds and $F = \Gamma^{-1}F_1$ is a contraction in the ball $D_{\alpha,\sigma}(\rho)$ for $\rho > 0$ and $\sigma \ge \sigma_3(\rho)$. This together with (3.3.33) shows that the equation (3.3.21) has a unique solution V in every ball $D_{\alpha,\sigma}(\rho)$, where $\rho > \rho_0$ and $\sigma \ge \sigma_4(\rho) = \max\{\sigma_2(\rho); \sigma_3(\rho)\}$. This proves the existence assertion of the theorem with $\sigma_1 = \sigma_4(2\rho_0)$.

It remains to prove the uniqueness assertion of the theorem. We start it proving the uniqueness in the following union: $\bigcup_{\sigma \geq \sigma_1} \mathcal{M}_{c,\alpha,\sigma}$ where $c = \Gamma^{-1}d$. Suppose that (3.1.9) has two solutions $Z^1 = \frac{c}{p} + V^1$ and $Z^2 = \frac{c}{p} + V^2$ in this union. Observing the relation

$$\mathcal{M}_{c,\alpha,\sigma} \subset \mathcal{M}_{c,\alpha,\widetilde{\sigma}}$$
 for any $0 \le \sigma \le \widetilde{\sigma}$,

following from the definition of $\mathcal{M}_{c,\alpha,\sigma}$, we see that there exists $\widehat{\sigma} \geq \sigma_1$ such that $Z^1, Z^2 \in \mathcal{M}_{c,\alpha,\widehat{\sigma}}$. This implies that the components V^1 and V^2 of these solutions belong to $(\mathcal{A}_{\alpha,\widehat{\sigma}})^N$. Moreover, V^1 and V^2 solve the equation (3.3.21). Let us define

$$\bar{\rho} := \max(2\rho_0; \|V^1\|_{\alpha,\widehat{\sigma}}; \|V^2\|_{\alpha,\widehat{\sigma}}) \quad \text{and} \quad \bar{\sigma} := \max(\widehat{\sigma}; \sigma_4(\bar{\rho})). \quad (3.3.34)$$

From the left equality in this formula we have $\|V^j\|_{\alpha,\widehat{\sigma}} \leq \bar{\rho}, j=1,2$. Since the norm $\|\cdot\|_{\alpha,\sigma}$ is non-increasing with respect to σ and $\bar{\sigma} \geq \widehat{\sigma}$, we derive $\|V^j\|_{\alpha,\bar{\sigma}} \leq \bar{\rho}, j=1,2$. Thus, by the definition of $D_{\alpha,\sigma}(\rho)$ we get

$$V^j \in D_{\alpha,\bar{\sigma}}(\bar{\rho}), \quad j=1,2.$$

But the uniqueness of the solution of the equation (3.3.21) in the ball $D_{\alpha,\bar{\sigma}}(\bar{\rho})$ has already been shown. This is so, because by (3.3.34) the inequalities $\bar{\rho} > \rho_0$ and $\bar{\sigma} \geq \sigma_4(\bar{\rho})$ are valid. Consequently, $V^1 = V^2$ implying $Z^1 = Z^2$.

To complete the proof of the uniqueness assertion we have to show that (3.1.9) has no solution in any space $\mathcal{M}_{\widetilde{c},\alpha,\sigma}$ where $\widetilde{c} \neq c = \Gamma^{-1}d$ and $\sigma \geq \sigma_1$. Suppose contrary that (3.1.9) has a solution $Z = \frac{\widetilde{c}}{p} + V$ in some of such spaces. Then, observing (3.1.9), (3.1.10), (3.3.18), (3.3.20) we see that component $V \in (\mathcal{A}_{\alpha,\sigma})^K$ of this solution satisfies the equation

$$\Gamma V = L_0 \left(\frac{c}{p} + V, B[Z] \right) + L_1 \left(\frac{c}{p} + V \right) + L_2 (B[Z]) + V + \frac{\widetilde{c} - c}{p}, \quad (3.3.35)$$

where and L_0 , L_1 , L_2 are given by (3.3.23) - (3.3.25). Observing the definitions of $(\mathcal{A}_{\alpha,\sigma})^K$, $\mathcal{M}_{c,\alpha,\sigma}$, $\|\cdot\|_{\alpha,\sigma}$, the relation $V\in(\mathcal{A}_{\alpha,\sigma})^K$, the assumption (3.3.20) and the estimates (3.3.27), (3.3.28), (3.3.29) we see that all terms in (3.3.35) except for $\frac{\widetilde{c}-c}{p}$ are estimated by quantities of the form $\frac{\mathrm{Const}}{|p|^{\alpha}}$ in the half-plane $\mathrm{Re}\,p>\sigma_0$, where the constant is independent of p and a>1. This is not possible in the case $\widetilde{c}-c\neq 0$. We reached the contradiction. The uniqueness is completely proved. \square

Finally, we deduce the following corollary for the generalized inverse problem in the time domain.

Corollary 3.1. Let conditions (3.1.4) hold with $\mathcal{R} = K_1$ implying the existence of the unique solution n_k^0 , $k = 1, \ldots, K_1$, to the system (3.1.3). Moreover, let (2.2.2) hold and the assumptions (3.2.1), (3.3.19), (3.3.20) be satisfied for the functions λ_k , μ_k , φ and the quantities Φ^0 , Φ^1 , Ψ given in terms of the Laplace transforms R, F_1 , F_2 , H_i of the data r, f_1 , f_2 , h_i by formulas (3.1.13), (3.1.15), (3.3.18), (3.1.11) and either (1.2.14) or (1.2.23) depending on the type of the boundary conditions. Assume that $\det \Gamma \neq 0$ for Γ , given by (3.1.7).

Then the generalized inverse problem with temperature observations has a unique solution of the form

$$n_{k}(t) = n_{k}^{0} + c_{k}t + \frac{1}{2\pi i} \int_{0}^{t} \int_{\xi-i\infty}^{\xi+i\infty} e^{\tau p} Z_{k}(p) \, dp \, d\tau \,, \quad k = 1, \dots K_{1} \,,$$

$$m_{k}(t) = c_{k+K_{1}} + \frac{1}{2\pi i} \int_{\xi-i\infty}^{\xi+i\infty} e^{tp} Z_{k+K_{1}}(p) \, dp \,, \quad k = 1, \dots K_{2} \,,$$

$$(3.3.36)$$

where
$$c = (c_1, ..., c_K) \in \mathbb{R}^K$$
, $Z = (Z_1, ..., Z_K) \in (\mathcal{A}_{\alpha, \sigma})^K$, $\sigma > 0$.

The functions n_k are continuously differentiable and m_k are continuous for $t \geq 0$. Moreover, the vector c in the formulas (3.3.36) is expressed by $c = \Gamma^{-1}d$, where d is the component of Ψ in the assumption (3.3.20). In addition, $n_k(0) = n_k^0$, $n_k'(0) = c_k$, $k = 1, \ldots, K_1$ and $m_k(0) = c_{k+K_1}$, $k = 1, \ldots, K_2$.

Proof. Corollary follows from Theorem 3.2 and Proposition 3.1 together with the inversion formula (1.2.4) of the Laplace transform, the properties 1, 3, 4 of the Laplace transform in Section 1.2.1, the definitions of the spaces $(\mathcal{A}_{\alpha,\sigma})^K$, $\mathcal{M}_{c,\alpha,\sigma}$ and the formulas $\mathcal{L}_{t\to p}(c) = \frac{c}{p}$, $\mathcal{L}_{t\to p}\left(\int_0^t w(\tau)d\tau\right) = \frac{1}{p}\mathcal{L}_{t\to p}w(t)$.

3.4 Interpretation of assumptions

This section consists of two parts. In the first part we interpret the regularity assumptions of the existence theorem in the time domain. In the second part we analyse the non-vanishing condition $\det \Gamma \neq 0$.

3.4.1 Interpretation of regularity assumptions

The assumptions of Theorem 3.2 and Corollary 3.1 contain assumptions on the Laplace transforms of the data of the inverse problem. Let us give sufficient conditions in the time domain implying these assumptions.

We recall that the condition $\sigma_0 > 0$ in (3.2.1) can be replaced by $\sigma_0 \in \mathbb{R}$ (see the remark after (3.2.1)). Thus, let us start with the following condition in (3.2.1):

$$\Phi^0$$
 given by (3.1.13) satisfies (3.1.16) where $c_{\Phi} \in C[0,1]$ and $\widetilde{\Phi}^0 \in \hat{\mathcal{B}}_{\alpha,\sigma}$ with some α satisfying (2.1.3).

We split
$$\Phi^0 = \Phi^{0,1} + \Phi^{0,2}$$
 with $\Phi^{0,1}(x,p) = \sum_{k=1}^{K_1} n_k^0 \int_0^1 G(x,y,p) \nu_k(y) \varphi(y) dy$ and $\Phi^{0,2}(x,p) = -pQ(x,p) - \varphi(x)$. In case $\nu_k, \varphi \in C^1[0,1]$ and (2.2.2) hold

Theorems 2.3 and 2.7 immediately imply

$$\Phi^{0,1}(x,p) = -\frac{1}{p} \sum_{k=1}^{K_1} n_k^0 \frac{\nu_k(x)\varphi(x)}{\beta(x)} + \widetilde{\Phi}^{0,1}(x,p)$$

with $\widetilde{\Phi}^{0,1} \in \hat{\mathcal{B}}_{3/2,0}$. The latter relation yields $\widetilde{\Phi}^{0,1} \in \hat{\mathcal{B}}_{\alpha,\sigma}$ for any α satisfying (2.1.3) and $\sigma > 0$. To study $\Phi^{0,2}$, let us denote by $u^0(x,t)$ the solution of the direct problem in the case n=m=0, i.e.

$$\beta(x)u_t^0(x,t) = \frac{\partial}{\partial x} (\lambda(x)u_x^0(x,t)) + r(x,t), \quad x \in (0,1), t > 0,$$

$$u^0(x,0) = \varphi(x), \quad x \in (0,1),$$
 (3.4.38)

 u^0 satisfies either (1.1.6) or (1.1.7) with m=0.

Assume

$$u^{0}(x,\cdot), u^{0}_{t}(x,\cdot), u^{0}_{x}(x,\cdot), u^{0}_{xx}(x,\cdot) \in \mathscr{E} \text{ for any } x \in (0,1)$$
 (3.4.39)

and denote $U^0(x,p)=\mathcal{L}_{t\to p}u^0(x,t)$. Then U^0 satisfies the equation (1.2.15) with $M_k=N_k=0$. From this equation we see that $U^0=-Q$. Thus, by virtue of the property 1 of the Laplace transform in Section 1.2.1 and the initial value $u^0(x,0)=\varphi(x)$, we have $\Phi^{0,2}(x,p)=\mathcal{L}_{t\to p}u^0_t(x,t)$. Applying again the property 1 for $\mathcal{L}_{t\to p}u^0_{tt}(x,t)$ we get $\Phi^{0,2}(x,p)=\frac{u^0_t(x,0)}{p}+\frac{1}{p}\mathcal{L}_{t\to p}u^0_{tt}(x,t)$. Consequently, by the properties 1 and 5 of the Laplace transform the assumptions

$$u^0_{tt}(x,\cdot)$$
 and $\frac{d^{\alpha-1}}{dt^{\alpha-1}}u^0_{tt}(x,\cdot)$ with some α satisfying (2.1.3) belong to $\mathscr E$ for $x\in[0,1]$, where the parameters C , σ of the space $\mathscr E$ (3.4.40) are independent of x

yield the relation

$$\Phi^{0,2}(x,p) = \frac{u_t^0(x,0)}{p} + \widetilde{\Phi}^{0,2}(x,p)$$

with $\widetilde{\Phi}^{0,2} \in \hat{\mathcal{B}}_{\alpha,\sigma}$. Summing up, (3.4.39), (3.4.40) with $\nu_k, \varphi \in C^1[0,1], u_t^0(x,0) \in C[0,1]$ and (2.2.2) imply (3.4.37) with

$$c_{\Phi}(x) = -\sum_{k=1}^{K_1} n_k^0 \frac{\nu_k(x)\varphi(x)}{\beta(x)} + u_t^0(x,0).$$

Secondly, let us consider the following assumption in (3.2.1):

 Φ^1 given by (3.1.15) belongs to $\hat{\mathcal{B}}_{\alpha-\frac{1}{2},\sigma}$ with α satisfying (2.1.3). (3.4.41)

Comparing (3.1.15) with (3.1.13) we see that $\Phi^1 = \Phi_x^0$. Thus, $\Phi^1 = \Phi_x^{0,1} + \Phi_x^{0,2}$. In case $\nu_k, \varphi \in C^2[0,1]$ Theorems 2.4, 2.7 imply the property (3.4.41) for Φ_x^0 .

Since $\Phi_x^{0,2}(x,p) = \mathcal{L}_{t\to p} u_{xt}^0(x,t)$, due to property 5 of the Laplace transform in §1.2.1, the conditions

$$u^0_{xt}(x,\cdot)$$
 and $\frac{d^{\alpha-1/2}}{dt^{\alpha-1/2}}u^0_{xt}(x,\cdot)$ with some α satisfying (2.1.3) belong to $\mathscr E$ for $x\in[0,1]$, where the parameters C , σ of the space $\mathscr E$ (3.4.42) are independent of x

yield the property (3.4.41) for $\Phi_x^{0,2}$. Summing up, (3.4.39), (3.4.42) with $\nu_k, \varphi \in C^2[0,1]$ imply (3.4.41).

Further, we deduce a formula in the time domain for the quantity

$$\lim_{\text{Re }p\to+\infty}p^2[H_i(p)+Q(x_i,p)]$$

in the rank condition (3.1.4). In view of property 1 of the Laplace transform and the relations $U^0 = -Q$, $u^0(x,0) = \varphi(x)$, the conditions (3.4.39), (3.4.40) with

$$h_i^{(j)} \in \mathcal{E}, \quad j = 0, 1, 2, \quad i = 1, \dots, K,$$
 (3.4.43)

imply the formula

$$p^{2}[H_{i}(p) + Q(x_{i}, p)] = \mathcal{L}_{t \to p} \left[h_{i}''(t) - u_{tt}^{0}(x_{i}, t) \right] + p \left[h_{i}(0) - \varphi(x_{i}) \right] + h_{i}'(0) - u_{t}^{0}(x_{i}, 0).$$
(3.4.44)

Assuming the consistency conditions

$$h_i(0) = \varphi(x_i), \quad i = 1, \dots, K,$$
 (3.4.45)

and observing that the Laplace transform vanishes as $\operatorname{Re} p \to +\infty$ (see (1.2.3)) we obtain the formula

$$\lim_{\text{Re }p\to+\infty} p^2[H_i(p) + Q(x_i, p)] = h_i'(0) - u_t^0(x_i, 0), \quad i = 1, \dots, K. \quad (3.4.46)$$

Finally, we interpret the assumption (3.3.20) for the vector $\Psi=(\Psi_1,\ldots,\Psi_K)$ given by (3.3.18) with (3.1.11). Let us decompose $\Psi=\Psi^1+\Psi^2+\Psi^3$ with $\Psi^j=(\Psi^j_1,\ldots,\Psi^j_K)$, where

$$\begin{split} &\Psi_{i}^{1}(p) = \sum_{k=1}^{K_{1}} n_{k}^{0} \left[\int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \varphi(y) dy + \frac{\nu_{k}(x_{i})\varphi(x_{i})}{\beta(x_{i})} \right], \\ &\Psi_{i}^{2}(p) = \sum_{k=1}^{K_{1}} n_{k}^{0} \int_{0}^{1} pG(x_{i}, y, p) \nu_{k}(y) \widetilde{B}^{0}(y, p) dy, \\ &\Psi_{i}^{3}(p) = -p^{2} [H_{i}(p) + Q(x_{i}, p)] + \lim_{R \neq x \to \infty} q^{2} [H_{i}(q) + Q(x_{i}, q)] \end{split}$$

and study the components Ψ^1, Ψ^2, Ψ^3 separately. Let us start with Ψ^1 . Note that Theorems 2.3 and 2.7 provide estimate of the order $|p|^{-1/2}$ for $\Psi^1_i(p)$ at infinity,

which is not enough for our purposes. To get necessary higher estimate we have to assume more regularity about the functions β , ν_k , φ inside Ψ^1 . We introduce these regularity assumptions in an implicit form in terms of the solution of a certain direct problem. Namely, let w(x,t) be the solution of the following problem:

$$\beta(x)w_{t}(x,t) = \frac{\partial}{\partial x} (\lambda(x)w_{x}(x,t)), \quad x \in (0,1), \ t > 0,$$

$$w(x,0) = -\sum_{k=1}^{K_{1}} n_{k}^{0} \frac{\nu_{k}(x)\varphi(x)}{\beta(x)}, \quad x \in (0,1),$$
(3.4.47)

w satisfies the homogeneous first or third kind boundary conditions.

Assuming

$$w(x,\cdot), w_t(x,\cdot), w_x(x,\cdot), w_{xx}(x,\cdot) \in \mathscr{E} \text{ for any } x \in (0,1)$$
 (3.4.48)

and denoting $W(x,p) = \mathcal{L}_{t \to p} w(x,t)$ the transformed problem reads

$$\frac{\partial}{\partial x} \left(\lambda(x) W_x(x, p) \right) - p\beta(x) W(x, p) = \sum_{k=1}^{K_1} n_k^0 \nu_k(x) \varphi(x), \quad x \in (0, 1),$$

W satisfies the homogeneous first or third kind boundary conditions.

Making use of the Green function we can represent the solution of this problem in the form $W(x,p)=\sum\limits_{k=1}^{K_1}n_k^0\int_0^1G(x,y,p)\nu_k(y)\varphi(y)dy$. Thus, by the property 1 of the Laplace transform, we get

$$\mathcal{L}_{t \to p} w_t(x, t) = pW(x, p) - w(x, 0)$$

$$= \sum_{k=1}^{K_1} n_k^0 \left[\int_0^1 pG(x, y, p) \nu_k(y) \varphi(y) dy + \frac{\nu_k(x) \varphi(x)}{\beta(x)} \right].$$

Consequently,

$$\Psi_i^1(p) = \mathcal{L}_{t \to p} w_t(x_i, t).$$

To get the condition (3.3.20) for Ψ^1 , we have to assume sufficient regularity of the solution of (3.4.47) in the interior points $x_i \in (0,1)$. More precisely, due to the property 5, is sufficient to assume

$$w_{tt}(x_i, \cdot)$$
 and $\frac{d^{\alpha-1}}{dt^{\alpha-1}}w_{tt}(x_i, \cdot)$ with some α satisfying (2.1.3) belong to \mathscr{E} for $i = 1, \dots, K$.

The condition (3.3.20) for Ψ^2 immediately follows from (3.4.37) with the help of assertions of Theorems 2.3 and 2.7. Here the relation (3.4.37) is the consequence of the assumptions (3.4.39), (3.4.40), $\nu_k, \varphi \in C^1[0,1], u_t^0(x,0) \in C[0,1]$ and (2.2.2), as was shown above. To study Ψ^3 we note that under the assumptions (3.4.39), (3.4.40), (3.4.43) and (3.4.45) the formulas (3.4.44) and (3.4.46) yield

$$\Psi_i^3(p) = -\mathcal{L}_{t\to p} \left[h_i''(t) - u_{tt}^0(x_i, t) \right].$$

By virtue of the properties 1 and 5 of the Laplace transform, the sufficient conditions for Ψ^2 to satisfy the condition (3.3.20), are (3.4.40), (3.4.43), (3.4.45) and

$$h_i''' - u_{ttt}^0(x_i, \cdot)$$
 and $\frac{d^{\alpha-1}}{dt^{\alpha-1}}[h_i''' - u_{ttt}^0(x_i, \cdot)]$ with some α satisfying (2.1.3) belong to \mathscr{E} for $i = 1, \dots, K$.

Summing up, sufficient conditions for the assumption (3.3.20) for Ψ are $\nu_k, \varphi \in C^1[0,1]$, (2.2.2), (3.4.39), (3.4.40), $u_t^0(x,0) \in C[0,1]$, (3.4.43), (3.4.45), (3.4.48), (3.4.49) and (3.4.50).

We gave the conditions in the time domain in the form of the smoothness of the solutions u^0 and w of the problems (3.4.38) and (3.4.47). Using known results about the regularity of solutions of parabolic problems (see e.g. [28]) it is possible to write these conditions in terms of the data r, φ , f_1 , f_2 , ν_k and β of these problems, as well.

3.4.2 Interpretation of non-vanishing condition

Let us analyse the non-vanishing condition $\det \Gamma \neq 0$ for the matrix Γ given by (3.1.7) in some particular cases. Firstly, we consider the case of piecewise homogeneous rod where $K_1 = K_2$, $K = 2K_1$ and the functions ν_k and μ_k are given by (1.1.9) with $0 < y_1 < y_2 < \ldots < y_{K_1-1} < 1$ and small $\epsilon > 0$. Let us choose $0 < x_1 < x_2 < \ldots < x_K < 1$ so that

$$x_1, x_2 \in (0, y_1 - \epsilon), x_3, x_4 \in (y_1 + \epsilon, y_2 - \epsilon), \dots,$$

 $x_{K-1}, x_K \in (y_{K_1+1} + \epsilon, 1).$ (3.4.51)

Then $\nu_k(x_{2k-1}) = \mu_k(x_{2k-1}) = \nu_k(x_{2k}) = \mu_k(x_{2k}) = 1$, $\nu_k(x_i) = \mu_k(x_i) = 0$ for $i \notin \{2k-1; 2k\}$ and $\mu'_k(x_i) = 0$ for any $i = 1, \dots, K$. From (3.1.7) we get

Consequently, $\det \Gamma \neq 0$ provided the initial value φ of u satisfies the contition

$$\det \left(\begin{array}{cc} \varphi(x_{2l-1}) & \varphi''(x_{2l-1}) \\ \varphi(x_{2l}) & \varphi''(x_{2l}) \end{array} \right) \neq 0 \quad \text{for any } l = 1, \dots, K_1.$$

It is not difficult to define φ so that it satisfies this condition.

Secondly, let us consider the power basis $\nu_k(x) = \mu_k(x) = x^{k-1}$. Then

$$\det \Gamma = \prod_{l=1}^{K} \frac{1}{\beta(x_l)}$$

$$\times \det \left(\left(x_i^k \varphi(x_i) \right)_{k=0,\dots,K_1-1} \left(x_i^k \varphi''(x_i) + k x_i^{k-1} \varphi'(x_i) \right)_{k=0,\dots,K_2-1} \right)_{i=1,\dots,K}$$

Here the matrix is formed by placing two matrices left to right. It is possible to choose the initial value φ and points x_i so that the condition $\det \Gamma \neq 0$ is valid. For instance, in case

$$\varphi(x) = \cos\left(\pi K x^{\frac{K_1+2}{2}}\right), \quad x_i = \left(\frac{i}{K}\right)^{\frac{2}{K_1+2}}$$

we can represent $\det \Gamma$ as a constant times the Vandermonde determinant:

$$\det \Gamma = \prod_{l=1}^{K} \frac{1}{\beta(x_l)} (-1)^{\left\lceil \frac{K}{2} \right\rceil} \left[-\left(\frac{\pi K(K_1 + 2)}{2} \right)^2 \right]^{K_2} \det \left(x_i^k \right)_{k=0,\dots,K-1 \atop i=1,\dots,K} \neq 0$$

Here $\lceil x \rceil$ is the smallest integer $\geq x$.

4. Inverse problem with flux observations

In this chapter we consider the generalized inverse problem with flux observations.

4.1 Reduction of the inverse problem to a fixed-point form

Inverse problem in the Laplace domain consist of the system (1.2.26) with additional equations (1.2.15), (1.2.18).

Using the relation (1.2.18) for U_x in the left-hand side of the conditions (1.2.26) we obtain

$$\sum_{k=1}^{K_1} N_k(p) \int_0^1 pG_x(x_i, y, p) \nu_k(y) U(y, p) dy$$

$$-\sum_{k=1}^{K_2} M_k(p) \int_0^1 G_{xy}(x_i, y, p) \mu_k(y) U_y(y, p) dy = Q_x(x_i, p) - \frac{H_i(p)}{\lambda(x_i)},$$
(4.1.1)

for
$$i = 1, ..., N$$
.

The system (4.1.1) with (1.2.15), (1.2.18) is equivalent to the system (1.2.26) with (1.2.15), (1.2.18). We are going to transform the system (4.1.1) into a fixed-point form extracting the principal part. As in Section 3.1 we use some proper asymptotics of the system in the process $\operatorname{Re} p \to +\infty$. Such an asymptotics is achieved under the conditions that the system (4.1.1) with (1.2.15), (1.2.18) has a solution $N_k \Big|_{k=1,\dots,K_1}$, $M_k \Big|_{k=1,\dots,K_2}$, U satisfying the following properties

- (1) $n_k(t) = \mathcal{L}_{p \to t}^{-1} N_k(p)$ and $m_k(t) = \mathcal{L}_{p \to t}^{-1} M_k(p)$ belong to $\mathscr E$ implying $N_k(p) \to 0$, $M_k(p) \to 0$ as $\operatorname{Re} p \to +\infty$ (cf. property 1 in Section 1.2.1);
- (2) for $u(x,t)=\mathcal{L}_{p\to t}^{-1}U(x,p)$ the relations $u(x,\cdot),u_t(x,\cdot),u_x(x,\cdot),u_{xt}(x,\cdot)\in\mathscr{E}$ hold with the initial condition (1.1.5) implying (3.1.2).

We emphasize that (1) and (2) are not assumptions. We use them to explain the ideas of transformations.

Let us introduce the new unknowns

$$Z = (Z_1, \dots, Z_K), \ Z_k(p) = \begin{cases} N_k(p), & k = 1, \dots, K_1, \\ M_{k-K_1}(p), & k = K_1 + 1, \dots, K. \end{cases}$$
(4.1.2)

Now the system (4.1.1) can be rewritten in the form

$$\sum_{k=1}^{K_{1}} Z_{k}(p) \left(\frac{\nu_{k}(x)\varphi(x)}{\beta(x)} \right)' \Big|_{x=x_{i}} + \sum_{k=K_{1}+1}^{K} Z_{k}(p) \frac{\mu_{k-K_{1}}(x_{i})\varphi(x_{i})}{\lambda(x_{i})} \right) \\
= \sum_{k=1}^{K_{1}} Z_{k}(p) \left\{ \int_{0}^{1} pG_{x}(x_{i}, y, p)\nu_{k}(y) \left[pU(y, p) - \varphi(y) \right] dy \right. (4.1.3) \\
+ \int_{0}^{1} pG_{x}(x_{i}, y, p)\nu_{k}(y)\varphi(y) dy + \left(\frac{\nu_{k}(x)\varphi(x)}{\beta(x)} \right)' \Big|_{x=x_{i}} \right\} \\
- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left\{ \int_{0}^{1} G_{xy}(x_{i}, y, p)\mu_{k-K_{1}}(y) \left[pU_{y}(y, p) - \varphi'(y) \right] dy \right. \\
+ \int_{0}^{1} G_{xy}(x_{i}, y, p)\mu_{k-K_{1}}(y)\varphi'(y) dy - \frac{\mu_{k-K_{1}}(x_{i})\varphi'(x_{i})}{\lambda(x_{i})} \right\} \\
- p \left[Q_{x}(x_{i}, p) - \frac{H_{i}(p)}{\lambda(x_{i})} \right], \qquad i = 1, \dots, K.$$

In view of (1), (2) and Theorems 2.2, 2.4, 2.6, 2.8 the left-hand side of (4.1.3) is the principal part of this system in the process $\text{Re}p \to +\infty$. Therefore, we introduce the matrix

$$\Gamma = (\gamma_{ik})_{i,k=1,\dots,K}, \, \gamma_{ik} = \begin{cases} \left. \left(\frac{\nu_k(x)\varphi(x)}{\beta(x)} \right)' \right|_{x=x_i}, & k = 1,\dots,K_1, \\ \frac{\mu_{k-K_1}(x_i)\varphi'(x_i)}{\lambda(x_i)}, & k = K_1 + 1,\dots,K, \end{cases}$$
(4.1.4)

related to the principal part and assume $\det \Gamma \neq 0$.

Next we define the functions

$$B^{0}[Z](x,p) = pU[Z](x,p) - \varphi(x),$$

$$B^{1}[Z](x,p) = pU_{x}[Z](x,p) - \varphi'(x),$$
(4.1.5)

where U[Z](x,p) is the solution of (1.2.15), (1.2.18) with the given vector Z in the form (4.1.2). The system (4.1.3) can be rewritten now in the fixed-point form

$$Z = \Gamma^{-1} \mathcal{F}(Z), \tag{4.1.6}$$

where
$$\mathcal{F}(Z) = (\mathcal{F}_1(Z), \dots, \mathcal{F}_K(Z)),$$

$$\mathcal{F}_{i}[Z](p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \left\{ \int_{0}^{1} pG_{x}(x_{i}, y, p)\nu_{k}(y)B^{0}[Z](y, p) dy + \int_{0}^{1} pG_{x}(x_{i}, y, p)\nu_{k}(y)\varphi(y) dy + \left(\frac{\nu_{k}(x)\varphi(x)}{\beta(x)}\right)'\Big|_{x=x_{i}} \right\} + \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left\{ -\int_{0}^{1} G_{xy}(x_{i}, y, p)\mu_{k-K_{1}}(y)B^{1}[Z](y, p) dy - (4.1.7) - \int_{0}^{1} G_{xy}(x_{i}, y, p)\mu_{k-K_{1}}(y)\varphi'(y) dy + \frac{\mu_{k-K_{1}}(x_{i})\varphi'(x_{i})}{\lambda(x_{i})} \right\} + \Psi_{i}(p),$$

and

$$\Psi_i(p) = -p \left[Q_x(x_i, p) - \frac{H_i(p)}{\lambda(x_i)} \right], \quad i = 1, \dots, K.$$
 (4.1.8)

We need to deduce a fixed-point system for the quantities $B^0[Z]$ and $B^1[Z]$ too. To this end we rewrite the system (1.2.15), (1.2.18) in terms of these quantities:

$$B^{0}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} pG(x,y,p)\nu_{k}(y) \left[B^{0}[Z](y,p) + \varphi(y)\right] dy$$

$$- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \int_{0}^{1} G_{y}(x,y,p)\mu_{k-K_{1}}(y) \left[B^{1}[Z](y,p) + \varphi'(y)\right] dy + \Phi^{0}(x,p)$$

$$B^{1}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} pG_{x}(x,y,p)\nu_{k}(y) \left[B^{0}[Z](y,p) + \varphi(y)\right] dy$$

$$+ \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left\{\frac{\mu_{k-K_{1}}(x)}{\lambda(x)} B^{1}[Z](y,p) - \int_{0}^{1} G_{xy}(x,y,p)\mu_{k-K_{1}}(y) \right\}$$

$$\times \left[B^{1}[Z](y,p) + \varphi'(y)\right] dy + \sum_{k=K_{1}+1}^{K} Z_{k}(p) \frac{\mu_{k-K_{1}}(x)\varphi'(x)}{\lambda(x)} + \Phi^{1}(x,p)$$

with

$$\Phi^{0}(x,p) = -pQ(x,p) - \varphi(x),
\Phi^{1}(x,p) = -pQ_{x}(x,p) - \varphi'(x) = \Phi^{0}_{x}(x,p).$$
(4.1.9)

We obtain the following fixed-point equation for vector $B[Z] = (B^0[Z], B^1[Z])$:

$$B[Z] = A[Z]B[Z] + b[Z],$$
 (4.1.10)

where $A[Z] = (A^0[Z], A^1[Z])$ is the Z-dependent linear operator of B with the components

$$(A^{0}[Z]B)(x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} pG(x,y,p)\nu_{k}(y)B^{0}(y,p) dy \qquad (4.1.11)$$

$$- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \int_{0}^{1} G_{y}(x,y,p)\mu_{k-K_{1}}(y)B^{1}(y,p) dy,$$

$$(A^{1}[Z]B)(x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} pG_{x}(x,y,p)\nu_{k}(y)B^{0}(y,p) dy$$

$$+ \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left\{ \frac{\mu_{k-K_{1}}(x)}{\lambda(x)} B^{1}(x,p) - \int_{0}^{1} G_{xy}(x,y,p)\mu_{k-K_{1}}(y)B^{1}(y,p) dy \right\}$$

$$- \int_{0}^{1} G_{xy}(x,y,p)\mu_{k-K_{1}}(y)B^{1}(y,p) dy$$

and $b[Z] = (b^0[Z], b^1[Z])$ is the Z-dependent B-free term with the components

$$b^{0}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} pG(x,y,p)\nu_{k}(y)\varphi(y) dy$$

$$- \sum_{k=K_{1}+1}^{K} Z_{k}(p) \int_{0}^{1} G_{y}(x,y,p)\mu_{k-K_{1}}(y)\varphi'(y) dy + \Phi^{0}(x,p),$$

$$b^{1}[Z](x,p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} pG_{x}(x,y,p)\nu_{k}(y)\varphi(y) dy$$

$$+ \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left\{ \frac{\mu_{k-K_{1}}(x)\varphi'(x)}{\lambda(x)} - \int_{0}^{1} G_{xy}(x,y,p)\mu_{k-K_{1}}(y)\varphi'(y) dy \right\} + \Phi^{1}(x,p),$$

$$(4.1.14)$$

with $\Phi^0(x,p)$ and $\Phi^1(x,p)$ from (4.1.9). Summing up, we have proved

Proposition 4.1. Let $\det \Gamma \neq 0$. Then the inverse problem in the Laplace domain with flux observations is equivalent to equation (4.1.6) with \mathcal{F} given by (4.1.7). The solutions of these two equivalent problems are related by (4.1.2) with (4.1.5). Moreover, the pair $B[Z] = (B^0[Z], B^1[Z])$ in (4.1.5) satisfies the system (4.1.10) with (4.1.11) - (4.1.14).

4.2 Analysis of direct problem system

In this section we study the system (4.1.10). Let us introduce the following basic assumptions:

$$\Phi^{0} \text{ and } \Phi^{1} \text{ given by (4.1.9) belong to } \hat{\mathcal{B}}_{1,\sigma_{0}} \text{ and } \hat{\mathcal{B}}_{\frac{1}{2},\sigma_{0}},$$
respectively, with some $\sigma_{0} > 0$;
$$\nu_{k} \in C[0,1], \ k = 1, \ldots, K_{1}, \ \mu_{l} \in C^{1}[0,1], \ l = 1, \ldots, K_{2};$$

$$\varphi \in C^{2}[0,1].$$
(4.2.1)

As in Section 3.2 we note that without a restriction of generality the condition $\sigma_0 > 0$ in (4.2.1) can be replaced by $\sigma_0 \in \mathbb{R}$.

We prove first some estimates for vectors $b[Z] = (b^0[Z], b^1[Z])$ and $A[Z] = (A^0[Z], A^1[Z])$.

Lemma 4.1. Let the assumptions (2.2.2), (4.2.1) hold. If $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma}$, then the vector function $b[Z] = (b^0[Z], b^1[Z])$, given by (4.1.13), (4.1.14), belongs to \mathcal{B}_{1,σ_0} and satisfies the estimate

$$||b[Z]||_{1,\sigma} \le \operatorname{Const}\left(1 + |c| + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right)$$
(4.2.2)

with any $\sigma \geq \sigma_0$ and some constant, where |c| is given by (3.2.3). Moreover, for every $\sigma \geq \sigma_0$ and $Z^1 = \frac{c}{p} + V^1$, $Z^2 = \frac{c}{p} + V^2 \in \mathcal{M}_{c,\alpha,\sigma}$ the difference $b[Z^1] - b[Z^2]$ fulfils the estimate

$$||b[Z^1] - b[Z^2]||_{1,\sigma} \le \text{Const} \frac{1}{\sigma^{\alpha - 1}} ||V^1 - V^2||_{\alpha,\sigma}$$
 (4.2.3)

with some constant.

Proof. Taking $Z = \frac{c}{p} + V$ in (4.1.13) and multiplying by |p| we have

$$|p| |b^{0}[Z](x,p)| \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha-1}} \right) |p| \int_{0}^{1} |G(x,y,p)| dy$$

$$\times \|\nu_{k}\varphi\|_{C[0,1]} + \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha-1}} \right) \sqrt{|p|} \int_{0}^{1} |G_{y}(x,y,p)| dy$$

$$\times \frac{\|\mu_{k-K_{1}}\varphi'\|_{C[0,1]}}{\sqrt{|p|}} + |p| |\Phi^{0}(x,p)|.$$

Using here the assertions of Theorems 2.1, 2.5 and definitions of the norms

 $\|\cdot\|_{\gamma,\sigma}$ and taking the supremum over $\operatorname{Re} p > \sigma, \ x \in [0,1]$ we obtain

$$\begin{aligned} & \left\| b^{0}[Z] \right\|_{1,\sigma} \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{\|V_{k}\|_{\alpha,\sigma}}{\sigma^{\alpha-1}} \right) C_{1} \|\nu_{k}\varphi\|_{C[0,1]} \\ & + \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{\|V_{k}\|_{\alpha,\sigma}}{\sigma^{\alpha-1}} \right) C_{3} \frac{\|\mu_{k-K_{1}}\varphi'\|_{C[0,1]}}{\sqrt{\sigma}} + \left\| \Phi^{0} \right\|_{1,\sigma_{0}} \end{aligned}$$

for Re $p>\sigma,\,\sigma\geq\sigma_0,\,x\in[0,1].$ From this relation in view of (3.2.7) with $\gamma'=1/2,\,\gamma=0$ we get finally

$$\|b^{0}[Z]\|_{1,\sigma} \le \operatorname{Const}\left(|c| + \frac{\|V\|_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) + \|\Phi^{0}\|_{1,\sigma_{0}}, \quad \sigma \ge \sigma_{0} \quad (4.2.4)$$

with some constant.

Next, we perform similar transformations with $b^1[Z]$ in (4.1.14) multiplying by $\sqrt{|p|}$ instead of |p|. We have

$$\sqrt{|p|} |b^{1}[Z](x,p)| \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{|p|^{\alpha} |V_{k}(p)|}{|p|^{\alpha-1}} \right) \sqrt{|p|} \int_{0}^{1} |G_{x}(x,y,p)| dy
\times \|\nu_{k}\varphi\|_{C[0,1]} + \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{|p|^{\alpha} |V_{k}(p)|}{|p|^{\alpha-1}} \right) \frac{1}{\sqrt{|p|}}
\times \left\{ \left| \frac{\mu_{k-K_{1}}(x)\varphi'(x)}{\lambda(x)} \right| + \int_{0}^{1} |G_{xy}(x,y,p)| dy \|\mu_{k-K_{1}}\varphi'\|_{C[0,1]} \right\}
+ \sqrt{|p|} |\Phi^{1}(x,p)|.$$

Using here the assumption (4.2.1), the assertions of Theorems 2.1, 2.5 and taking the supremum over $\text{Re}p > \sigma$, $x \in [0, 1]$ we get

$$||b^{1}[Z]||_{\frac{1}{2},\sigma} \leq \sum_{k=1}^{K_{1}} \left(|c_{k}| + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) C_{2} ||\nu_{k}\varphi||_{C[0,1]} + \sum_{k=K_{1}+1}^{K} \left(|c_{k}| + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) \left(\frac{1}{\lambda_{0}} + C_{4}\right) \frac{||\mu_{k-K_{1}}\varphi'||_{C[0,1]}}{\sqrt{\sigma}} + ||\Phi^{1}||_{\frac{1}{2},\sigma_{0}}$$

for $\sigma \geq \sigma_0$, where $\lambda_0 = \min_{x \in [0,1]} \lambda(x)$. Consequently,

$$||b^{1}[Z]||_{\frac{1}{2},\sigma} \le \operatorname{Const}\left(|c| + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right) + ||\Phi^{1}||_{\frac{1}{2},\sigma_{0}}, \quad \sigma \ge \sigma_{0}$$
 (4.2.5)

with some constant.

Observing inequalities (4.2.4), (4.2.5) and using Lemmas 2.2, 2.3 for the terms in (4.1.13), (4.1.14) we see that $b[Z]=(b^0[Z],b^1[Z])\in\mathcal{B}_{1,\sigma}$ for $\sigma\geq\sigma_0$. Moreover, (4.2.4), (4.2.5) imply (4.2.2). To prove (4.2.3) take $Z=Z^1-Z^2$. Then the components $b^0[Z]$ and $b^1[Z]$ of the vector $b[Z]=b[Z^1]-b[Z^2]$ are expressed by the formulas (4.1.13) and (4.1.14) with $\Phi^0=0$ and $\Phi^1=0$, respectively. Using the estimates (4.2.4) and (4.2.5) for the components of b[Z] and observing that $Z^1-Z^2=V^1-V^2$ we deduce (4.2.3). The lemma is proved.

Lemma 4.2. Let the assumptions (2.2.2), (4.2.1) hold. If $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma}$, then the linear operator $A[Z] = (A^0[Z], A^1[Z])$, defined by (4.1.11), (4.1.12), is well-defined and bounded in $\mathcal{B}_{1,\sigma}$ and satisfies the estimate

$$||A[Z]||_{\mathscr{L}(\mathcal{B}_{1,\sigma})} \le \widetilde{C_0} \left(\frac{|c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}} \right)$$
(4.2.6)

for any $\sigma \geq \sigma_0$ with a constant $\widetilde{C_0}$. Moreover, taking $Z^1 = \frac{c}{p} + V^1$, $Z^2 = \frac{c}{p} + V^2 \in \mathcal{M}_{c,\alpha,\sigma}$, the estimate for difference

$$||A[Z^1] - A[Z^2]||_{\mathcal{L}(\mathcal{B}_{1,\sigma})} \le \overline{C_0} \frac{1}{\sigma^{\alpha}} ||V^1 - V^2||_{\alpha,\sigma}$$
 (4.2.7)

holds for any $\sigma \geq \sigma_0$ with a constant $\overline{C_0}$.

Proof. Taking $Z = \frac{c}{p} + V$ in (4.1.11) we get

$$|p| \left| \left(A^{0}[Z]B \right)(x,p) \right| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha}} \right) |p| \int_{0}^{1} |G(x,y,p)| \, dy$$

$$\times \|\nu_{k}\|_{C[0,1]} |p| \max_{0 \leq x \leq 1} \left| B^{0}(x,p) \right| + \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha}} \right)$$

$$\times \sqrt{|p|} \int_{0}^{1} |G_{y}(x,y,p)| \, dy \, \|\mu_{k-K_{1}}\|_{C[0,1]} \sqrt{|p|} \max_{0 \leq x \leq 1} \left| B^{1}(x,p) \right| \, .$$

Using Theorems 2.1, 2.5 and taking the supremum over $\text{Re}p > \sigma$, $x \in [0,1]$ we deduce

$$||A^{0}[Z]B||_{1,\sigma} \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{\sigma} + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha}}\right) C_{1} ||\nu_{k}||_{C[0,1]} ||B^{0}||_{1,\sigma} + \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{\sigma} + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha}}\right) C_{3} ||\mu_{k-K_{1}}||_{C[0,1]} ||B^{1}||_{\frac{1}{2},\sigma}$$

for $\sigma \geq \sigma_0$. This implies

$$||A^{0}[Z]B||_{1,\sigma} \leq \operatorname{Const}\left(\frac{|c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}}\right)||B||_{1,\sigma}, \quad \sigma \geq \sigma_{0}$$
 (4.2.8)

with some constant.

Further, from (4.1.12) we get

$$\sqrt{|p|} \left| \left(A^{1}[Z]B \right)(x,p) \right| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha}} \right) \sqrt{|p|} \int_{0}^{1} |G_{x}(x,y,p)| \, dy
\times \|\nu_{k}\|_{C[0,1]} \left| p \right| \max_{0 \leq x \leq 1} \left| B^{0}(x,p) \right| + \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|} + \frac{|p|^{\alpha}|V_{k}(p)|}{|p|^{\alpha}} \right)
\times \left(\frac{1}{\lambda_{0}} + \int_{0}^{1} |G_{xy}(x,y,p)| \, dy \right) \|\mu_{k-K_{1}}\|_{C[0,1]} \sqrt{|p|} \max_{0 \leq x \leq 1} |B^{1}(x,p)|.$$

Using Theorems 2.1, 2.5 and taking the supremum over $\operatorname{Re} p > \sigma, \ x \in [0,1]$ we have

$$||A^{1}[Z]B||_{\frac{1}{2},\sigma} \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{\sigma} + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha}}\right) C_{2} ||\nu_{k}||_{C[0,1]} ||B^{0}||_{1,\sigma} + \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{\sigma} + \frac{||V_{k}||_{\alpha,\sigma}}{\sigma^{\alpha}}\right) ||\mu_{k-K_{1}}||_{C[0,1]} \left(\frac{1}{\lambda_{0}} + C_{4}\right) ||B^{1}||_{\frac{1}{2},\sigma}$$

for $\sigma \geq \sigma_0$. Finally, we obtain

$$||A^{1}[Z]B||_{\frac{1}{2},\sigma} \leq \operatorname{Const}\left(\frac{|c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}}\right)||B||_{1,\sigma}, \quad \sigma \geq \sigma_{0}$$
 (4.2.9)

with some constant.

Putting together (4.2.8) and (4.2.9) we have

$$||A[Z]B||_{1,\sigma} \le \operatorname{Const}\left(\frac{|c|}{\sigma} + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha}}\right) ||B||_{1,\sigma}, \quad \sigma \ge \sigma_0.$$
 (4.2.10)

Observing this relation and using Lemmas 2.2, 2.3 for the terms in (4.1.11), (4.1.12) we see that A[Z] is well-defined and bounded in \mathcal{B}_{σ} . Moreover, (4.2.10) yields (4.2.6). Denoting $Z=Z^1-Z^2$ the components $A^0[Z]$ and $A^1[Z]$ of the vector $A[Z]=A[Z^1]-A[Z^2]$ are expressed by the formulas (4.1.11) and (4.1.12) respectively. Using the estimate (4.2.10) for A[Z] and observing that $Z=\frac{c}{p}+V$ with c=0 and $V=V^1-V^2$ we deduce (4.2.7). The lemma is proved.

Due to Lemmas 4.1, 4.2 and the contraction principle, the equation (4.1.10) has a unique solution $B=B[Z]\in\mathcal{B}_{1,\sigma}$ provided $Z=\frac{c}{p}+V\in\mathcal{M}_{c,\sigma}$ and $\sigma\geq\sigma_0$ satisfy the relation

$$\widetilde{\eta}(Z,\sigma) := \frac{|c|}{\sigma} + \frac{\|V\|_{\alpha,\sigma}}{\sigma^{\alpha}} \le \frac{1}{2\widetilde{C_0}}.$$
 (4.2.11)

From (4.1.10) we have $||B[Z]||_{\sigma} \leq (1 - ||A[Z]||_{\mathscr{L}(\mathcal{B}_{1,\sigma})})^{-1} ||b[Z]||_{\sigma}$. This, taking into consideration (4.2.2), (4.2.6) and (4.2.11), gives the estimate for the solution of (4.1.10)

$$||B[Z]||_{1,\sigma} \le \operatorname{Const}\left[1 + |c| + \frac{||V||_{\alpha,\sigma}}{\sigma^{\alpha-1}}\right]$$
(4.2.12)

with some constant.

Next we will find an estimate for $B[Z^1]-B[Z^2]$. Let $\sigma \geq \sigma_0$ and $Z^1=\frac{c}{p}+V^1$, $Z^2=\frac{c}{p}+V^2$ be such that (4.2.11) is valid for V replaced by V^1 and V^2 i.e. $\widetilde{\eta}(Z^j,\sigma)\leq \frac{1}{2\widetilde{C}_0},\ j=1,2.$ Subtracting equation (4.1.10) for $Z=Z^2$ from the corresponding equation for $Z=Z^1$ we get

$$\begin{array}{rcl} B[Z^1] - B[Z^2] & = & A[Z^2] \left(B[Z^1] - B[Z^2] \right) + \left(A[Z^1] - A[Z^2] \right) B[Z^1] \\ & + & b[Z^1] - b[Z^2]. \end{array}$$

This implies

$$||B[Z^{1}] - B[Z^{2}]||_{1,\sigma} \leq (1 - ||A[Z^{2}]||_{\mathscr{L}(\mathcal{B}_{1,\sigma})})^{-1} \times \left[||A[Z^{1}] - A[Z^{2}]||_{\mathscr{L}(\mathcal{B}_{1,\sigma})} ||B[Z^{1}]||_{1,\sigma} + ||b[Z^{1}] - b[Z^{2}]||_{1,\sigma} \right].$$

Using this relation and the estimates (4.2.3), (4.2.6), (4.2.7), (4.2.11), (4.2.12) we obtain

$$||B[Z^{1}] - B[Z^{2}]||_{1,\sigma}$$

$$\leq \operatorname{Const} \left\{ \frac{1}{\sigma^{\alpha}} \left[1 + |c| + \frac{||V^{1}||_{\alpha,\sigma}}{\sigma^{\alpha-1}} \right] + \frac{1}{\sigma^{\alpha-1}} \right\} ||V^{1} - V^{2}||_{\alpha,\sigma}$$
(4.2.13)

with some constant. Summing up, we have proved the following theorem.

Theorem 4.1. Let the assumptions (2.2.2), (4.2.1) hold. Then for any $\sigma \geq \sigma_0$ and $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma}$, satisfying the inequality (4.2.11), where \widetilde{C}_0 is the constant from (4.2.6), equation (4.1.10) has a unique solution $B[Z] = \left(B^0[Z], B^1[Z]\right)$ in $\mathcal{B}_{1,\sigma}$. This solution satisfies estimate (4.2.12). Moreover, for every $\sigma \geq \sigma_0$ and $Z^1 = \frac{c}{p} + V^1$, $Z^2 = \frac{c}{p} + V^2 \in \mathcal{M}_{c,\alpha,\sigma}$ such that $\widetilde{\eta}(Z^j,\sigma) \leq \frac{1}{2\widetilde{C}_0}$, j = 1, 2, the difference $B[Z^1] - B[Z^2]$ fulfils estimate (4.2.13).

4.3 Existence and uniqueness for inverse problem

In this section we study the inverse problem with flux observations in the fixed-point form (4.1.6) and thereupon infer a result for the corresponding generalized inverse problem in the time domain.

Theorem 4.2. Assume that (2.2.2), (4.2.1) hold and

$$\nu_k \in C^2[0,1], \quad k = 1, \dots, K_1.$$
 (4.3.1)

Moreover, let $\det \Gamma \neq 0$ for Γ , given by (4.1.4), and the vector $\Psi = (\Psi_1, \dots, \Psi_K)$, defined in (4.1.8), satisfies

$$\Psi = \frac{d}{p} + Y \in \mathcal{M}_{d,\alpha,\sigma_0} \tag{4.3.2}$$

with some $d \in \mathbb{R}^K$. Then there exists $\sigma_1 \geq \sigma_0$ such that equation (4.1.6) has a unique solution $Z = \frac{c}{p} + V \in \mathcal{M}_{c,\alpha,\sigma_1}$. Here $c = \Gamma^{-1}d$. The solution is unique in the union of spaces $\bigcup_{\substack{\sigma \geq \sigma_1 \\ c \in \mathbb{R}}} \mathcal{M}_{c,\alpha,\sigma}$.

Proof. Setting $c = \Gamma^{-1}d$ the problem (4.1.6) in $\mathcal{M}_{c,\alpha,\sigma}$ is equivalent to the following equation for V in $(\mathcal{A}_{\alpha,\sigma})^K$:

$$V = F(V), (4.3.3)$$

where $F = \Gamma^{-1}\widetilde{F}$ and

$$\widetilde{F} = \widetilde{L_0} \left(\frac{c}{p} + V, B[Z] \right) + \widetilde{L_1} \left(\frac{c}{p} + V \right) + Y,$$
 (4.3.4)

 $\widetilde{L_0}$ is the bilinear operator of $Z\in\mathcal{M}_{c,\alpha,\sigma}$, $B=(B^0,B^1)\in\mathcal{B}_{1,\sigma}$ given by

$$\left(\widetilde{L}_{0}(Z,B)\right)_{i}(p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \int_{0}^{1} pG_{x}(x_{i},y,p)\nu_{k}(y)B^{0}(y,p) dy$$

$$-\sum_{k=K_{1}+1}^{K} Z_{k}(p) \int_{0}^{1} G_{xy}(x_{i},y,p)\mu_{k-K_{1}}(y)B^{1}(y,p) dy, \quad i = 1,\dots,K, (4.3.5)$$

and $\widetilde{L_1}$ is the linear operator of $Z \in \mathcal{M}_{c,\alpha,\sigma}$ defined by

$$\left(\widetilde{L}_{1}(Z)\right)_{i}(p) = \sum_{k=1}^{K_{1}} Z_{k}(p) \left\{ \int_{0}^{1} pG_{x}(x_{i}, y, p)\nu_{k}(y)\varphi(y) dy + \left(\frac{\nu_{k}(x)\varphi(x)}{\beta(x)}\right)'\Big|_{x=x_{i}} \right\} - \sum_{k=K_{1}+1}^{K} Z_{k}(p) \left\{ \int_{0}^{1} G_{xy}(x_{i}, y, p)\mu_{k-K_{1}}(y)\varphi'(y) dy - \frac{\mu_{k-K_{1}}(x_{i})\varphi'(x_{i})}{\lambda(x_{i})} \right\}, \quad i = 1, \dots, K.$$
(4.3.6)

As in the proof of Theorem 3.2 we make use of the fixed-point argument in the balls $D_{\alpha,\sigma}(\rho)$ defined by (3.3.26).

First we obtain some estimates for $\widetilde{L_0}$, $\widetilde{L_1}$. Multiplying by $|p|^{\alpha}$ in (4.3.5) and estimating we have

$$|p|^{\alpha} \left| \left(\widetilde{L_0} \left(\frac{c}{p} + V, B \right) \right)_i(p) \right| \leq \sum_{k=1}^{K_1} \left(\frac{|c_k|}{|p|^{\frac{3}{2} - \alpha}} + \frac{|p|^{\alpha} |V_k(p)|}{\sqrt{|p|}} \right)$$

$$\times \sqrt{|p|} \int_0^1 |G_x(x_i, y, p)| \, dy \, \|\nu_k(x)\|_{C[0,1]} \, |p| \max_{0 \leq x \leq 1} |B^0(x, p)|$$

$$+ \sum_{k=K_1+1}^K \left(\frac{|c_k|}{|p|^{\frac{3}{2} - \alpha}} + \frac{|p|^{\alpha} |V_k(p)|}{\sqrt{|p|}} \right) \int_0^1 |G_{xy}(x_i, y, p)| \, dy$$

$$\times \|\mu_{k-K_1}\|_{C[0,1]} \sqrt{|p|} \max_{0 \leq x \leq 1} |B^1(x, p)|, \qquad i = 1, \dots, K.$$

Using the assumptions (4.2.1), the assertions of Theorems 2.1, 2.5 and the definitions of the norms $\|\cdot\|_{\gamma,\sigma}$ in $\mathcal{A}_{\gamma,\sigma}$, $\hat{\mathcal{B}}_{\gamma,\sigma}$ we obtain

$$|p|^{\alpha} \left| \left(\widetilde{L}_{0} \left(\frac{c}{p} + V, B \right) \right)_{i} (p) \right|$$

$$\leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) C_{2} \|\nu_{k}\|_{C[0, 1]} \|B^{0}\|_{1, \sigma}$$

$$+ \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) C_{4} \|\mu_{k-K_{1}}\|_{C[0, 1]} \|B^{1}\|_{\frac{1}{2}, \sigma}$$

for $i=1,\ldots K$, $\mathrm{Re}p>\sigma$, $\sigma\geq\sigma_0$, $x\in[0,1]$. Taking here the supremum over $\mathrm{Re}p>\sigma$, $x\in[0,1]$ we get

$$\left\| \widetilde{L_0} \left(\frac{c}{p} + V, B \right) \right\|_{\alpha, \sigma} \le \operatorname{Const} \left(\frac{|c|}{\sigma^{\frac{3}{2} - \alpha}} + \frac{\|V\|_{\alpha, \sigma}}{\sqrt{\sigma}} \right) \|B\|_{1, \sigma}, \quad \sigma \ge \sigma_0 \quad (4.3.7)$$

with some constant.

Similarly, multiplying in (4.3.6) by $|p|^{\alpha}$ and estimating we have

$$|p|^{\alpha} \left| \left(\widetilde{L}_{1} \left(\frac{c}{p} + V \right) \right)_{i}(p) \right| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{|p|^{\alpha} |V_{k}(p)|}{\sqrt{|p|}} \right)$$

$$\times \sqrt{|p|} \left| \int_{0}^{1} pG_{x}(x_{i}, y, p) \nu_{k}(y) \varphi(y) \, dy + \left(\frac{\nu_{k}(x) \varphi(x)}{\beta(x)} \right)' \Big|_{x=x_{i}} \right|$$

$$+ \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{|p|^{\alpha} |V_{k}(p)|}{\sqrt{|p|}} \right)$$

$$\times \sqrt{|p|} \left| \int_{0}^{1} pG_{xy}(x_{i}, y, p) \mu_{k-K_{1}}(y) \varphi'(y) \, dy - \frac{\mu_{k-K_{1}}(x_{i}) \varphi'(x_{i})}{\lambda(x_{i})} \right|$$

for $i = 1, \dots, K$. Using here Theorems 2.2, 2.4, 2.6 and 2.8 we get

$$|p|^{\alpha} \left| \left(\widetilde{L}_{1} \left(\frac{c}{p} + V \right) \right)_{i}(p) \right| \leq \sum_{k=1}^{K_{1}} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) \overline{C}_{1} \|\nu_{k} \varphi\|_{C^{2}[0, 1]} + \sum_{k=K_{1}+1}^{K} \left(\frac{|c_{k}|}{|p|^{\frac{3}{2} - \alpha}} + \frac{\|V_{k}\|_{\alpha, \sigma}}{\sqrt{|p|}} \right) \overline{C}_{2} \|\mu_{k-K_{1}} \varphi'\|_{C^{1}[0, 1]}, \quad i = 1, \dots K,$$

where $\overline{C}_1 = \max_{i=1,\dots,K} \{C_7(x_i); C_{10}(x_i)\}$ and $\overline{C}_2 = \max_{i=1,\dots,K} \{C_5; C_8(x_i)\}$. Taking here the supremum over $\text{Re}p > \sigma$, $x \in [0,1]$, we reach the estimate

$$\left\| \widetilde{L_1} \left(\frac{c}{p} + V \right) \right\|_{\alpha, \sigma} \le \operatorname{Const} \left(\frac{|c|}{\sigma^{\frac{3}{2} - \alpha}} + \frac{\|V\|_{\alpha, \sigma}}{\sqrt{\sigma}} \right), \quad \sigma \ge \sigma_0 \tag{4.3.8}$$

with some constant.

Returning to the equation (4.3.3) with the operator $F = \Gamma^{-1}\widetilde{F}$ and taking into consideration (4.3.7), (4.3.8) and the relation $||Y||_{\alpha,\sigma} \leq ||Y||_{\alpha,\sigma_0}$, we obtain

$$\|\widetilde{F}(V)\|_{\alpha,\sigma} \le \operatorname{Const}\left(\frac{1}{\sigma^{\frac{3}{2}-\alpha}} + \frac{\|V\|_{\alpha,\sigma}}{\sqrt{\sigma}}\right) [\|B[Z]\|_{1,\sigma} + 1] + \|Y\|_{\alpha,\sigma_0}$$
 (4.3.9)

with $\sigma \geq \sigma_0$ and some constant depending on |c|.

Further, let us suppose that $V \in D_{\alpha,\sigma}(\rho)$, where σ and ρ satisfy

$$\widetilde{\eta}_0(\rho, \sigma) := \frac{|c|}{\sigma} + \frac{\rho}{\sigma^{\alpha}} \le \frac{1}{2\widetilde{C_0}}$$
 (4.3.10)

and $\sigma \geq \sigma_0$. Then, using Lemmas 2.2, 2.3 for the terms in (4.3.5), (4.3.6) and observing (4.3.4), (4.3.9) we see that $\widetilde{F}(V) \in (\mathcal{A}_{\alpha,\sigma})^K$. Furthermore, by (4.3.10) the inequality (4.2.11) holds, hence we can apply the estimate (4.2.12) of Theorem 4.1 for $\|B[Z]\|_{1,\sigma}$. Plugging (4.2.12) into (4.3.9) and estimating $\|V\|_{\alpha,\sigma}$ by ρ we have

$$\|\widetilde{F}(V)\|_{\alpha,\sigma} \le \operatorname{Const}\left(\frac{1}{\sigma^{\frac{3}{2}-\alpha}} + \frac{\rho}{\sqrt{\sigma}}\right) \left(1 + \frac{\rho}{\sigma^{\alpha-1}}\right) + \|Y\|_{\alpha,\sigma_0}$$
 (4.3.11)

with some constant depending on ρ and σ . From (4.3.11), due to the equality $F = \Gamma^{-1}\widetilde{F}$, we see that for every $\rho > \rho_0 := |\Gamma^{-1}| \, \|Y\|_{\alpha,\sigma_0}$ there exists $\sigma_2 = \sigma_2(\rho) \geq \sigma_0$ such that the inequalities $\widetilde{\eta}_0(\rho,\sigma) \leq \frac{1}{2\widetilde{C}_0}$ and $\|F(V)\|_{\alpha,\sigma} \leq \rho$ hold for any $\sigma \geq \sigma_2(\rho)$. Consequently,

$$F: D_{\alpha,\sigma}(\rho) \to D_{\alpha,\sigma}(\rho) \quad \text{for} \quad \rho > \rho_0 \quad \text{and} \quad \sigma \ge \sigma_2(\rho).$$
 (4.3.12)

Next, we prove that F is a contraction. From (4.3.4), due to the bilinearity of $\widetilde{L_0}$ and the linearity of $\widetilde{L_1}$, we have

$$\widetilde{F}(V^1) - \widetilde{F}(V^2) = \widetilde{L_0}(V^1 - V^2, B[Z^1]) + \widetilde{L_0}(\frac{c}{p} + V^2, B[Z^1] - B[Z^2]) + \widetilde{L_1}(V^1 - V^2),$$

where $Z^1 = \frac{c}{p} + V^1$ and $Z^2 = \frac{c}{p} + V^2$. Using here (4.3.7) and (4.3.8) we get

$$\|\widetilde{F}(V^{1}) - \widetilde{F}(V^{2})\|_{\alpha,\sigma} \leq \operatorname{Const}\left\{\frac{\|V^{1} - V^{2}\|_{\alpha,\sigma}}{\sqrt{\sigma}} \left(\|B[Z^{1}]\|_{1,\sigma} + 1\right) + \left(\frac{1}{\sigma^{\frac{3}{2} - \alpha}} + \frac{\|V^{2}\|_{\alpha,\sigma}}{\sqrt{\sigma}}\right) \|B[Z^{1}] - B[Z^{2}]\|_{1,\sigma}\right\}, \quad \sigma \geq \sigma_{0}$$

with some constant. Supposing that V^1 , $V^2 \in D_{\alpha,\sigma}(\rho)$ with $\sigma \ge \sigma_0$ and ρ such that (4.3.10) hold, by the estimates (4.2.12) and (4.2.13) of Theorem 4.1 we obtain

$$\|\widetilde{F}(V^{1}) - \widetilde{F}(V^{2})\|_{\alpha,\sigma} \le \operatorname{Const}\left\{\left[\frac{1}{\sqrt{\sigma}} + \frac{1}{\sigma^{\alpha}}\left(\frac{1}{\sigma^{\frac{3}{2}-\alpha}} + \frac{\rho}{\sqrt{\sigma}}\right)\right]\left(1 + \frac{\rho}{\sigma^{\alpha-1}}\right) + \frac{1}{\sigma^{\alpha-1}}\right\}\|V^{1} - V^{2}\|_{\alpha,\sigma}$$

with a constant. The coefficient of $\|V^1-V^2\|_{\alpha,\sigma}$ on the right-hand side of this estimate approaches zero, as $\sigma\to\infty$ for a fixed $\rho>0$. Hence, for every $\rho>0$ there exists $\sigma_3=\sigma_3(\rho)\geq\sigma_0$ such that the inequality $\eta_0(\rho,\sigma)\leq\frac{1}{2\widetilde{C}_0}$ holds and $F=\Gamma^{-1}\widetilde{F}$ is a contraction in the ball $D_{\alpha,\sigma}(\rho)$ for $\rho>0$ and $\sigma\geq\sigma_3(\rho)$. This together with (4.3.12) shows that equation (4.3.3) has a unique solution V in every ball $D_{\alpha,\sigma}(\rho)$, where $\rho>\rho_0$ and $\sigma\geq\sigma_4(\rho)=\max\{\sigma_2(\rho);\sigma_3(\rho)\}$. This proves the existence assertion of theorem with $\sigma_1=\sigma_4(2\rho_0)$.

It remains to prove the uniqueness assertion of theorem. The proof of uniqueness in the union $\bigcup_{\sigma \geq \sigma_1} \mathcal{M}_{c,\alpha,\sigma}$ where $c = \Gamma^{-1}d$ is completely the same as in the proof of Theorem 3.2. Thus, let us to show that (4.1.6) has no solution in any space $\mathcal{M}_{\bar{c},\alpha,\sigma}$ where $\bar{c} \neq c = \Gamma^{-1}d$ and $\sigma \geq \sigma_1$. Suppose contrary that (4.1.6) has a solution $Z = \frac{\bar{c}}{p} + V$ in some of such spaces. Then, observing (4.1.6), (4.1.7), (4.3.2) we see that component $V \in (\mathcal{A}_{\alpha,\sigma})^K$ of this solution satisfies the equation

$$\Gamma V = \widetilde{L}_0 \left(\frac{c}{p} + V, B[Z] \right) + \widetilde{L}_1 \left(\frac{c}{p} + V \right) + Y + \frac{\overline{c} - c}{p}, \tag{4.3.13}$$

where and \widetilde{L}_0 and \widetilde{L}_1 are given by (4.3.5), (4.3.6). Observing the assumption (4.3.2) and the estimates (4.3.7), (4.3.8) we see that all terms in (4.3.13) except for $\frac{\overline{c}-c}{p}$ are estimated by quantities of the form $\frac{\mathrm{Const}}{|p|^{\alpha}}$ in the half-plane $\mathrm{Re}\,p>\sigma_0$, where the constant is independent of p and $\alpha>1$. This is not possible in the case $\overline{c}-c\neq 0$. We reached the contradiction. The uniqueness is proved.

Finally, we deduce the following corollary from Theorem 4.2.

Corollary 4.1. Let (2.2.2) hold and the assumptions (4.2.1), (4.3.1), (4.3.2) be satisfied for the functions λ_k , μ_k , φ and the quantities Φ^0 , Φ^1 , Ψ given in terms of the Laplace transforms R, F_1 , F_2 , H_i of the data r, f_1 , f_2 , h_i by formulas (4.1.9), (4.1.8) and either (1.2.14) or (1.2.23) depending on the type of the boundary conditions. Assume that $\det \Gamma \neq 0$ for Γ , given by (4.1.4).

Then the generalized inverse problem with flux observations in the time domain has the unique solution (n, m) with coefficients n_j and m_k of the form

$$n_k(t) = c_k + \frac{1}{2\pi i} \int_{\xi - i\infty}^{\xi + i\infty} e^{tp} Z_k(p) dp, \quad k = 1, \dots K_1,$$

$$m_k(t) = c_{k+K_1} + \frac{1}{2\pi i} \int_{\xi - i\infty}^{\xi + i\infty} e^{tp} Z_{k+K_1}(p) dp, \quad k = 1, \dots K_2,$$

$$(4.3.14)$$

where
$$c = (c_1, ..., c_K) \in \mathbb{R}^K$$
, $Z = (Z_1, ..., Z_K) \in (\mathcal{A}_{\alpha, \sigma})^K$, $\sigma > 0$.

The functions n_k and m_k are continuous for $t \ge 0$. Moreover, the vector c in the formulas (4.3.14) is expressed by $c = \Gamma^{-1}d$, where d is the component of Ψ in the assumption (4.3.2). In addition, $n_k(0) = c_k$, $k = 1, \ldots, K_1$ and $m_k(0) = c_{k+K_1}$, $k = 1, \ldots, K_2$.

Proof. Corollary follows from Theorem 4.2 and Proposition 4.1 together with the inversion formula (1.2.4) of the Laplace transform, the properties 1, 3, 4 of the Laplace transform in Section 1.2.1, the definitions of the spaces $(\mathcal{A}_{\alpha,\sigma})^K$, $\mathcal{M}_{c,\alpha,\sigma}$ and the formula $\mathcal{L}_{t\to p}(c)=\frac{c}{p}$.

4.4 Interpretation of assumptions

4.4.1 Interpretation of regularity assumptions

In this subsection we deduce sufficient conditions in the time domain for the assumptions of Theorem 4.2, given in the Laplace domain. Firstly, we consider the assumptions $\Phi^0 \in \hat{\mathcal{B}}_{1,\sigma_0}$ and $\Phi^1 \in \hat{\mathcal{B}}_{\frac{1}{2},\sigma_0}$ in (4.2.1). Here we let σ_0 be an arbitrary real number (see the remark after (4.2.1)). Observing (4.1.9) and recalling the discussions of Section 3.4.1 we see that $\Phi^0(x,p) = \mathcal{L}_{t\to p}u^0_{xt}(x,t)$ and $\Phi^1(x,p) = \mathcal{L}_{t\to p}u^0_{xt}(x,t)$, where u^0 is the solution of (3.4.38). Thus, by properties 1 and 5 of the Laplace transform in Section 1.2.1 the assumptions (3.4.39) and

$$u^0_{tt}(x,\cdot),\,u_{xt}(x,\cdot)\ \ \text{and}\ \ \tfrac{d^{1/2}}{dt^{1/2}}u^0_{tt}(x,\cdot)\quad \text{belong to \mathscr{E} for $x\in[0,1]$},$$

where the parameters C, σ of the space $\mathscr E$ are independent of x

imply
$$\Phi^0 \in \hat{\mathcal{B}}_{1,\sigma}$$
 and $\Phi^1 \in \hat{\mathcal{B}}_{\frac{1}{2},\sigma}$.

Secondly, we consider the assumption (4.3.2) for $\Psi = (\Psi_1, \dots, \Psi_K)$ given by (4.1.8). Due to the relations $\mathcal{L}_{t\to p}u_t^0(x,t) = -pQ(x,p) - \varphi(x)$ and $\mathcal{L}_{t\to p}h_i'(t) = pH_i(p) - h_i(0)$ (see Section 3.4.1 and property 1 of the Laplace transform), we obtain

$$\mathcal{L}_{t \to p} \left[u_{xt}^{0}(x_i, t) + \frac{h_i'(t)}{\lambda(x_i)} \right] = \Psi_i(p) - \varphi'(x_i) - \frac{h_i(0)}{\lambda(x_i)}. \tag{4.4.1}$$

Let us introduce the consistency conditions

$$-\lambda(x_i)\varphi'(x_i) = h_i(0), \quad i = 1, \dots, K,$$

related to the flux data (1.1.11). In view of these conditions from the formula (4.4.1) and properties 1 and 5 of the Laplace transform, we see that the assumptions

$$\frac{d^j}{dt^j} \left[u_x^0(x_i, \cdot) + \frac{h_i(\cdot)}{\lambda(x_i)} \right] \quad \text{with } j = 0, 1, 2, 1 + \alpha \text{ and some } \alpha$$
satisfying (2.1.3) belong to \mathscr{E} for $i = 1, \dots, K$

imply (4.3.2).

4.4.2 Interpretation of non-vanishing condition

In this subsection we analyse the non-vanishing condition $\det \Gamma \neq 0$ for the matrix Γ given by (4.1.4). Let us begin by the piecewise homogeneous case, i.e. when $K_1 = K_2$ and ν_k , μ_k are given by (1.1.9). In this case it is natural to suppose that the given functions $\beta(x)$ and $\lambda(x)$ are also piecewise homogeneous, i.e.

$$\beta(x) = \sum_{k=1}^{K_1} \beta_k \nu_k(x), \quad \lambda(x) = \sum_{k=1}^{K_1} \lambda_k \mu_k(x)$$
 (4.4.2)

with some β_k , $\lambda_k > 0$. Then, choosing x_i like in Section 3.4.2 (see (3.4.51)), the matrix Γ is singular because it contains rows that are pairwise linearly dependent. This means that the inverse problem with flux observations is problematic in the piecewise homogeneous case. We could get regular matrix Γ if we drop the physical assumption (4.4.2), i.e. let β and λ change in the intervals $(y_{k-1} + \epsilon, y_k - \epsilon)$. For instance, if

$$\beta'(x_k) \neq 0, \ k = 1, \dots, K, \quad \frac{\beta(x_k)}{\lambda(x_k)} \neq \frac{\beta(x_{k+1})}{\lambda(x_{k+1})}, \ k = 1, 3, \dots, K-1,$$

then setting $\varphi = \beta^2$ we have

$$\det \Gamma = 2(-1)^{\frac{K_1(K_1-1)}{2}} \prod_{l=1}^{K_1} \beta'(x_{2l-1}) \beta'(x_{2l}) \left[\frac{\beta(x_{2l})}{\lambda(x_{2l})} - \frac{\beta(x_{2l-1})}{\lambda(x_{2l-1})} \right] \neq 0.$$

Also we could get regular matrices in case we take some measurement points in the transmission intervals $(y_k - \epsilon, y_k + \epsilon)$. But in these intervals the functions ν_k change rapidly, hence incorporation of derivatives of ν_k in the matrix Γ is very inaccurate

Next, let us us analyze the usage of power basis. Setting $\nu_k(x) = \mu_k(x) = x^{k-1}$ the matrix Γ reads

$$\Gamma = \left(\left(\left[\frac{x^k \varphi(x)}{\beta(x)} \right]' \bigg|_{x=x_i} \right)_{k=0,\dots,K_1-1} \left(\frac{x^k \varphi'(x_i)}{\lambda(x_i)} \right)_{k=0,\dots,K_2-1} \right)_{i=1,\dots,K}.$$

This is a complicated expression. It is quite difficult to see what initial functions φ and measurement points x_i provide regular Γ for given functions β and λ . Even

4.5. The case of singular matrix. Severe ill-posedness of homogeneous problem

more, in some cases of β and λ the matrix Γ is always singular. This is so, for instance, when $\beta(x)=\mathrm{Const}$ and $\lambda(x)=\mathrm{Const}$. Indeed, then the first and K_1 's column of Γ are linearly dependent. On the other hand, in some cases of β and λ it is possible to choose φ and x_i so that the regularity holds. For example, when $\frac{(x\beta)'}{\lambda}$ is a polynomial of the order K_1 , i.e. $\frac{(x\beta(x))'}{\lambda(x)}=a_0x^{K_1}+a_1x^{K_1-1}+\ldots+a_{K_1}$ with $a_0\neq 0$ then defining $\varphi(x)=x\beta(x)$ we have

$$\det \Gamma \, = \, K_1! \, a_0^{K_2} \det \left(x_i^k \right)_{\substack{k=0,\dots,K-1 \\ i=1,\dots,K}} \neq 0.$$

We obtain simpler determinant in case we use the weighted power basis instead of the usual one. For instance, defining $\nu_k(x) = a(x)x^{k-1}$, $\mu_k(x) = b(x)x^{k-1}$ with $a(x) = \beta(x)e^x$ and $b(x) = \lambda(x)e^x$ we deduce

$$\det \Gamma = \exp \left[\sum_{i=1}^{K} x_i \right]$$

$$\times \det \left(\left(x_i^k \varphi'(x_i) + (k x_i^{k-1} + x_i^k) \varphi(x_i) \right)_{k=0,\dots,K_1-1} \left(x_i^k \varphi'(x_i) \right)_{k=0,\dots,K_2-1} \right)_{i=1,\dots,K}.$$

It is not difficult to construct proper initial functions φ for it. For example, under the assumption $K_1 \leq K_2$ we can take $\varphi(x) = \exp\left(\frac{x^{K_1+1}}{K_1+1}\right)$ to get

$$\det \Gamma = \exp \left[\sum_{i=1}^{K} \left(x_i + \frac{x_i^{K_1+1}}{K_1+1} \right) \right] \det \left(x_i^k \right)_{k=0,\dots,K-1} \neq 0.$$

4.5 The case of singular matrix. Severe ill-posedness of homogeneous problem

The natural question arises: what to do when the matrix Γ in (4.1.4) (or (3.1.7)) is singular? Such a case means that the principal part extracted form the problem appears to be generate. One could try to overcome this difficulty separating some higher order regular principal part from the system.

This can be done as follows. Let $\operatorname{rank} \Gamma = \rho < K$. Then the system $\Gamma Z = \mathcal{F}(Z)$ contains a subsystem of ρ equations formed by indices $i = i_1, \ldots i = i_\rho$ such that

$$\operatorname{rank}\left(\widetilde{\gamma}_{jk}\right)_{\substack{j=1,\ldots,\rho\\k=1,\ldots,K}} = \rho \quad \text{where } \ \widetilde{\gamma}_{jk} = \gamma_{i_j,k}.$$

Let us denote this subsystem by S_1 .

Further, let us compute $K-\rho$ different linear combinations of the equations of the system $\Gamma Z=\mathcal{F}(Z)$ such that the left-hand sides of these linear combinations are zero. Thereupon, we use the ideas of Section 4.1 (Section 3.1) to treat these linear combinations. Namely we multiply them by some power of p and separate

the principal parts in the process $\operatorname{Re} p \to +\infty$. Suppose that for a properly chosen power of p, these principal parts have the forms

$$\widetilde{\gamma}_{j1}Z_1(p) + \ldots \widetilde{\gamma}_{jK}Z_K(p),$$

where $j=\rho+1,\ldots,K$ and $\widetilde{\gamma}_{j1},\ldots,\widetilde{\gamma}_{jK}$ are some numbers. Shifting the principal parts to the left-hand sides we obtain a system of $K-\rho$ equations. Denote it by S_2 .

Putting S_1 and S_2 together we reach the new inverse problem system

$$\widetilde{\Gamma}Z = \widetilde{\mathcal{F}}(Z)$$

where $\widetilde{\Gamma}=(\widetilde{\gamma}_{jk})_{j,k=1,\dots,K}$. In case $\det\widetilde{\Gamma}\neq 0$, we can prove an existence and uniqueness theorem for this system using the same technique as in Theorem 4.2 (Theorem 3.2). In case $\widetilde{\Gamma}$ is still singular, we repeat the described procedure with the system $\widetilde{\Gamma}Z=\widetilde{\mathcal{F}}(Z)$ and so on.

However, in some particular cases of singular Γ the described method doesn't work, because the matrices corresponding to all higher principal parts are singular. An example is the inverse problem with flux observations in the case of the homogeneous rod. Let us analyze this problem. Then $\beta(x) = \beta = \mathrm{Const} > 0$, $\lambda(x) = \lambda = \mathrm{Const} > 0$, $K_1 = K_2 = 1$, $\nu_1(x) = \mu_1(x) = 1$ and the number of unknowns reduces to two: $n(t) = n_1(t)$, $m(t) = m_1(t)$. For the sake of simplicity, let us be limited to the boundary conditions of the first kind.

The integro-differential equation in the Laplace domain (i.e. (1.2.8)) is

$$\beta[pU(x,p) - \varphi(x)] + pN(p)U(x,p) = [\lambda - M(p)]U_{xx}(x,p) + R(x,p)$$

with $x \in (0,1)$ and $N(p) = \mathcal{L}_{t\to p}n(t)$, $M(p) = \mathcal{L}_{t\to p}m(t)$. Let us define

$$\omega(p) = \frac{p(\beta + N(p))}{\lambda - M(p)}. (4.5.1)$$

Then we can rewrite the differential equation in the form

$$U_{xx}(x,p) - \omega(p)U(x,p) = -\frac{R(x,p) + \beta\varphi(x)}{\lambda - M(p)}, \ x \in (0,1).$$
 (4.5.2)

Let $G[\omega](x,y,p)$ stand for the Green function of the operator $\frac{d^2}{dx^2} - \omega(p)$ with the first kind boundary conditions, i.e. $G[\omega](x,y,p)$ solves the problem

$$\left(\frac{d^2}{dy^2} - \omega(p)\right) G[\omega](x, y, p) = \delta(y - x), \quad x \in (0, 1), \quad y \in (0, 1),$$

$$G[\omega](x, 0, p) = G[\omega](x, 1, p) = 0.$$

Then the solution of the boundary value problem (4.5.2), (1.2.9) is

$$U(x,p) = -\frac{1}{\lambda - M(p)} \int_0^1 G[\omega](x,y,p) [R(y,p) + \beta \varphi(y)] dy$$
$$-G[\omega]_y(x,0,p) F_1(p) + G[\omega]_y(x,1,p) F_2(p). \tag{4.5.3}$$

The additional conditions in the Laplace domain (see (1.2.26)) are

$$[-\lambda + M(p)]U_x(x_i, p) = H_i(p), \quad i = 1, 2.$$

Plugging (4.5.3) into these conditions we arrive at the following system for the unknowns N and M:

$$\int_{0}^{1} G[\omega]_{x}(x_{i}, y, p)[R(y, p) + \beta \varphi(y)]dy$$

$$+[\lambda - M(p)] \Big[G[\omega]_{xy}(x_{i}, 0, p)F_{1}(p) - G[\omega]_{xy}(x_{i}, 1, p)F_{2}(p)\Big]$$

$$= H_{i}(p), \quad i = 1, 2. \tag{4.5.4}$$

It is convenient to make a change of variables in (4.5.4). Let us introduce the new pair of unknowns (W, S) related to (N, M) as follows:

$$W(p) = \frac{\lambda N(p) + \beta M(p)}{\lambda(\lambda - M(p))}, \quad S(p) = M(p). \tag{4.5.5}$$

Evidently, the problems for (N,M) to (W,S) are equivalent, because the system (4.5.5) is uniquely solvable for (N,M). According to the convolution rule (property 2 of the Laplace transform in Section 1.2.1) the time domain originals $w(t) = \mathcal{L}_{p \to t}^{-1} W(p)$ and $s(t) = \mathcal{L}_{p \to t}^{-1} S(p)$ are related to n and m by the following equations containing a Volterra equation of the second kind

$$\lambda^2 w(t) - \lambda \int_0^t m(t - \tau) w(\tau) d\tau = \lambda n(t) + \beta m(t), \quad t > 0,$$

$$s(t) = m(t).$$
(4.5.6)

Performing the mentioned change of variables in (4.5.4) we get

$$\int_{0}^{1} G[\omega]_{x}(x_{i}, y, p)[R(y, p) + \beta \varphi(y)]dy$$

$$+[\lambda - S(p)] \Big[G[\omega]_{xy}(x_{i}, 0, p)F_{1}(p) - G[\omega]_{xy}(x_{i}, 1, p)F_{2}(p)\Big]$$

$$= H_{i}(p), \quad i = 1, 2. \tag{4.5.7}$$

Due to (4.5.1) and (4.5.5) we have

$$\omega(p) = p\left(\frac{\beta}{\lambda} + W(p)\right). \tag{4.5.8}$$

Thus, ω depends only on the unknown W. We see that first addend in the left-hand side of (4.5.7) depends only on the unknown W.

In the particular case when the boundary conditions are zero, e.g., when $f_1 = f_2 = 0$, $F_1 = F_2 = 0$, the second addend in the left-hand side of (4.5.7) vanishes. This implies that the solution of the inverse problem is not unique. The system

contains information about W, but not about S. The component S of the solution can be arbitrary. In terms of the preliminary unknowns this means that for any W satisfying the system (4.5.7) we can assign infinitely many pairs (n,m) satisfying the first equation in (4.5.6) with $w(t) = \mathcal{L}_{p \to t}^{-1} W(p)$.

The problem is bad in the case of non-zero boundary conditions, too. To study this case we make use of the formula

$$G[\omega](x,y,p) = -\frac{1}{\sqrt{\omega(p)} sh\sqrt{\omega(p)}} \left\{ \begin{array}{ll} sh\sqrt{\omega(p)}x \; sh\sqrt{\omega(p)}(y-1) & x < y \\ sh\sqrt{\omega(p)}y \; sh\sqrt{\omega(p)}(x-1) & x > y \end{array} \right.$$

of the Green function. Here $\sqrt{\omega(p)}$ is the main branch of the square root of $\omega(p)$. We note that $\sqrt{\omega(p)}$ is uniquely defined in some half-plane $\operatorname{Re} p > \sigma$. Indeed, due to $M(p), N(p) \to 0$, as $\operatorname{Re} p \to +\infty$ uniformly in $\operatorname{Im} p$, we have $W(p) \to 0$, as $\operatorname{Re} p \to +\infty$ uniformly in $\operatorname{Im} p$, hence in view of (4.5.8), we get

$$\omega(p) \sim \frac{\beta}{\lambda} p$$
, as $\operatorname{Re} p \to +\infty$ uniformly in $\operatorname{Im} p$. (4.5.9)

This, due to the uniqueness of $\sqrt{\frac{\beta}{\lambda}p}$ in $\operatorname{Re} p>0$, implies the uniqueness of $\sqrt{\omega(p)}$ in some half-plane $\operatorname{Re} p>\sigma$.

From the formula of $G[\omega]$ we immediately get

$$G[\omega]_{xy}(x_i, 0, p) = -\frac{\sqrt{\omega(p)}}{sh\sqrt{\omega(p)}} ch\sqrt{\omega(p)} (1 - x_i),$$

$$G[\omega]_{xy}(x_i, 1, p) = -\frac{\sqrt{\omega(p)}}{sh\sqrt{\omega(p)}} ch\sqrt{\omega(p)} x_i.$$

In view of the definition of $\sqrt{\omega(p)}$, the inequalities $\beta, \lambda > 0$ and (4.5.9), we have $\operatorname{Re} \sqrt{\omega(p)} > 0$ for $\operatorname{Re} p > \sigma$. Thus, by means of (4.5.9) and (2.2.20), we derive the estimates

$$|G[\omega]_{xy}(x_{i}, 0, p)| \leq \sqrt{|\omega(p)|} \frac{ch\operatorname{Re}\sqrt{\omega(p)}(1-x_{i})}{sh\operatorname{Re}\sqrt{\omega(p)}}$$

$$\leq \operatorname{Const} e^{-\operatorname{Re}\sqrt{\omega(p)}x_{i}} \leq \operatorname{Const} e^{-x_{i}\sqrt{\frac{\beta}{2\lambda}}\sqrt{|p|}},$$

$$|G[\omega]_{xy}(x_{i}, 1, p)| \leq \sqrt{|\omega(p)|} \frac{ch\operatorname{Re}\sqrt{\omega(p)}(x_{i})}{sh\operatorname{Re}\sqrt{\omega(p)}}$$

$$\leq \operatorname{Const} e^{-\operatorname{Re}\sqrt{\omega(p)}(x_{i}-1)} \leq \operatorname{Const} e^{(x_{i}-1)\sqrt{\frac{\beta}{2\lambda}}\sqrt{|p|}}$$

$$(4.5.10)$$

for Re $p > \sigma$. Since $x_i \in (0,1)$ and $F_j(p) \to 0$ as Re $p \to +\infty$ uniformly in Im p, from (4.5.10) we obtain the estimate

$$\left|G[\omega]_{xy}(x_i,0,p)F_1(p) - G[\omega]_{xy}(x_i,1,p)F_2(p)\right| \le \frac{C_{\kappa}}{|p|^{\kappa}}, \quad \operatorname{Re} p > \sigma_{\text{(4.5.11)}}$$
 for any $\kappa > 0$ and $i = 1,2$,

4.5. The case of singular matrix. Severe ill-posedness of homogeneous problem

where C_{κ} are constants.

The estimate (4.5.11) shows that the method of this paper is not applicable for the problem to determine the pair (W,S). Indeed, we cannot find a factor p^l with l>0 such that the equations (4.5.7) multiplied with p^l have principal parts containing S(p) in the process $\operatorname{Re} p \to +\infty$. The term with S(p) exponentially vanishes as $\operatorname{Re} p \to +\infty$.

This means that the problem for (w, s) is severely ill-posed with respect to s. Indeed, in the time domain the system (4.5.7) reads

$$q_i^1[w](t) + \int_0^t s(\tau)q_i^2[w](t-\tau)d\tau = h_i(t), \ t>0, \ i=1,2,$$
 (4.5.12)

where

$$q_i^1[w](t) = \mathcal{L}_{p \to t}^{-1} \int_0^1 G[\omega]_x(x_i, y, p) [R(y, p) + \beta \varphi(y)] dy + \lambda \mathcal{L}_{p \to t}^{-1} \Big[G[\omega]_{xy}(x_i, 0, p) F_1(p) - G[\omega]_{xy}(x_i, 1, p) F_2(p) \Big], q_i^2[w](t) = -\mathcal{L}_{p \to t}^{-1} \Big[G[\omega]_{xy}(x_i, 0, p) F_1(p) - G[\omega]_{xy}(x_i, 1, p) F_2(p) \Big].$$

Therefore, by the properties 1 and 4 of the Laplace transform in Section 1.2.1 and (4.5.11), we see that the operator that assigns to s the s-dependent terms in (4.5.12) is infinitely smoothing, i.e for any $s \in \mathscr{E}$ we have

$$\int_0^{\cdot} s(\tau) q_i^2[w](\cdot - \tau) d\tau \in C^{\infty}[0, \infty), \quad i = 1, 2,$$

$$\frac{d^j}{dt^j} \int_0^t s(\tau) q_i^2[w](t - \tau) d\tau \Big|_{t=0} = 0, \quad j = 0, 1, 2, \dots, \quad i = 1, 2.$$

We also observe that in case w is a priori known, the inverse problem to determine s on the basis of a single measurement h_1 is linear and can be reduced to the Volterra equation of the first kind with infinitely smoothing operator:

$$\int_0^t s(\tau)q_1^2[w](t-\tau)d\tau = h_1(t) - q_1^1[w](t), \ t > 0.$$

Abstract

In the present thesis inverse problems to determine kernels of heat flux and internal energy in the one-dimensional non-homogeneous case were studied0 We conside/red the case when these kernels are degenerate, i.e. representable as sums of the known space-dependent functions times the unknown time-dependent coefficients. We posed and proved existence and uniqueness of two problems of such kind. The first one is a problem with purely temperature observations. Then the kernel of internal energy is determined with higher smoothness than the kernel of heat flux. The second one is the problem with purely flux observations. Then the kernels of internal energy and heat flux are determined with the same level of accuracy. The solvability theorems are complemented with sufficient conditions for the assumptions of the theorems in the time domain. Moreover, it has been shown that the homogeneous inverse problem with flux observations is severely ill-posed.

The method involves application of the Laplace transform to the problems, reduction of the transformed problem to a fixed-point form and application of the contraction principle in some complex right half-plane.

Pöördülesanded mittehomogeensete mälutuumade määramiseks soojuslevi protsessis

Mäluga materjalide korral sisaldab soojusjuhtivuse võrrand konvolutsioonitüüpi integraale, milles on ajast sõltuvad tuumafunktsioonid. Need aine mälu väljendavad liikmed põhjustavad teatud inertsi soojusprotsessides ja võimaldavad modelleerida soojuse levimist lõpliku kiirusega [3, 14, 26, 27, 33, 34]. Mäluga materjalide mudelit kasutatakse ka termoelastsuse teoorias [40] ja faasiüleminekuga protsesside kirjeldamiseks [1, 4, 5].

Kui mälutuumad on teada, siis saab lahendada pöördülesannet, leides tundmatud tuumafunktsioonid. Pöördüesande lahendamiseks kasutatakse lisainformatsiooni (vaatlusandmeid), mõõtes temperatuure või soojusvoogu.

Pöördülesandeid kasutatakse põhiliselt kahel eesmärgil:

- Matemaatilise mudeli adekvaatsuse hindamiseks.
 Kui erinevate lähteandmete korral saadakse piisavalt hästi kokkulandevad lahendid, siis mudel töötab. Kui aga tulemused erinevad tunduvalt, siis kirjeldab mudel protsesse halvasti ja on kõrbmatu.
- 2. Konkreetsete materjalide mälutuumade leidmiseks.

Erinevaid pöördülesandeid homogeensete materjalide mälutuumade leidmiseks on uuritud paljudes artiklites, nagu [2, 10, 11, 12, 16, 18, 19, 20, 22, 25, 29, 30, 31, 41, 42, 43]. Enamus töid vaatlevad juhtu, kui soojusvõrrand sisaldab vaid soojusvoo mälutuuma. Artiklites [11, 20] vaadeldakse kahe mälutuuma, nii soojusvoo kui ka siseenergia mälutuuma üheaegset leidmist. Niisugune pöördülesanne sõltub oluliselt vaatlusandmete tüübist. Juhul, kui mõõdetakse kahes punktis temperatuure, leitakse siseenergia mälutuum siledama funktsioonina, kui soojusvoo mälutuum [20]. Kasutades üheaegselt nii temperatuuri kui ka soojusvoo näite, leitakse mälutuumad sama siledusega [11, 20]. Nendes artiklites ei vaadelda ülesannet, milles mõõdetakse vaid soojusvoogu. See juhtum on keerulisem.

Kui materjal on mittehomogeenne, siis sõltuvad mälutuumad ka ruumimuutuja(te)st. Sel juhul vajavad pöördülesanded rohkem vaatlusandmeid. Kõige üldi/sem uurimismeetod on Dirichlet-Neumanni teisenduse rakendamine soojusvoo mälutuuma leidmiseks [15]. See meetod nõuab suurt hulka vaatlusandmeid, mis on saadud erinavatel katsetel. Olukord lihtsustub, kui mälutuumade kohta on teada täiendavat *a priori* informatsiooni. Näiteks, kui vaadeldav keha on kihiline või rahuldab teatud sümmeetriatingimusi, saab soojusvoo mälutuuma leida üheainsa katse mõõtmistulemuste abil [6, 7, 9, 17, 32].

Teatud juhtudel on mittehomogeensed mälutuumad kõdunud, see tähendab,

et nad kujutavad endast lõplikku summat korrutistest, milles tuntud ruumimuutuja(te)st sõltuvad funktsioonid on korrutatud tundmatute ajast sõltuvate kordajatega. Selline olukord tekib tükati homogeense aine korral või siis juhul, kui tegelik tuum on asendatud lihtsustatud lähendusega. Tundmatud kordajad saab leida temperatuuri või soojusvoo mõõtmisega lõplikus arvus punktides. Selleks läheb vaja vaid ühte eksperimenti. Töödes [21, 23, 24] vaadeldakse seda tüüpi pöördülesannet vaid soojusvoo mälutuuma leidmiseks.

Antud väitekirja põhieesmärgiks on uurida pöördülesandeid mõlema kõdunud mälutuuma üheaegseks leidmiseks mittehomogeense aine korral. Piirdutakse ühemõõtmelise juhuga ja vaadeldakse kahte erinevat ülesannet.

Esimese ülesande korral kasutatakse vaid temperatuuri mõõtmisi. Tõestatakse lahendi olemasolu ja ühesus. Seejuures saadakse siseenergia mälutuum ühe tuletise võrra siledam, kui soojusvoo mälutuum. Selle ülesande uurimise tulemused esimest liiki rajatingimuste korral on avaldatud autori töös [37] ja kolmandat liiki rajatingimuste korral töös [38].

Teiseks pöördülesandeks on ülesanne, mille korral kasutatakse vaid soojusvoo mõõtmisi. Seda ülesannet vaadeldakse esmakordselt ja selle käsitlemine on keerulisem. Laplace'i kujutiste ruumis vaadeldakse vastavat elliptilist võrrandit ja selle Greeni funktsiooni. Greeni funktsiooni asümptootiliste hinnangute leidmine nõuab suurt tehnilist tööd. Selle ülesande uurimise tulemused on avaldatud töödes [35, 36]. Väitekirjas on parandatud ka üks eksitus, mis esineb neis artiklites. Hinnangud (2.4.7) ja (2.5.7) sisaldavad seal liiget $\frac{V'(x)}{\beta(x)}$, $\left(\frac{V(x)}{\beta(x)}\right)'$ asemel. Lisaks on veel tõestatud selle ülesande lahendi olemasolu ja ühesus kolmandat liiki rajatingimuste korral. Neid tulemusi pole autor veel jõudnud avaldada artiklina.

Pöördülesanne soojusvoo mõõtmisega on oma iseloomult halvem, kui tempera/tuuri mõõtmisega ülesanne. Paljudel juhtudel on ülesande maatriks singulaarne. Vastav homogeenne ülesanne on isegi tugevalt mittekorrektne. Siin märgitakse seda esmakordselt.

Väitekirja tulemumuste tutvustamiseks olid tehtud järgmised ettekanded:

- 1. Rahvusvaheline konverents "5-th International Conference on Inverse Problems in Engineering", Cambridge, 11 15.07.2005 poster.
- 2. Seminar "Funktsionaalanalüüsi rakendused", Tallinn, 03.03.2006 loeng.
- 3. 5. rahvusvaheline konverents "Simulation and Optimization in Business and Industry", Tallinn, 17 20.05.2006 ettekanne.
- 4. Rahvusvaheline Matemaatikute Kongress "International Congress of Mathematicians", Madriid, 22 30.08.2006 lühiettekanne.

References

- [1] G. Bonfanti and L. Luterotti, Regularity and convergence results for a phase-field model with memory. *Math. Meth. Appl. Sci.* **21** (1998), 1085–1105.
- [2] A. L. Bukhgeim, Inverse problems of memory reconstruction. *J. Inv. Ill-Posed Problems* **1** (1993), 193–205.
- [3] B. D. Coleman and M. E. Gurtin, Equipresence and constitutive equation for rigid heat conduction. *Z. Angew. Math. Phys.* **18** (1967), 199–208.
- [4] P. Colli, G. Gilardi, M. Grasselli and G. Schimperna, The conserved phase-field system with memory. *Adv. Math. Sci. Appl.* **11** (2001), 265–292.
- [5] F. Colombo, Direct and inverse problems for a phase-field model with memory. *J. Math. Anal. Appl* **260** (2001), 517–545.
- [6] F. Colombo and A. Lorenzi, Identification of time and space dependent relaxation kernels for materials with memory related to cylindrical domains I. J. Math. Anal. Appl. 213 (1997), 32–62.
- [7] F. Colombo and A. Lorenzi, Identification of time and space dependent relaxation kernels for materials with memory related to cylindrical domains II. *J. Math. Anal. Appl.* **213** (1997), 63–90.
- [8] G. Doetsch, Einführung in Theorie und Anwendung der Laplace-Transformation. Birkhäuser Verlag, Basel, 1958.
- [9] A. Favaron and A. Lorenzi, Parabolic integrodifferential identification problems related to memory kernels with special symmetries. *J. Inv. Ill-Posed Problems* **11** (2003), 18–28.
- [10] G. Gentili, Dissipativity conditions and inverse problems for heat conduction with linear memory. *Inverse problems* **7** (1991), 77–84.
- [11] M. Grasselli, An identification problem for a linear integro-differential equation occurring in heat flow. *Math. Meth. Appl. Sci.* **15** (1992), 167–186.
- [12] M. Grasselli and A. Lorenzi, An inverse problem for an abstract nonlinear parabolic integrodifferential equation. *Diff. Int. Equ.* **6** (1993), 63–81.

- [13] R. B. Gunther and J. W. Lee, *Partial Differential Equations of Mathematical Physics and Integral Equations*. Prentice-Hall, Englewood Cliffs, New Jersey, 1988.
- [14] M. E. Gurtin and A. C. Pipkin, A general theory of the heat conduction with finite wave speeds. *Arch. Rational Mech. Anal.* **31** (1968), 113–126.
- [15] J. Janno, Determination of a time- and space-dependent heat flux relaxation function by means of a restricted Dirichlet-to-Neumann operator. *Math. Meth. Appl. Sci.* **27** (2004), 1241–1260.
- [16] J. Janno, E. M. Kiss and L. v. Wolfersdorf, On Tikhonov regularization for identifying memory kernels in heat conduction and viscoelasticity. *Z. Angew. Math. Mech.* **80** (2000), 259–272.
- [17] J. Janno and A. Lorenzi, A parabolic integro-differential identification problem in a barrelled smooth domain. *Z. Anal. Anwen.* **25** (2006), 103 130.
- [18] J. Janno and L. v. Wolfersdorf, Inverse problems for identification of memory kernels in heat flow. *J. Inv. Ill-Posed Problems* **4** (1996), 39–66.
- [19] J. Janno and L. v. Wolfersdorf, Identification of weakly singular memory kernels in heat conduction. *Z. Angew. Math. Mech.* **77** (1997), 243–257.
- [20] J. Janno and L. v. Wolfersdorf, Identification of memory kernels in general linear heat flow. *J. Inv. Ill-Posed Problems* **6** (1998), 141–164.
- [21] J. Janno and L. v. Wolfersdorf, An inverse problem for identification of a time- and space-dependent memory kernel of a special kind in heat conduction. *Inverse Problems* **15** (1999), 1455–1467.
- [22] J. Janno and L.v. Wolfersdorf, Inverse problems for memory kernels by Laplace transform methods. *Z. Anal. Anwen.* **19** (2000), 489–510.
- [23] J. Janno and L. v. Wolfersdorf, Identification of a special class of memory kernels in one-dimensional heat flow. *J. Inv. Ill-Posed Problems* **9** (2001), 389–411.
- [24] J. Janno and L. v. Wolfersdorf, Identification of memory kernels in one-dimensional heat flow with boundary conditions of the third kind. *Inverse Problems in Engineering* **9** (2001), 179 198.
- [25] J.Janno and L.v.Wolfersdorf, Identification of exponentially decreasing memory kernels in heat conduction and viscoelasticity by finite-dimensional inverse problems. *J. Inv. Ill-Posed Problems* **13** (2005), 65–92.
- [26] D. D. Joseph and L. Preziosi, Heat waves. Rev. Mod. Phys. **61** (1989), 47–71.

- [27] D. D. Joseph and L. Preziosi, Addendum to the paper "Heat waves". Rev. Mod. Phys. 62 (1990), 375–394.
- [28] O. A. Ladyzhenskaya, V. A. Solonnikov and N. N. Ural'tseva, *Linear and Quasilinear Equations of Parabolic Type*. Nauka, Moscow (1967) (in Russian). English translation: American Mathematical Society, Providence, Rhode Island (1968).
- [29] A. Lorenzi, Identification problems for integrodifferential equations. In: *Ill-Posed Problems in Natural Sciences* (Ed. A. Tikhonov), TVP Sci. Publ., Moscow, 1992, pp. 342–366.
- [30] A. Lorenzi and E. Paparoni, Direct and inverse problems in the theory of materials with memory. *Rend. Sem. Math. Univ. Padova* **87** (1992), 105–138.
- [31] A. Lorenzi and E. Sinestrari, An inverse problem in the theory of materials with memory. *Nonlin. Anal., Theory, Meth. and Appl.* **12** (1988), 1317–1335.
- [32] A. Lorenzi and V. G. Yakhno, An identification problem related to an isotropic nonhomogeneous stratified viscoelastic body. *J. Inv. Ill-Posed Problems* **5** (1997), 29 53.
- [33] A. Lunardi, On the linear equation with fading memory. *SIAM J. Math. Anal.* **21** (1970), 1213–1224.
- [34] J. W. Nunziato, On heat conduction in materials with memory. *Quart. Appl. Math.* **29** (1971), 187–204.
- [35] E. Pais, Identification of degenerate time- and space-dependent kernels in heat flow. In: *Proc. 5-th International Conference on Inverse Problems in Engineering* (Cambridge, 11-15.07.2005), D. Lesnic ed., Leeds Univ. Press, Leeds, 2005, Vol. III, P01, pp. 1–10.
- [36] E. Pais, Degenerate memory kernels identification problem with flux-type additional conditions. *J. Inv. Ill-Posed Problems* **14** (2006), 397 418.
- [37] E. Pais and J. Janno, Identification of two degenerate time- and space-dependent kernels in a parabolic equation. *Electron. J. Diff. Eqns.* (http://www.emis.de/journals/EJDE) 2005, No. 108, 1–20.
- [38] E. Pais and J. Janno, Inverse problem to determine degenerate memory kernels in heat flux with third kind boundary conditions. *Math. Model. Anal.* 11 (2006), 427–450.
- [39] I. G. Petrowski, *Vorlesungen über partielle Differentialgleichungen*. Teubner, Leipzig, 1955.
- [40] J. Prüß, Evolutionary Integral Equations and Applications. Birkhäuser Verlag, Boston, 1993.

- [41] F. Unger and L. v. Wolfersdorf, On a control problem for memory kernels in heat conduction. *Z. Angew. Math. Mech.* **75** (1995), 365–370.
- [42] L. v. Wolfersdorf, On identification of memory kernels in linear theory of heat conduction. *Math. Meth. Appl. Sci.* **17** (1994), 919–932.
- [43] L.v. Wolfersdorf, Inverse problems for memory kernels in heat flow and viscoelasticity. *J. Inv. Ill-Posed Problems* **4** (1996), 341–354.

Elulookirjeldus

1. Isikuandmed:

Ees- ja perekonnanimi: Enno Pais

Sünniaeg ja -koht: 30.08.1944, Venemaa

Kodakondsus: Eesti Perekonnaseis: abielus Lapsed: 3 tütart

2. Kontaktandmed:

Aadress: Paekaare'36-140

Telefon: (+372) 56684492, (+372) 6363975

E-posti aadress: ennopais@stv.ee

3. Hariduskäik:

Õppeasutus Lõpetamise aeg Haridus (eriala/kraad)

Moskva Riiklik Ülikool 1967 matemaatik, diplom(5a,

MSc ekvivalent)

Pärnu II Keskkool 1962 keskharidus

4. Keelteoskus:

Keel Keel Tase Tase Soome Vene kõrgtase kesktase kõrgtase Rootsi algtase prantsuse Inglise kesktase Hispaania algtase

Saksa algtase

5. Täiendwuõpe:

Õppimise aeg Õppeasutuse või muu organisatsiooni nimetus

1977 - 1978 Leningradi Riiklik Ülikool, prantsuse keele kur-

hear

1970 - 1973 Moskva Riiklik Ülikool, Arvutusmatemaatika ja

Küberneetika teaduskonna aspirantuur

6. Teenistuskäik:

Ülikooli, teadusasutuse või muu organisatsiooni nimetus	Töötamise aeg	Ametikoht
Tallinna Tehnikaülikool, Matemaatika Instituut	1991 -	lektor
Tallinna Polütehniline Instituut, Matemaatika kateeder	1981 -"1991	vanemõpetaja
Bamako Kõrgem Inseneride Kool, Mali vabariik, Aafrika	1978 - 1981	matemaatika õppejõud
Tallinna Polütehniline Instituut, Matemaatika kateeder	1973 -"1978	assistent
Tallinna Polütehniline Instituut, Arvutus- matemaatika kateeder	1967 -"1970	assistent

7. Kaitstud lõputööd:

Diplomitöö teemal: Iteratsioonmeetodid silindrilise kooriku tasakaaluülesande võrrandisüsteemi lahendamiseks diferentskujul.

8. Teadustegevus:

Enno Pais. O gladkosti sobstvennõh funktsi sžatoi plastinõ okolo uglovõh totšek. *Differentsialnõje Uravnenija*, Vol. 9, 9, 1700 - 1706, 1973.

Enno Pais. Identification of degenerate time- and space-dependent kernels in heat flow. In *Proc. of 5-th International Conference on nverse Problems in Engineering* (Cambridge, 11-15.07.2005), D. Lesnic ed., Leeds Univ. Press, Vol. III, P01, pp1 - 10, 2005.

Enno Pais and Jaan Janno. Identification of two time- and space-dependent kernels in a parabolic equation. *Electron. J. Diff. Eqns.*, No. 108, 1 -20, 2005.

Enno Pais. Degenerate memory kernels identification problem with flux-type additional conditions. *J. Inv. Ill-Posed Problems*, 14, 397 - 418, 2006.

Enno Pais and Jaan Janno. Inverse problem to determinate degenerate memory kernels in heat flux with third kind boundary conditions. *Math. Model. Anal.*, 11, 427 - 450, 2006.

9. Teadustöö põhisuunad:

Osatuletistega diferentsiaalvõrrandid ja mittekorrektsed ülesanded.

Curriculum vitae

1. Personal information:

Name: Enno Pais

Place and date of birth: 30.08.1944, Russia

Citizenship: Estonian
Marital status: married
Children: 3 daughters

2. Contact information:

Address: 16-140, Paekaare str, 13621 Tallinn Telephone: (+372) 56684492, (+372) 6363975

Email: ennopais@stv.ee

3. Education:

Institution Graduation date Education

Moscow State University 1967 MSc in Mathematics

Pärnu Secondary School No. 2 1962 secondary

4. Languages:

Level Language Language Level Russian advanced Finnish medium French advanced Swedish basic English medium Spanish basic

German Basic

5. Further Education:

Date Institution or Organization

1977 - 1978 Leningrad State University, Courses of French
1970 - 1973 Moscow State University, postgraduate studies

6. Professional Employment:

University or Organization	Date	Position
Tallinn University of Technology, Institute of Mathematics	1991 -	Lecturer
Tallinn Polytechnic University, Department of Mathematics	1981 -"1991	Teatcher
Higher Engineering School of Bamako, Rep. of Mali, Africa	1978 - 1981	Lecturer
Tallinn Polytechnic University, Department of Mathematics	1973 -"1978	Teaching Assistant
Tallinn Polytechnic University, Department of Applied Mathematics	1967 -"1970	Teaching Assistant

7. Theses:

Master thesis: *Iterative methods to solve finite difference equations describing the balance of cylindrical shells.*

8. Scientific Work:

Enno Pais. On the smoothness of eigenfunctions of the clamped plate near corners. (Russian) *Differential Equations*, Vol. 9, 9, 1700 - 1706, 1973.

Enno Pais. Identification of degenerate time- and space-dependent kernels in heat flow. In *Proc. of 5-th International Conference on nverse Problems in Engineering* (Cambridge, 11-15.07.2005), D. Lesnic ed., Leeds Univ. Press, Vol. III, P01, pp1 - 10, 2005.

Enno Pais and Jaan Janno. Identification of two time- and space-dependent kernels in a parabolic equation. *Electron. J. Diff. Eqns.*, No. 108, 1-20, 2005.

Enno Pais. Degenerate memory kernels identification problem with flux-type additional conditions. *J. Inv. Ill-Posed Problems*, 14, 397 - 418, 2006.

Enno Pais and Jaan Janno. Inverse problem to determinate degenerate memory kernels in heat flux with third kind boundary conditions. *Math. Model. Anal.*, 11, 427 - 450, 2006.

9. Main Areas of Scientific Work:

Partial differential equations and ill-posed problems.